Basel Committee on Banking Supervision



Instructions for Basel III monitoring

Version for banks **not** providing data for the trading book part of the exercise

REVISED - 13 February 2015



This publication is available on the BIS website (www.bis.org/bcbs/qis/). Grey underlined text in this publication shows where hyperlinks are available in the electronic version.
, , , , , , , , , , , , , , , , , , ,
© Bank for International Settlements 2015. All rights reserved. Brief excerpts may be reproduced or
translated provided the source is stated.
ISSN 92-9197- 870-1 (print) ISSN 92-9131- 870-1 (online)

Contents

1.		Introduction	1	
2.		General	2	
	2.1	Scope of the exercise	2	
	2.2	Filling in the data	3	
	2.3	Process	4	
	2.4	Reporting date	4	
	2.5	Structure of the Excel questionnaire	5	
3.		General information	5	
	3.1	General bank data (panel A)	5	
	3.2	Current capital (panel B)	8	
	3.3	Capital distribution data (panel C)	9	
4.		Risk-weighted assets, exposures and fully phased-in eligible capital	10	
	4.1	Overall capital requirements and actual capital ratios (worksheet "Requirements")	10	
	4.2	Definition of capital	12	
	4.3	Impact of the revised floors including the new standardised approach to operational risk	15	
5.		Leverage ratio	20	
	5.1	Introduction	20	
	5.2	On-balance sheet items (panel A)	21	
	5.3	Derivatives and off-balance sheet items (panel B)	27	
	5.4	On- and off-balance sheet items – additional breakdown of exposures (panel C)	31	
	5.5	Reconciliation (panel D)	32	
	5.6	Adjusted notional exposures for written credit derivatives (panel E)	33	
	5.7	Calculation of the leverage ratio (panel F)	35	
	5.8	Alternative methods for derivative exposures (panel G)	35	
	5.9	Alternative currency criteria for eligible cash variation margin in derivative transactions (panel H)	38	
	5.10	Memo items related to initial margin for centrally cleared transactions (panel I)	40	
	5.11	Business model categorisation (panel J)	41	
	5.12	2 EU-specific (panel K)	43	
6.		Liquidity	43	
	6.1	1 Liquidity coverage ratio (LCR)		
	6.2	Net Stable Funding Ratio (NSFR)	89	

7.		Trading book	. 115
8.		Standardised approach to credit risk	.115
	8.1	Paragraph references	.116
	8.3	Asset class specific information in "BB SA general"	.118
	8.4	Background on the "BB SA additional" worksheet	.121
9.		TLAC	.122
	9.1	Scope of application	.122
	9.2	General info for MPE	.123
	9.3	External TLAC and deduction treatment for G-SIBs according to the national implementation of Basel III	
	9.4	Internal TLAC	.137
	9.5	Holdings of TLAC (non-G-SIBs)	.142
	9.6	External TLAC (location)	.144
Anne	ex 1:	Changes compared to versions 2.8.x of the reporting template	. 148
Anne	ex 2:	Tentative schedule for upcoming Basel III monitoring exercises	. 149

Quantitative Impact Study Working Group of the Basel Committee on Banking Supervision

Chairman Mr Martin Birn, Secretariat of the Basel Committee on Banking Supervision,

Bank for International Settlements, Basel

Argentina Ms Verónica Balzarotti Central Bank of Argentina

Australia Mr David Connolly Australian Prudential Regulation Authority

Belgium Ms Claire Renoirte Banking, Finance and Insurance Commission

Brazil Mr Frederico Torres de Souza Central Bank of Brazil

Canada Mr Brian Rumas Office of the Superintendent of Financial Institutions

China Ms Jin Wang China Banking Regulatory Commission
France Ms Dominique Durant French Prudential Supervisory Authority

Germany Ms Dorothee Holl Deutsche Bundesbank

Hong Kong SAR Mr Andy Cheung Hong Kong Monetary Authority

IndiaMr Rajnish KumarReserve Bank of IndiaIndonesiaMr Boyke W SuadiIndonesia FSA (OJK)

ItalyMr Francesco PiersanteBank of ItalyJapanMr Jutaro KanekoBank of Japan

Mr Takahiro Ito Financial Services Agency
Mr Sunghoon Park Financial Supervisory Service

Luxembourg Ms Natalia Katilova Surveillance Commission for the Financial Sector

Netherlands Mr Ron Jongen Netherlands Bank

Russia Mr Aleksandr Stezhkin Central Bank of the Russian Federation

Saudi Arabia Mr Syed Mehdi Hassan Saudi Arabian Monetary Agency
Singapore Ms Sandy Ho Monetary Authority of Singapore

South Africa Mr Jaco Vermeulen South African Reserve Bank

Spain Ms Beatriz Domingo Bank of Spain

Sweden Mr Andreas Borneus Finansinspektionen
Ms Amelie Stierna Sveriges Riksbank

Switzerland Mr Uwe Steinhauser Swiss Financial Market Supervisory Authority FINMA

Turkey Mr Erhan Cetinkaya Banking Regulation and Supervision Agency

United Kingdom Mr Pavel Izmaylov Prudential Regulation Authority

United States Mr Eric Kennedy Board of Governors of the Federal Reserve System

Ms Eva Shi Federal Reserve Bank of New York

Ms Andrea Plante Federal Deposit Insurance Corporation

Mr Benjamin Pegg Office of the Comptroller of the Currency

Korea

European Central Mr Gernot Stania

Bank

ECB Single Supervisory Mechanism

Observers Mr Lampros Kalyvas

Mr Gintaras Griksas

European Banking Authority

European Commission

Mr Motohiro Hatanaka

Ms Raquel Lago Mr Davy Reinard

Mr Nik Faris Sallahuddin

Mr Carlos Sosa Mr Ju Quan Tan Ms Bat-el Berger Mr Otakar Cejnar Ms Lorraine Chung Ms Alisa Dombrovskaya

Ms Lillie Lam Ms Crystal Pun **Bank for International Settlements**

Instructions for Basel III monitoring

1. Introduction

The Basel Committee on Banking Supervision ("the Committee") is monitoring the impact of *Basel III: A global regulatory framework for more resilient banks and banking systems* ("the Basel III standards"), the *Basel III leverage ratio framework and disclosure requirements* ("the Basel III leverage ratio framework"), *Basel III: The Liquidity Coverage Ratio and liquidity risk monitoring tools* ("the Basel III LCR standards") and *Basel III: The Net Stable Funding Ratio* ("Basel III NSFR standards") on participating banks. The exercise will be repeated semi-annually with end-December and end-June reporting dates.

In addition to these recurring items, worksheets have been added to collect data on four ongoing policy initiatives of the Committee, the fundamental review of the trading book conducted by the Committee's Trading Book Group,² the review of the standardised approaches for credit risk³ and operational risk,⁴ and the design of a capital floors framework based on standardised approaches.⁵ Furthermore, data is being collected on the proposal for a common international standard on Total Loss-Absorbing Capacity (TLAC) for global systemic banks by the Financial Stability Board.⁶

The Committee will treat all individual bank data collected in this exercise strictly confidential and will not attribute them to individual banks.

The descriptions of data items in these instructions intend to facilitate the completion of the monitoring questionnaire and are not to be construed as an official interpretation of other documents published by the Committee.

This version of the instructions refers to versions 3.0.1 or later of the reporting template which should be used for the 31 December 2014 reporting date. Changes compared to the previous version of the reporting template are highlighted in the Annex.

The remainder of this document is organised as follows. Sections 2 and 3 discuss general issues such as the scope of the exercise, the process and the overall structure of the quantitative questionnaire. Section 4 discusses the worksheets for data collection on the definition of capital, capital requirements and the floors based on the standardised approaches including the new standardised approach to operational risk. Sections 5 and 6 discuss the Basel III leverage ratio and liquidity, respectively. Section 7 presents the worksheets for collecting data for the Committee's fundamental review of the trading book

- Basel Committee on Banking Supervision, Basel III: A global regulatory framework for more resilient banks and banking systems (revised June 2011), June 2011, www.bis.org/publ/bcbs189.htm; Basel Committee on Banking Supervision, Basel III: The Liquidity Coverage Ratio and liquidity risk monitoring tools, January 2013, www.bis.org/publ/bcbs238.htm; Basel Committee on Banking Supervision, Basel III: The Net Stable Funding Ratio, October 2014, www.bis.org/bcbs/publ/d295.htm.
- Basel Committee on Banking Supervision, Fundamental review of the trading book: outstanding issues consultative document, December 2014, www.bis.org/publ/d305.htm.
- Basel Committee on Banking Supervision, *Revisions to the standardised approach for credit risk consultative document*, December 2014, www.bis.org/publ/d307.htm.
- Basel Committee on Banking Supervision, Operational risk Revisions to the simpler approaches consultative document, October 2014, www.bis.org/publ/bcbs291.htm.
- Basel Committee on Banking Supervision, *Capital floors: the design of a framework based on standardised approaches*, December 2014, www.bis.org/publ/d306.htm.
- Financial Stability Board, Adequacy of loss-absorbing capacity of global systemically important banks in resolution, consultative document, November 2014, www.financialstabilityboard.org/wp-content/uploads/TLAC-Condoc-6-Nov-2014-FINAL.pdf.

while Section 8 describes the worksheet for the collection of data relevant to the Committee's proposed new standardised approach to credit risk. Section 9 introduces the worksheets to collect data on the proposed standard on Total Loss-Absorbing Capacity (TLAC).

General

2.1 Scope of the exercise

Participation in the monitoring exercise is voluntary. The Committee expects both large internationally active banks and smaller institutions to participate in the study, as all of them will be materially affected by some or all of the revisions of the various standards. Where applicable and unless noted otherwise, data should be reported for consolidated⁷ groups.

The monitoring exercise is targeted at both banks under the Basel II/III frameworks and at those still subject to Basel I.⁸ However, as outlined in the remainder of these instructions some parts of the questionnaire are only relevant for banks subject to Basel II or to banks applying a particular approach. If **Basel I** figures are used, they should be calculated based on the **national implementation**, referred to as "Basel I" in this document. In some countries supervisors may have implemented additional rules beyond the 1988 Accord or may have made modifications to the Accord in their national implementation, and these should be considered in the calculation of "Basel I" capital requirements for the purposes of this exercise. **Unless stated otherwise**, if a bank has implemented **Basel II** at a particular reporting date, it should calculate capital requirements based on the **national implementation** of the Basel II framework, referred to as "Basel II" in this document. **Unless stated otherwise**, the changes to the risk-weighted asset calculation of the Basel II framework introduced in 2009 which are collectively referred to as "Basel 2.5" (*Revisions to the Basel II market risk framework* ("the Revisions") and *Enhancements to the Basel II framework* ("the Enhancements")) and through the Basel III framework should only be reflected if they are part of the applicable regulatory framework at the reporting date.

When providing data on Basel III, banks should also take into account the frequently asked questions on capital, counterparty credit risk and the leverage ratio published by the Committee.¹¹

This data collection exercise should be completed on a best-efforts basis. Ideally, banks should include all their assets in this exercise. However, due to data limitations, inclusion of some assets (for example the portfolio of a minor subsidiary) may turn out to be an unsurpassable hurdle. In these cases, banks should consult their relevant national supervisor to determine how to proceed.

⁷ This refers to the consolidation for regulatory rather than accounting purposes.

⁸ Basel Committee on Banking Supervision, *International convergence of capital measurement and capital standards (updated to April 1998)*, 1998, www.bis.org/publ/bcbsc111.htm.

Basel Committee on Banking Supervision, Revisions to the Basel II market risk framework - updated as of 31 December 2010, February 2011, www.bis.org/publ/bcbs193.htm.

Basel Committee on Banking Supervision, Enhancements to the Basel II framework, July 2009, www.bis.org/publ/bcbs157.htm.

Basel Committee on Banking Supervision, Basel III definition of capital – Frequently asked questions, December 2011, www.bis.org/publ/bcbs211.htm; Basel Committee on Banking Supervision, Basel III counterparty credit risk – Frequently asked questions, December 2012, www.bis.org/publ/bcbs237.htm; Basel Committee on Banking Supervision, Frequently asked questions on the Basel III leverage ratio framework, www.bis.org/publ/bcbs293.htm.

2.2 Filling in the data

The Basel III monitoring workbook available for download on the Committee's website is for information purposes only. While the structure of the workbooks used for the Basel III monitoring exercise is the same in all participating countries, it is important that banks only use the workbook obtained from their respective national supervisory agency to submit their returns. Only these workbooks are adjusted to reflect the particularities of the regulatory frameworks in participating countries. National supervisory agencies may also provide additional instructions if deemed necessary.

Data should only be entered in the yellow and green shaded cells. There are also some pink cells which will be completed by the relevant national supervisory agency. It is important to note that any modification to the worksheets might render the workbook unusable both for the validation of the final results and the subsequent aggregation process.

Cell colours	used in the	Basel III	monitoring	reporting	template
CCII COIOGIS	asca III ciic		11101111011119		complace

Colour	Worksheet(s)	Content
Yellow	All	Input cell.
Green	Requirements	To be completed if requested by the national supervisor or in order to calculate the capital ratios in panel C.
	DefCap	To be filled in if necessary based on the national implementation of the definition of capital.
	DefCap-MI	Worksheet is optional, can be used by banks to generate inputs for the "DefCap" worksheet.
	Leverage Ratio	Additional information needed to monitor the leverage ratio and its components during the transition period, in accordance with the transitional arrangements set out in paragraphs 165 to 167 of the Basel III standards. Banks are encouraged to fill in green cells on a best-efforts basis as well.
	LCR, NSFR	To be completed if requested by the national supervisor in light of national discretion choices.
	TB, Standardised approach BB	Additional information to be completed on a best efforts basis.
Pink	All	To be completed by the supervisor.
White, orange	All	Calculation result. Must not be changed.

Where information is not available, the corresponding cell should be left empty. No text such as "na" should be entered in these cells. Also, banks must not fill in any arbitrary numbers to avoid error messages or warnings which may be provided by their supervisors. However, leaving a cell empty could trigger exclusion from some or all of the analyses if the respective item is required, ie it should be aimed at providing data for all yellow cells. The automated calculations in the workbook indicate whether or not a certain item can be calculated using the data provided. The national supervisor will provide guidance on which of the green cells should be filled in by a particular bank.

Data can be reported in the most convenient currency. The currency which has been used should be recorded in the "General Info" worksheet. Supervisors will provide the relevant exchange rate for converting the reporting currency to euros. If 1,000 or 1,000,000 currency units are used for reporting, this should also be indicated in this worksheet. When choosing the reporting unit, it should be considered that the worksheet shows all amounts as integers. The same currency and unit should be used for all amounts throughout the workbook, irrespective of the currency of the underlying exposures.

Percentages should be reported as decimals and will be converted to percentages automatically. For example, 1% should be entered as 0.01. 12

Banks using the Basel II internal ratings-based (IRB) approaches should, where applicable, report risk-weighted assets after applying the scaling factor of 1.06 to credit risk-weighted assets.

The reporting template includes checks in several of the worksheets. If one of these checks shows "No" or "Fail", please refer to the explanatory text and the formula in the check cell and correct the input data to which the check refers. An overview of the results of all checks is provided on the "Checks" worksheet.

The Committee is aware that some banks might not yet have implemented some of the models and processes required for the calculations. In such cases banks may provide quantitative data on a "best-efforts" basis. In case of doubt, they should discuss with the relevant national supervisor how to proceed. Where the approach used for the Basel III monitoring differs materially from the final implementation, this should be explained in a separate note.

Unless noted otherwise, banks should only report data for the approach they are currently using or are intending to use. Cells provided for various approaches are in general intended to facilitate partial use and do **not** require banks to conduct alternative calculations for the same set of exposures.

2.3 Process

The Basel Committee or its Secretariat will not collect any data directly from banks. Therefore, banks in participating countries should contact their supervisory agency to discuss how the completed workbooks should be submitted. National supervisors will forward the relevant data to the Secretariat of the Basel Committee where individual bank data will be treated strictly confidential and will not be attributed to individual banks.

Similarly, banks should direct all questions related to this study, the related rules, standards and consultative documents to their national supervisory agencies. Where necessary, they will coordinate their responses through the Secretariat of the Basel Committee to provide responses that are consistent across countries. A document with responses to frequently asked questions will be maintained on the Basel Committee's website. ¹³

Banks should specify any instance where they had to deviate from the instructions provided in an additional document.

2.4 Reporting date

If possible, and unless the national supervisor has provided different guidance, generally all data should be reported as of end-December or end-June, as applicable. If data availability does not allow a bank to use these reporting dates or if the financial year differs from the calendar year, suitable alternatives should be discussed with the relevant national supervisor.

Depending on the regional options of the operating system used, it might be necessary to use a different decimal symbol. It might also be necessary to switch off the option "Enable automatic percent entry" in the Tools/Options/Edit dialog of Excel if percentages cannot be entered correctly.

www.bis.org/bcbs/qis/.

2.5 Structure of the Excel questionnaire

The Excel workbook consists of 20 worksheets. All banks participating in the impact study should generally complete them. Some banks may be directed by their supervisor to complete only certain parts of the workbook. Finally, the "Checks" worksheet provides an overview of all the checks included on the other worksheets. The worksheets requiring data input are the following:

- The worksheet "General Info" is intended to capture **general information** regarding the bank, eligible capital and deductions as well as capital distribution data. This worksheet should be completed by all banks.
- The worksheet "Requirements" captures overall capital requirements and actual capital ratios. This worksheet should be completed by all banks.
- The "DefCap" worksheet is related to the **definition of capital**. It captures more detailed information on the Basel III definition of capital and its impact on risk-weighted assets. The "DefCap-MI" worksheet helps banks with the calculation of regulatory adjustments for minority interest which is an input required on the "DefCap" worksheet; providing data on this worksheet is optional.
- The "Floors" worksheet collects some additional data which will allow the Committee to assess
 the impact of the proposed capital floors based on the standardised approaches and the
 proposed new standardised approach to operational risk.
- The **leverage ratio** worksheet ("Leverage Ratio") captures data necessary for the calculation of the leverage ratio.
- The **liquidity** worksheets ("LCR" and "NSFR") are intended to capture key data regarding the liquidity coverage ratio and net stable funding ratio measures.
- The "BB" worksheets collect data for the Committee's QIS on the proposed **new standardised approach to credit risk** in the banking book.
- The "TB" worksheets collect data for the Committee's QIS on the fundamental review of the trading book. These worksheets should be completed by banks which currently use either the internal models approach or the standardised approach to market risk.
- The "TLAC" worksheets collect data on the proposal for a common international standard on Total Loss-Absorbing Capacity (TLAC) for global systemic banks by the Financial Stability Board. Non G-SIBs should only complete the worksheet "TLAC holdings (non-G-SIBs), while G-SIBs should leave this worksheet empty and only complete the three worksheets "G-SIB TLAC External", "G-SIB TLAC Internal" and "G-SIB TLAC location". Certain G-SIBs will also be asked by their supervisors to provide additional data on separate templates.

3. General information

The "General Info" worksheet gathers basic information that is needed to process and interpret the survey results. Banks only providing data for liquidity, the fundamental review of the trading book or the standardised approach to credit risk are only required to fill in panels A and B.

3.1 General bank data (panel A)

Panel A of the "General Info" worksheet deals with bank and reporting data conventions.

A1) Reporting data 5	Row	Column	Heading	Description
6 C Region code Leave blank 7 C Bank number 8 C CMG-relevant Leave blank 9 C Data validation only Leave blank 10 C Bank is a single legal entity Leave blank 11 C Bank is a subsidiary of a banking group 12 C Bank is a subsidiary with a non-EU parent (EU only) 13 C Bank type Leave blank 14 C Bank group Leave blank 15 C Bank type (numeric) Leave blank 16 C G-SIB surcharge, Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, TeET1 capital 19 C Domestic surcharges, Total capital 20 C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous period 26 C Use Lex data Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous period 20 C Use shanking book SA data Leave blank 20 C Use tading book general data Leave blank 21 C Comparable to the previous Leave blank 22 C Use banking book SA data Leave blank 23 C Comparable to the previous Leave blank 24 C Use tading book general data Leave blank 25 C Comparable to the previous Leave blank 26 C Use banking book SA data Leave blank 27 C Comparable to the previous Leave blank 28 C Use tading book general data Leave blank 30 C Use sanking book SA data Leave blank 31 C Comparable to the previous Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	A1)	Reporting d	lata	
7 C Bank number Leave blank 8 C CMG-relevant Leave blank 9 C Data validation only Leave blank 10 C Bank is a single legal entity Leave blank 11 C Bank is a subsidiary of a banking group 12 C Bank is a subsidiary with a non-EU parent (EU only) 13 C Bank type Leave blank 14 C Bank group Leave blank 15 C Bank type (numeric) Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 Leave blank 18 C Domestic surcharges, Tier 1 Leave blank 19 C Domestic surcharges, Tier 1 Leave blank 20 C Conversion rate (in eurostriagia) 20 C Conversion rate (in eurostriagia) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous period 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous period 28 C Use Leverage ratio data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous period 20 C Comparable to the previous Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 20 C Use banking book SA data Leave blank 21 C Use banking book SA data Leave blank 22 C Use banking book SA data Leave blank 23 C Comparable to the previous Leave blank 24 C Use trading book general data Leave blank 25 C Comparable to the previous Leave blank	5	С	Country code	Leave blank
8 C CMG-relevant Leave blank 9 C Data validation only Leave blank 10 C Bank is a single legal entity Leave blank 11 C Bank is a subsidiary of a banking group 12 C Bank is a subsidiary with a non-EU parent (EU only) 13 C Bank type Leave blank 14 C Bank group Leave blank 15 C Bank type (numeric) Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 capital 19 C Domestic surcharges, total capital 20 C Conversion rate (in euros/sreporting currency) 21 C Submission date (ynyy-mm-dd) 22 C U Se capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous period 26 C Use LCR data Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous period 20 C Comparable to the previous period 21 C Submission data Leave blank 22 C C Use capital data Leave blank 23 C Comparable to the previous Leave blank 24 C Duse thors data Leave blank 25 C Comparable to the previous Leave blank 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous period 30 C Use NSFR data Leave blank 31 C Comparable to the previous Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	6	С	Region code	Leave blank
9 C Data validation only Leave blank 10 C Bank is a single legal entity Leave blank 11 C Bank is a subsidiary of a banking group 12 C Bank sa subsidiary with a non-EU parent (EU only) 13 C Bank type Leave blank 14 C Bank type Leave blank 15 C Bank type (numeric) Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 Leave blank 18 C Domestic surcharges, Tier 1 Leave blank 19 C Domestic surcharges, Tier 1 Leave blank 20 C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous period 26 C Use LCR data Leave blank 27 C Comparable to the previous period 30 C Use NSFR data Leave blank 31 C Comparable to the previous period 32 C Use tarding book SA data Leave blank 33 C Comparable to the previous period 34 C Use trading book SA data Leave blank 35 C Comparable to the previous Leave blank 26 C Use tarding book SA data Leave blank 27 C Comparable to the previous Leave blank 28 C Use tarding book SA data Leave blank 29 Leave blank 29 C Use banking book SA data Leave blank 29 Leave blank	7	С	Bank number	Leave blank
10 C Bank is a single legal entity 11 C Bank is a subsidiary of a banking group 12 C Bank is a subsidiary of a banking group 13 C Bank type Leave blank 14 C Bank group Leave blank 15 C Bank type Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 Leave blank 19 C Domestic surcharges, Tier 1 Leave blank 20 C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-m-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous period 20 C Comparable to the previous Leave blank 21 C Comparable to the previous Leave blank 22 C Use LCR data Leave blank 23 C Comparable to the previous Leave blank 24 C Use floors data Leave blank 25 C Comparable to the previous Leave blank 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous Leave blank 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous Leave blank 20 C Use NSFR data Leave blank 31 C Comparable to the previous Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	8	С	CMG-relevant	Leave blank
11 C Bank is a subsidiary of a banking group 12 C Bank is a subsidiary with a non-EU parent (EU only) 13 C Bank type Leave blank 14 C Bank group Leave blank 15 C Bank type Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 capital 19 C Domestic surcharges, Tier 1 capital 19 C Domestic surcharges, total capital 20 C C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-modd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous period 26 C Use LCR data Leave blank 27 C Comparable to the previous Leave blank 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 20 C Use NSFR data Leave blank 20 C Use NSFR data Leave blank 21 C Lomparable to the previous Leave blank 22 C Use banking book SA data Leave blank 23 C Lows trading book general data Leave blank 24 C Use trading book general data Leave blank 25 C Lows trading book general data Leave blank 26 C Use trading book general data Leave blank 27 C Comparable to the previous Leave blank	9	С	Data validation only	Leave blank
banking group 12 C Bank is a subsidiary with a non- EU parent (EU only) 13 C Bank type Leave blank 14 C Bank group Leave blank 15 C Bank type (numeric) Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 capital 19 C Domestic surcharges, total capital 20 C C Conversion rate (in Leave blank capital) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C C Comparable to the previous period 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous Leave blank 29 C Lows LEVR data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous Leave blank 29 C Lows LEVR data Leave blank 20 C Use trading book general data Leave blank 20 C Leave blank 21 C Leave blank 22 C Leave blank 23 C Leave blank 24 Leave blank 25 Leave blank 26 Leave blank 27 Leave blank 28 Leave blank 29 Leave blank 29 Leave blank 29 Leave blank 20 Leave blank 20 Leave blank 21 Leave blank 22 Leave blank 23 Leave blank 24 Leave blank 25 Leave blank 26 Leave blank 27 Leave blank 28 Leave blank 29 Leave blank 29 Leave blank 20 Leave blank 20 Leave blank 21 Leave blank 22 Leave blank	10	С	Bank is a single legal entity	Leave blank
EU parent (EU only) 13	11	С	,	Leave blank
14	12	С		Leave blank
15 C Bank type (numeric) Leave blank 16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 capital 19 C Domestic surcharges, total capital 20 C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous period 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 21 C Comparable to the previous Leave blank 22 C Use Leverage ratio data Leave blank 23 C Comparable to the previous Leave blank 24 Leave blank 25 C Comparable to the previous Leave blank 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous Leave blank 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 30 C Use NSFR data Leave blank 31 C Comparable to the previous Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	13	С	Bank type	Leave blank
16 C G-SIB surcharge Leave blank 17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 capital 19 C Domestic surcharges, Tier 1 Leave blank capital 20 C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous Leave blank 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous Leave blank 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 20 C Use LCR data Leave blank 21 C Comparable to the previous Leave blank 22 C Use LCR data Leave blank 23 C Comparable to the previous Leave blank 24 C Use LCR data Leave blank 25 C Comparable to the previous Leave blank 26 C Use LCR data Leave blank 27 C Comparable to the previous Leave blank 28 C Use LCR data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous Leave blank 20 C Use NSFR data Leave blank 21 C Comparable to the previous Leave blank 22 C Use banking book SA data Leave blank 23 C Use banking book SA data Leave blank 24 C Use trading book general data Leave blank 25 C Comparable to the previous Leave blank 26 C Use trading book general data Leave blank	14	С	Bank group	Leave blank
17 C Domestic surcharges, CET1 capital 18 C Domestic surcharges, Tier 1 Leave blank capital 19 C Domestic surcharges, total capital 20 C Conversion rate (in euros/reporting currency) 21 C Submission date (yyyy-mm-dd) 22 C Use capital data Leave blank 23 C Comparable to the previous period 24 C Use floors data Leave blank 25 C Comparable to the previous Leave blank 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous period 28 C Use Lexerage ratio data Leave blank 27 C Comparable to the previous Leave blank 28 C Use Lexerage ratio data Leave blank 29 C Comparable to the previous Leave blank 29 C Comparable to the previous Leave blank 29 C Use NSFR data Leave blank 30 C Use NSFR data Leave blank 31 C Comparable to the previous Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank	15	С	Bank type (numeric)	Leave blank
capital C Domestic surcharges, Tier 1 capital C Domestic surcharges, total capital C Conversion rate (in euros/reporting currency) C Submission date (yyyy-mm-dd) C Use capital data Leave blank C Comparable to the previous period C Use floors data Leave blank C Comparable to the previous Leave blank C Comparable to the previous Leave blank C Comparable to the previous Leave blank C C Use Leverage ratio data Leave blank C C Comparable to the previous Leave blank C C Comparable to the previous Leave blank C C Comparable to the previous Leave blank C C Use LCR data Leave blank C Use LCR data Leave blank C Use NSFR data Leave blank C Comparable to the previous Leave blank C C Comparable to the previous Leave blank	16	С	-	Leave blank
capital 19	17	С		Leave blank
capital 20	18	С	_	Leave blank
euros/reporting currency) 21	19	С	_	Leave blank
(yyyy-mm-dd) 22	20	С	1	Leave blank
Comparable to the previous period Leave blank Comparable to the previous Leave blank	21	С		Leave blank
period 24	22	С	Use capital data	Leave blank
25 C Comparable to the previous period 26 C Use Leverage ratio data Leave blank 27 C Comparable to the previous period 28 C Use LCR data Leave blank 29 C Comparable to the previous period 30 C Use NSFR data Leave blank 31 C Comparable to the previous period 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	23	С		Leave blank
period 26	24	С	Use floors data	Leave blank
27 C Comparable to the previous period Leave blank 28 C Use LCR data Leave blank 29 C Comparable to the previous period Leave blank 30 C Use NSFR data Leave blank 31 C Comparable to the previous period Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank Leave blank Leave blank	25	С		Leave blank
period 28	26	С	Use Leverage ratio data	Leave blank
29 C Comparable to the previous period 30 C Use NSFR data Leave blank 31 C Comparable to the previous period 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank C Comparable to the previous Leave blank C Comparable to the previous Leave blank Leave blank	27	С		Leave blank
period C Use NSFR data Leave blank C Comparable to the previous period C Use banking book SA data Leave blank C Comparable to the previous Leave blank C Comparable to the previous Leave blank C Use trading book general data Leave blank C Comparable to the previous Leave blank C Comparable to the previous Leave blank	28	С	Use LCR data	Leave blank
31 C Comparable to the previous Leave blank 32 C Use banking book SA data Leave blank 33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	29	С		Leave blank
period 32	30	С	Use NSFR data	Leave blank
33 C Comparable to the previous Leave blank 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	31	С		Leave blank
period 34 C Use trading book general data Leave blank 35 C Comparable to the previous Leave blank	32	С	Use banking book SA data	Leave blank
35 C Comparable to the previous Leave blank	33	С		Leave blank
	34	С	Use trading book general data	Leave blank
period	35	С		Leave blank
36 C Use trading book SBA data Leave blank	36	С	Use trading book SBA data	Leave blank

Row	Column	Heading	Description
37	С	Comparable to the previous period	Leave blank
38	С	Use trading book IMA data	Leave blank
39	С	Comparable to the previous period	Leave blank
40	С	Use trading book IMA P&L attribution data	Leave blank
41	С	Comparable to the previous period	Leave blank
42	С	Use operational risk data	Leave blank
43	С	Comparable to the previous period	Leave blank
44	С	Use TLAC holdings data	Leave blank
45	С	Comparable to the previous period	Leave blank
46	С	Use G-SIB TLAC data	Leave blank
47	С	Comparable to the previous period	Leave blank
48	С	Reporting date (yyyy-mm-dd)	Date as of which all data are reported in worksheets.
49	С	Reporting currency (ISO code)	Three-character ISO code of the currency in which all data are reported (eg USD, EUR).
50	С	Unit (1, 1000, 1000000)	Units (single currency units, thousands, millions) in which results are reported.
51	С	Accounting standard	Indicate the accounting standard used.

A2) Approaches to credit risk

Banks using more than one approach to calculate risk-weighted assets for credit risk should select **all** those approaches in rows 53 to 56. However, if a bank uses the foundation IRB approach for all non-retail portfolios subject to the PD/LGD approach and the IRB approach to retail for the retail portfolio, "foundation IRB" should be selected as the only IRB approach (and additionally Basel I or the standardised approach if applicable). If an IRB bank has only a retail portfolio and no other exposures subject to a PD/LGD approach, then "advanced IRB" should be selected as the only IRB approach (and additionally Basel I or the standardised approach if applicable).

Sasel I Indicate whether Basel I is used to calculate capital requirements for a portion of the exposures reported in this study. C Basel II/III standardised approach is used to calculate capital requirements for a portion of the exposures reported in this study. C Basel II/III FIRB approach Indicate whether the foundation IRB approach of Basel II or III is used to calculate capital requirements for a portion of the exposures reported in this study. C Basel II/III FIRB approach Indicate whether the foundation IRB approach of Basel II or III is used to calculate capital requirements for a portion of the exposures reported in this study. Indicate whether the advanced IRB approach of Basel II or III is used to calculate capital requirements for a portion of the exposures reported in this study. A3) Accounting information Total assets following the relevant accounting balance sheet (considering the regulatory consolidation).	• •		, ,	
approach is used to calculate capital requirements for a portion of the exposures reported in this study. 55 C Basel II/III FIRB approach Indicate whether the foundation IRB approach of Basel II or III is used to calculate capital requirements for a portion of the exposures reported in this study. 56 C Basel II/III AIRB approach Indicate whether the advanced IRB approach of Basel II or III is used to calculate capital requirements for a portion of the exposures reported in this study. A3) Accounting information 58 C Accounting total assets Total assets following the relevant accounting balance sheet	53	С	Basel I	requirements for a portion of the exposures reported in this
III is used to calculate capital requirements for a portion of the exposures reported in this study. 56	54	С	'	is used to calculate capital requirements for a portion of the
is used to calculate capital requirements for a portion of the exposures reported in this study. A3) Accounting information 58 C Accounting total assets Total assets following the relevant accounting balance sheet	55	С	Basel II/III FIRB approach	III is used to calculate capital requirements for a portion of
58 C Accounting total assets Total assets following the relevant accounting balance sheet	56	С	Basel II/III AIRB approach	·
	A3)	Accounting	information	
	58	С	Accounting total assets	9

3.2 Current capital (panel B)

Panel B of the "General Info" worksheet deals with information on eligible capital and deductions. While the relevant amounts under the fully phased-in Basel III standards and under the fully phased-in national implementation of these standards are calculated automatically based on input on the "DefCap" worksheet, banks should enter the capital amounts eligible at the reporting date in column C according to the national implementation of the Basel standards. This calculation should be conducted in the same way as the calculation of eligible capital for solvency reporting to the national supervisory agency at the reporting date.

The regulatory adjustments should be assigned to the tier of capital **from which they are actually taken**. For example, if a bank has not enough additional Tier 2 capital to make all those regulatory adjustments which can be made to Tier 2 capital, the adjustment should be reported as an adjustment to the relevant higher tier of capital.

Row	Column	Heading	Description
For repor		which the bank is not yet subject t	o Basel III, those elements of Tier 1 capital which are not asel I or Basel II should be reported in column C of these rows.
65	С	Prior to regulatory adjustments, national rules as at reporting date	Amount of gross Common Equity Tier 1 capital. This line should not include any regulatory adjustments.
66	С	Regulatory adjustments, national rules as at reporting date	Enter all regulatory adjustments to Common Equity Tier 1 capital elements. Banks should generally not report regulatory adjustments in this row that are applied to total Tier 1 capital as these should generally be reported in row 69. The only exception to this is in cases where the deductions in row 69 would otherwise exceed the Additional Tier 1 instruments reported in row 68.
For repor	der the natior	which the bank is not yet subject t	o Basel III, those elements of Tier 1 capital which are subject to sel II (eg hybrid capital) should be reported in column C of
68	С	Prior to regulatory adjustments, national rules as at reporting date	Enter the amount of gross Additional Tier 1 capital. This line should not include any regulatory adjustments.
69	С	Regulatory adjustments, national rules as at reporting date	Enter all regulatory adjustments to Additional Tier 1 capital elements. If the sum of the regulatory adjustments exceeds the amount reported in row 68 the excess should be reported in row 66 (ie the regulatory adjustments reported in row 69 must not exceed the capital reported in row 68).
Tier 2 cap	ital	'	
73	С	Prior to regulatory adjustments, national rules as at reporting date	Enter the amount of gross Tier 2 capital. This line should not include any regulatory adjustments.
74	С	Regulatory adjustments, national rules as at reporting date	Enter all regulatory adjustments to Tier 2 capital elements and to total capital elements. If the sum of the regulatory adjustments exceeds the amount reported in row 73 the excess should be reported in row 69 (ie the regulatory adjustments reported in row 74 must not exceed the capital reported in row 73).
Tier 3 cap	ital		
76	С	Tier 3 capital	Enter the amount of Tier 3 capital. For banks which are subject to Basel III at the reporting date, this cell should be 0.

3.3 Capital distribution data (panel C)

Panel C of the "General Info" worksheet deals with data on banks' income, capital distributions and capital raised. All data should be provided for the six-month period ending on the reporting date. Distributions should be reported in the period in which they are recognised on the balance sheet.

Row	Column	Heading	Description
Income			
81	С	Profit after tax	Enter the total amount of profit (loss) after tax. This should include profits attributable to minority shareholders.
82	C	Profit after tax prior to the deduction of relevant (ie expensed) distributions below	Enter the total amount of profit (loss) after tax including profits attributable to minority shareholders, but prior to the relevant distributions listed in the section below. The relevant distributions are only those which were included in the income statement in such a way as to reduce profit after tax as set out in row 81 (ie items that were expensed), and thus the relevant distributions are not necessarily the sum of the items listed below. The line seeks to collect the profit after tax which would have been reported had none of the distributions listed below been paid. As such any tax impact of making such payments should also be reversed in this line.
Distribution	1	I	
84	С	Common share dividends	Enter the total common share dividend payments. The amount entered should be the amount paid in cash, not stock.
85	С	Other coupon/dividend payments on Tier 1 instruments	Enter the total coupon/dividend payments paid to other Tier 1 instruments. The amount entered should be the amount paid in cash, not stock. It should include both amounts which were reported in the income statement as an interest expense and amounts which were reported as a distribution of profits.
86	С	Common stock share buybacks	Enter the total common stock share buybacks (effective amounts).
87	С	Other Tier 1 buyback or repayment (gross)	Enter the total gross buyback or repayment of other Tier 1 instruments (effective amounts).

Row	Column	Heading	Description
88	С	Discretionary staff compensation/bonuses	Enter the total amount of discretionary staff bonuses and other discretionary staff compensation. These amounts should be included if and when they result in a reduction of Tier 1 capital.
			For purposes of the Basel III monitoring exercise, discretionary staff bonuses and other discretionary compensation include all variable compensation to staff that the bank is not contractually obliged to make. Banks should only include such amounts if they result in a reduction in Tier 1 capital or would have resulted in an increase in Tier 1 capital if they had not been made. For example, under US GAAP, a bank is required to classify as a liability certain shares that give employees the right to require their employer to repurchase shares in exchange for cash equal to the fair value of the shares. As such discretionary compensation results in a reduction in GAAP equity and consequently Tier 1 capital, it would be included in row 88 of the "General Info" worksheet. Similarly, discretionary compensation made out of retained net income would have resulted in an increase in Tier 1 capital if it had not been made and therefore should also be included in row 88. By contrast, compensation to employees in the form of newly issued shares may in certain circumstances result in an increase in the number of outstanding shares with no change in GAAP equity and consequently no reduction in Tier 1 capital. These amounts should not be included in row 88 of the "General Info" worksheet.
89	С	Tier 2 buyback or repayment (gross)	Enter the total gross buyback or repayment of Tier 2 instruments (effective amounts).

Capital raised (gross)

Since these are cells to report newly issued capital amounts, the amounts of capital raised must always be positive or zero. Banks should apply the Basel III definition of capital in all reporting periods. Even if Basel III is not yet in force in a jurisdiction at the reporting date, all amounts in rows 88 to 90 should be reported based on Basel III definitions, including the 13 January 2011 press release on loss absorbency at the point of non-viability. Profit retention should not be included in the amounts of capital raised reported in this panel.

91	С	CET1	Enter the total gross Common Equity Tier 1 capital issued.
92	С	Additional Tier 1	Enter the total gross Additional Tier 1 capital issued.
93	С	Tier 2	Enter the total gross Tier 2 capital issued.

4. Risk-weighted assets, exposures and fully phased-in eligible capital

4.1 Overall capital requirements and actual capital ratios (worksheet "Requirements")

The "Requirements" worksheet deals with overall capital requirements and actual capital ratios. It is not strictly required to fill in the green cells on this worksheet unless a bank has been asked to do so by their supervisor. However, filling in the green cells allows for calculation of the capital ratios in panel C of the "Requirements" worksheet which is a useful check.

Kow Column Heading Description	Row	Column	Heading	Description
--------------------------------	-----	--------	---------	-------------

A) Risk-weighted assets according to the framework in place at the reporting date

In rows 6 to 10, banks have to report in column C risk-weighted assets for their exposures subject to the Basel I credit risk framework, in column D risk-weighted assets from the Basel II/III standardised approach to credit risk and in column E risk-weighted assets from the foundation or advanced internal ratings-based approach. The yellow cells for all approaches a bank is using according to the information provided in rows 53 to 56 of the "General Info" worksheet must be filled in completely. For example, a bank using the IRB approach and partial use of the standardised approach must fill in both columns D and E. If a bank does not have a particular portfolio, risk-weighted assets should be reported as 0.

The sets of exposures for which RWA are reported in columns C, D and E must be mutually exclusive.

Risk-weighted assets should reflect the 1.06 scaling factor to IRB credit risk-weighted assets where relevant and, unless noted otherwise, be calculated using the standards in place at the reporting date.

6	C–E	Credit risk (including CCR and non-trading credit risk); of which:	Overall credit risk-weighted assets including counterparty credit risk exposures and non-trading credit risk, but not including CVA capital charges or exposures to CCPs, after applying the 1.06 scaling factor to IRB credit risk-weighted assets.
7	C–E	Counterparty credit risk exposures (not including CVA charges or charges for exposures to CCPs)	Of the amount reportable in row 6, risk-weighted assets for counterparty credit risk exposures.
8	C–E	Sovereign exposures; of which; RWA	Of the credit risk-weighted asset amount reportable in row 6, amount for sovereign exposures.
8	F–H	Sovereign exposures; of which; Exposure amount	Exposure amount post credit risk mitigation for which risk-weighted assets have been reported in cell C8 to E8.
9	C–E	Domestic sovereign exposures; RWA	Of the risk-weighted asset amount reported in cells C8 to E8, amount for domestic sovereign exposures.
9	F–H	Domestic sovereign exposures; Exposure amount	Of the exposure amount reported in cells F8 to H8, amount for domestic sovereign exposures.
10	D, E	Securitisations	Of the amount reported in row 6, risk-weighted assets for securitisation exposures.
15	С	Qualifying central counterparties; Trade exposures (including client cleared trades); RWA	This includes any risk-weighted assets for trade exposures under Method 1 or Method 2, including RWA for SFT cleared through QCCPs. Of note, this includes risk-weighted assets for QCCPs subject to Method 2 where the cap of 20% of trade exposures is binding. Banks should enter a 0 in years in which this capital charge is not yet in force.
16	С	Qualifying central counterparties; Default fund exposures; RWA	This includes any risk-weighted assets for default fund contributions to QCCPs. Of note, this excludes risk-weighted assets for QCCPs subject to Method 2 where the cap of 20% of trade exposures is binding. Banks should enter a 0 in years in which this capital charge is not yet in force.
18	С	CVA capital charge (risk- weighted asset equivalent); Advanced CVA risk capital charge	Risk-weighted asset equivalent of the advanced CVA risk capital charge (ie the advanced CVA risk capital charge times 12.5). Banks should enter a 0 in years in which this capital charge is not yet in force.
19	С	CVA capital charge (risk- weighted asset equivalent); Standardised CVA risk capital charge	Risk-weighted asset equivalent of the standardised CVA risk capital charge (ie the standardised CVA risk capital charge times 12.5). Banks should enter a 0 in years in which this capital charge is not yet in force.
19	D	CVA capital charge (risk- weighted asset equivalent); standardised CVA risk capital charge	The exposure amount subject to the standardised CVA risk capital charge under Basel III as per paragraph 99 of the Basel III document which is externally rated.

Row	Column	Heading	Description
19	E	CVA capital charge (risk- weighted asset equivalent); standardised CVA risk capital charge	The exposure amount subject to the standardised CVA risk capital charge under Basel III as per paragraph 99 of the Basel III document which is unrated.
20	С	Market risk	Total market risk capital requirements. The capital charge should be converted to risk-weighted assets.
21	F	Amount of sovereign exposures in the trading book; of which	Amount of sovereign exposures in the trading book.
22	F	Domestic sovereign exposures in the trading book	Amount of domestic sovereign exposures in the trading book.
23	С	Operational risk	Total operational risk capital requirements. The capital charge should be converted to risk-weighted assets.
24	С	Settlement risk	Risk-weighted assets for settlement risk. The capital charge should be converted to risk-weighted assets.
25	С	Other Pillar 1 requirements	Risk-weighted assets for other Pillar 1 capital requirements according to national discretion. The capital charge should be converted to risk-weighted assets. If no such requirements exist, 0 should be entered.
B)	RWA effect	ts from Basel III definition of cap	ital and other national phase-in arrangements
31	С	RWA impact of applying future definition of capital rules	RWA impact of applying fully the phased-in national implementation of the Basel III definition of capital. If items which will be deducted in the fully phased-in treatment are currently risk-weighted, this amount should be reported as a negative number.
33	С	RWA impact of national phase- in arrangements for CVA if any	Incremental RWA impact of full implementation of the national CVA capital requirements. If the CVA capital requirements have already been fully phased-in, banks should report 0.
34	С	RWA impact of any other national phase-in arrangements	Incremental RWA impact of full implementation of the national implementation of Basel III capital requirements. If the capital requirements have already been fully phased-in or no phase-in agreements exist, banks should report 0.

4.2 Definition of capital

The "DefCap" worksheet collects the data necessary to calculate the definition of capital under the fully phased-in nationally implemented rules ("2022 national implementation", column D) and according to the fully phased-in Basel III standards ("2022 Basel III pure", column E).

The column headers in row 3 inform participating banks which of the columns they have to fill in depending on their jurisdiction. If one of the cells shows "No", then both the green and the yellow cells in that column can be left empty.

All data should be provided in the yellow and, where relevant, green cells and the data provided should reflect the application of the final Basel III standards or fully phased-in national rules and not the transitional arrangements (eg those set out in paragraphs 94 to 96 of the Basel III framework).

To be reported in the Basel III pure column of this worksheet instruments must comply with both the relevant entry criteria set out in the December 2010 Basel III standards and the 13 January 2011 press release on loss absorbency at the point of non-viability.

While some additional guidance on completing the worksheets is set out below, the worksheets themselves include detailed descriptions of each item to be provided and references to the relevant paragraphs of the Basel III standards. The instructions for completing the worksheets are therefore the combination of the Basel III standards, national rules, the descriptions included in the worksheets themselves and the additional guidance below.

4.2.1 Panel A: Provisions and expected losses

The data collected in panel A are the provisions and expected losses for exposures in the IRB portfolios, for exposures subject to the standardised approach and for exposures subject to the Basel I approach to credit risk, respectively. The "2022 national implementation" column provides two additional cells which should only be filled in by IRB banks in the EU where a separate calculation is conducted for defaulted and non-defaulted assets. As with all other sections, banks should contact their national supervisory agency if they are unclear as to how to complete this panel.

4.2.2 Panels B1, C1 and D1: Positive elements of capital

Panels B1, C1 and D1 collect the positive elements of capital (eg issued instruments and related reserves) that meet the criteria set out in the national rules and the Basel III standards, respectively, for inclusion in Common Equity Tier 1, Additional Tier 1 and Tier 2 capital.

Amounts are to be reported gross of all regulatory adjustments and follow the measurement approach that applies under the relevant accounting standards (ie reported amounts should equal the amounts reported on the balance sheet in respect of each item). This means that retained earnings and other reserves should include interim/final profits and losses to the extent that they are permitted or required to be included on the balance sheet under the prevailing accounting standards (eg if a bank reports its capital position for 30 June, this should be based on its balance sheet on 30 June, which will reflect profits earned and losses incurred up to and including 30 June). Similarly retained earnings and other reserves should exclude dividends only to the extent that these are required to be excluded from the relevant balance sheet under the prevailing accounting standards.

Banks must report data on shares and capital instruments issued by the parent of the consolidated group separately from data on shares and capital instruments issued by subsidiaries of the consolidated group. Shares and capital instruments issued by the parent of the consolidated group should be reported in rows 25, 66 and 84. These rows should not include any capital that has been issued out of subsidiaries of the group irrespective of whether the capital represents equity accounted instruments that appear in the consolidated accounts as minority interest or liability accounted instruments that appear as liabilities. The only exception to this rule is where capital has been raised by the parent of the consolidated group through an SPV that meets the criteria set out in paragraph 65 of the Basel III standards. Such amounts may be included in rows 66 and 84 as appropriate.

Shares and capital instruments issued by subsidiaries¹⁴ of the consolidated group that are held by third parties should be reported in rows 29, 67 and 85. The amount to be included in each cell should exclude amounts in accordance with the procedure set out in paragraphs 62 to 65 of the Basel III standards.

4.2.3 Panels B2, C2 and D2: Regulatory adjustments

Panels B2, C2 and D2 collect the data necessary to calculate the various regulatory adjustments required by paragraphs 66 to 89 of the Basel III standards and the related national rules. Set out below is some additional guidance on certain of the regulatory adjustments to supplement the information provided in

Subsidiaries includes all consolidated subsidiaries of the group, irrespective of whether they are fully owned or partially owned.

the relevant section of the Basel III standards, the related national rules and the description provided in the "DefCap" worksheet.

In addition to the regulatory adjustments under the fully phased-in national rules (column D) and the fully phased-in Basel III framework (column E), banks should also enter the marginal impact on risk-weighted assets if they would apply Basel III pure rather than the national implementation. For example, if a country is risk weighting a certain item while Basel III requires deduction, the relevant cell in column F should include the risk-weighted asset amount under the national rules as a **negative** number. Alternatively, if the national rules for a line item are exactly equivalent to the Basel III standard, banks should enter zero in the relevant cell of column F.

Cells D47 to D50 and D61 are only mandatory for banks in the EU and capture optional deductions for certain items which are subject to a 1,250% risk weighting treatment under the Basel III standards. For these items, the risk-weighted asset impact in column F is calculated automatically.

Furthermore, column D of rows 51, 62, 78 and 98 captures deductions according to national rules which are not based on the Basel III standards. The risk-weighted asset amount applicable under the Basel III framework if these items were not deducted should be entered in the relevant cell of column F (as a **positive** number).

4.2.4 Panel E: Investments in the capital of banking, financial and insurance entities that are outside the scope of regulatory consolidation and below the threshold for deduction

For investments in the capital of banking, financial and insurance entities that are outside the scope of regulatory consolidation and below the threshold for deduction, banks should report both the amount and the related risk-weighted asset amount in panel E. The risk-weighted asset amount should **also** be included in the relevant item on the "Requirements" worksheet.

4.2.5 Capital issued out of subsidiaries to third parties (paragraphs 62 to 65)

The "DefCap-MI" worksheet can be used to calculate the amount of each subsidiary's capital that will be permitted to be included in the consolidated capital of the group and the amount that will be excluded due to the application of paragraphs 62 to 65 of the Basel III standards. Annex 4 of the Basel III standards sets out an illustrative example of the treatment of capital issued out of subsidiaries.

The amounts reported in respect of each consolidated subsidiary that has issued capital instruments to third parties should reflect the application of the final standards set out in paragraphs 49 to 90 of the Basel III standards to that subsidiary and not the transitional arrangements set out in paragraphs 94 to 96.

For each subsidiary that has issued capital to third parties, the relevant data can be included in the green cells in the "DefCap-MI" worksheet. A separate column can be completed for each subsidiary. The aggregated amount to be included in consolidated capital in respect of all consolidated subsidiaries of the group is calculated automatically in cells D29, D30 and D31. These amounts should be reported in the "DefCap" worksheet in cells E32, E70 and E88 respectively. Alternatively, banks can also calculate the relevant numbers outside the reporting template without filling in the "DefCap-MI" worksheet.

4.3 Impact of the revised floors including the new standardised approach to operational risk

The "Floors" worksheet collects some additional data which, together with the data collected on other worksheets, will allow the Committee to assess the impact of the proposed capital floors based on the standardised approaches and the new standardised approach to operational risk. ¹⁵

4.3.1 Credit risk (panel A)

Row	Column	Heading	Description
1)	Counterpar	ty credit risk	
8	С	Counterparty credit risk exposures (not including CVA charges or charges for exposures to CCPs); Using the proposed new standardised approach to credit risk and SA-CCR for counterparty credit risk; Exposures currently subject to the standardised approach	Banks should report the risk-weighted assets if the proposed new standardised approach to credit risk and the SA-CCR for counterparty credit risk were applied to the set of exposures for which current risk-weighted assets have been reported in cell D7 of the "Requirements" worksheet.
8	D	Counterparty credit risk exposures (not including CVA charges or charges for exposures to CCPs); Using the proposed new standardised approach to credit risk and SA-CCR for counterparty credit risk; Exposures currently subject to IRB approaches	Banks should report the risk-weighted assets if the proposed new standardised approach to credit risk and the SA-CCR for counterparty credit risk were applied to the set of exposures for which current risk-weighted assets have been reported in cell E7 of the "Requirements" worksheet.

2) Securitisation

Data are collected both for the current securitisation framework and for the revised hierarchy of approaches if the *Revisions to the securitisation framework*¹⁶ are applied. The set of exposures reported in cells C16 to C18 should be the same for which risk-weighted assets are shown or reported in cells C12 to C14. Similarly, the set of exposures reported in cell C18 should be the same for which risk-weighted assets are shown or reported in cells C12 to C14.

14	С	Current risk-weighted assets; of which: RWA equivalent of exposures subject to deduction treatment	Banks should report the risk-weighted asset equivalent of securitisation exposures which are subject to a deduction treatment at the reporting date and, therefore, not included in the risk-weighted assets reported in row 10 of the "Requirements" worksheet. If no such exposures exist, banks should enter 0 in this cell.
16	С	Risk-weighted assets under the revised hierarchy of approaches; Standardised approach	Banks should report the risk-weighted assets for exposures which will become subject to the standardised approach under the revised hierarchy of approaches.

Basel Committee on Banking Supervision, Capital floors: the design of a framework based on standardised approaches - consultative document, December 2014, www.bis.org/bcbs/publ/d306.htm.

Basel Committee on Banking Supervision, Revisions to the securitisation framework, December 2014, www.bis.org/bcbs/publ/d303.htm.

Row	Column	Heading	Description
17	С	Risk-weighted assets under the revised hierarchy of approaches; External ratingsbased approach	Banks should report the risk-weighted assets for exposures which will become subject to the external ratings-based approach under the revised hierarchy of approaches.
18	С	Risk-weighted assets under the revised hierarchy of approaches; Internal ratingsbased approach	Banks should report the risk-weighted assets for exposures which will become subject to the internal ratings-based approach under the revised hierarchy of approaches.
19	С	Risk-weighted assets when applying the standardised approach to all exposures	Banks should report the risk-weighted assets for exposures if the new standardised approach is applied to all securitisation exposures.
3)	CCPs		
24	С	Capital charge for trade exposures (including client cleared trades); Using SA-CCR	Banks should report the risk-weighted asset amount if SA-CCR was applied to calculate the capital charge for trade exposures, including client cleared trades. The set of exposures for which risk-weighted assets are reported in cell C24 should be the same as the one for which risk-weighted assets are shown in cell C23.
4)	CVA		
29	С	CVA capital charge (risk- weighted asset equivalent); Using SA-CCR; RWA	Banks should report the risk-weighted asset amount if SA-CCR was applied to calculate the CVA capital charge. The set of exposures for which risk-weighted assets are reported in cell C29 should be the same as the one for which risk-weighted assets are shown in cell C28.
29	D	CVA capital charge (risk- weighted asset equivalent); Using SA-CCR; Externally rated exposures	Banks should report the exposures for which risk-weighted assets are reported in cell C29 and which are externally rated.
29	E	CVA capital charge (risk- weighted asset equivalent); Using SA-CCR; Unrated exposures	Banks should report the exposures for which risk-weighted assets are reported in cell C29 and which are unrated.

4.3.2 Operational risk (panel B)

Panel B1 collects data on the current approaches to operational risk, while panel B2 collects data aimed at revising the standardised approaches on operational risk and introducing the new standardised approach (NSA), as set out in the consultative document issued in October 2014.

The data necessary for building the business indicator (BI), the proxy indicator that will replace the gross income in the NSA, had been collected in the QIS launched in July 2014. However, some of the items requested in the previous exercise may have been misunderstood by the respondents, in all likelihood because the business indicator and its structure had not yet been made public. Furthermore, during the consultation of the proposal on the NSA, a few respondents requested that some of the business indicator components were more clearly defined and/or slightly amended.

In light of the above, panel B2 in this QIS collects again the data necessary for building the business indicator, although for a shorter period, and the instructions for filling its items are enriched or slightly amended in light of comments received. The information that will be collected will serve to confirm or fine tune the current proposal on the NSA for operational risk and may be also useful for floors calculation, in case these will be based on NSA figures.

Panel B2 consists of two parts: balance sheet and other items (panel B2a) and income statement (panel B2b). These panels should be completed by all the banks on a best-effort basis, regardless of how banks currently calculate operational risk capital (ie AMA, TSA/ASA, BIA).

As for other parts of the reporting template, the data should be reported on a group-wide consolidated basis (ie data should be consolidated for all entities which are consolidated by the bank for risk-based regulatory purposes).

Data should be reported in the reporting currency and unit as specified in the "General Info" worksheet as of end-December of the reference years. For each reference year, year T refers to the fiscal years closed in the period from end-September T to end-June T+1. For example the reference year 2012 encompasses all the fiscal years closed between end-September 2012 and end-June 2013.

Current approaches (panel B1)

Row	Column	Heading	Description			
•	If a particular approach to operational risk is not applicable or not used by a bank, risk-weighted assets should be reported as 0 .					
35	С	Basic indicator approach	Risk-weighted assets for operational risk of the parts under the basic indicator approach. The capital charge should be converted to risk-weighted assets.			
36	С	Standardised approach	Risk-weighted assets for operational risk of the parts under the standardised approach. The capital charge should be converted to risk-weighted assets.			
37	С	Alternative standardised approach	Risk-weighted assets for operational risk of the parts under the alternative standardised approach. The capital charge should be converted to risk-weighted assets.			
38	С	Advanced measurement approach	Risk-weighted assets for operational risk of the parts under the advanced measurement approach. The capital charge should be converted to risk-weighted assets.			

New standardised approach (panel B2)

Row	Column	Heading	Description	
2a)	2a) Balance sheet and other items			
Panel 2a d	collects inform	nation on specific items of the Bala	nce Sheet and a few other items.	
42	C–G	Total assets	Total on-balance sheet assets.	
43	C–G	Interest-earning assets	Total on-balance sheet assets generating interest income.	
44	C–G	Interest-bearing liabilities	Total on-balance sheet liabilities bearing interest expenses.	

Row	Column	Heading	Description	Typical sub-items			
2b)	Income statement						
Panel 2b	nel 2b collects information on specific items of the income statement.						
47	C-G	Gross income	Gross income as defined in paragraph 650 of the Basel II framework. The definition used for regulatory purposes or as defined by the relevant national supervisor should be adopted (for instance, in the EU the "Relevant Indicator" definition should be used).				

Row	Column	Heading	Description	Typical sub-items
48	C-G	Interest income	Interest income coming from all financial assets, both primary financial instruments (included either in trading or non-trading books) and hedge accounting derivatives, as well as other interest income (eg from financial leasing).	Interest income from Loans and Advances, Available For Sales, Held to Maturity. Fair Value Option, Held for Trading Interest income from hedge accounting derivatives Other interest income
49	C-G	Interest expenses	Interest expense coming from all financial liabilities, both primary financial instruments (included either in trading or non-trading books) and hedge accounting derivatives, as well as other interest expenses (eg from financial leasing).	Interest expenses from deposits Interest expenses from debt securities issued Interest expenses from hedge accounting derivatives Other interest expenses
50	C-G	Fee and commission income	Income received for providing fee-based advices and services referring to both on-balance and off-balance sheet activities. The item includes income received by the bank as outsourcer of financial services.	Fee and commission income from: Securities (issuance/ origination or reception/ transmission/execution of orders on behalf of customers) Clearing and settlement Asset management Custody Fiduciary transactions Payment services Structured finance Servicing from securitisation activities Loan commitments and guarantees given Foreign transactions
51	C-G	Fee and commission expenses	Expenses paid for receiving fee- based advices and services referring to both on-balance and off-balance sheet activities. The item includes outsourcing fees paid by the bank for the supply of financial services (eg clearing and settlement, custody, etc) but not outsourcing fees paid for the supply of non-financial services (ie logistical, IT, human resources)	Fee and commission expenses from: Clearing and settlement Custody Servicing fees for securitisation activities Loan commitments and guarantees received Foreign transactions
52	C–G	Net profit (loss) on financial operations (trading book)	To distinguish trading from non- trading books items, the same criteria as those adopted to fill	Net gains/losses on financial assets and liabilities held for trading (derivatives, debt

Row	Column	Heading	Description	Typical sub-items
53	C-G	Net profit (loss) on financial operations (non-trading book)	the TB worksheets should be followed (see Section 7).	securities, equity securities, loans and advances, short positions, other assets and liabilities) Net gains/losses on financial assets or liabilities measured at fair value through profit or loss Realised net gains/losses on financial assets and liabilities not measured at fair value through profit or loss (available for sale financial assets, loans and advances, held to maturity investments, financial liabilities measured at amortised cost) Net gains and losses from hedge accounting Net exchange differences
54	C-G	Other operating income	Income from ordinary banking operations not classified in other business indicators' items but of similar nature.	Rental income from investment properties Income from operating leasing Gains from non-recurrent assets and disposal group classified as held for sale not qualifying as discontinued operations (IFRS 5.37)
55	C–G	Other operating income; of which: due to operating leasing	Income from operating leasing.	Indicate the amount of other operating income that is due to operating leasing, according to the relevant accounting criteria (eg IFRS)
56	C-G	Other operating expenses	i) ordinary banking operations not classified in other business indicators' items but of similar nature; ii) operational risk events (not provisioned for in previous financial years)	Expenses for operating leasing Losses from non-recurrent assets and disposal group classified as held for sale not qualifying as discontinued operations (IFRS 5.37) Direct charges to the P&L and costs incurred as a consequence of operational risk events (eg fines, penalties and litigation settlements), which have not bee provisioned for in previous financial years
57	C-G	Other operating expenses; of which: due to operating leasing	Expenses from operating leasing.	Indicate the amount of other operating expenses that is due to operating leasing, according to the relevant accounting criteria (eg IFRS)
58	C-G	Dividend income	Dividend income from investment in stocks and funds not consolidated in the bank's financial statements, including that from non-consolidated subsidiaries, associates and joint ventures.	

The following sub-items should not contribute to any of the items requested in panel B2b:

- Income and expenses from insurance or reinsurance business;
- Premium paid and reimbursement/payment received for insurance or reinsurance policies purchased;
- Administrative expenses: staff expenses (including salaries, pension and similar benefits), outsourcing fees paid for the supply of non-financial services (ie logistical, IT, human resources), other administrative expenses (including expenses for IT, utilities, telephone, travel, office supplies, postage etc);
- Recovery of administrative expenses, including recovery of payments on behalf of customers (eg taxes debited to customers);
- Expenses on share capital repayable on demand;
- Net gains/losses on derecognition of financial assets, non-financial assets, liabilities not measured at fair value through profit or loss;
- Depreciation/amortisation (eg on properties, tangible assets, intangible assets);
- Provisions/reversal of provisions (eg on pensions, commitments and guarantees given, legal issues);
- Impairment/reversal of impairment (eg on financial assets, non-financial assets, investments in subsidiaries, joint ventures and associates);
- Negative goodwill recognised in profit or loss;
- Share of the profit or loss of investments in subsidiaries, joint ventures and associates;
- Income tax, corporate tax (tax based on profits, including current tax and deferred tax).

5. Leverage ratio

5.1 Introduction

The "Leverage Ratio" worksheet collects data on the exposure measure of the Basel III leverage ratio (the denominator of the ratio) as defined by the Basel III leverage ratio framework and the *Frequently asked questions on the Basel III leverage ratio framework* (referred to as "the FAQs on the Basel III leverage ratio framework").¹⁷

As for other parts of the reporting template, exposures are to be reported in the worksheet on a group-wide consolidated basis for all entities which are consolidated by the bank for risk-based regulatory purposes.

Yellow cells are fundamental to the calculation of the Basel III leverage ratio based on the design agreed by the Group of Governors and Heads of Supervision on 12 January 2014 and will serve as the basis for testing during the parallel run period. The yellow cells are in (i) panel A, which covers on-balance sheet items; (ii) panel B, which covers the add-on for potential future exposure for derivatives calculated in accordance with paragraphs 19 to 21 of the Basel III leverage ratio framework and off-

Basel Committee on Banking Supervision, *Frequently asked questions on the Basel III leverage ratio framework*, www.bis.org/publ/bcbs293.htm.

balance sheet items calculated in accordance with paragraph 39 of the Basel III leverage ratio framework; (iii) panel E, which includes data on the offsetting of credit derivatives in accordance with paragraphs 29 to 31 of the Basel III leverage ratio framework.

The green cells collect additional information necessary to monitor the Basel III leverage ratio and its components during the transition period. Green cells are in (i) panels A, B, and E as described above; (ii) panel C, which provides an additional breakdown of on- and off-balance sheet exposures, according to their risk weights under the Basel II framework; ¹⁸ (iii) panel D, which allows for a reconciliation of accounting standards; (iv) panel G which requests data on alternative methods for calculating derivative exposures; (v) panel H, which requests additional data on cash variation margin associated with derivative transactions subject to *currency mismatches*; (vi) panel I, which requests additional data on initial margin associated with centrally cleared derivative transactions; (vii) panel J, which provides additional data for the purposes of the categorisation of business models.

Data on the capital measure of the Basel III leverage ratio (the numerator of the ratio) are collected in the "General Info" and "DefCap" worksheets.

The Basel III leverage ratio standards ensure consistency between the capital and exposure measures in the design of the leverage ratio, and paragraph 16 of the Basel III leverage ratio framework by stating that any deductions from regulatory capital may also be made from the exposure measure. However, when reporting data for the "Leverage Ratio" worksheet, banks should not make these deductions from the exposure measure as these will be made during the calculation phase, in panel F.

The worksheet should be compiled on a quarterly basis ¹⁹ by including end-of-quarter exposures (cf Basel III leverage ratio framework, paragraph 53). The data for the most recent quarter, ending as of the reporting date, should be entered in columns J through N (labelled "Reporting date"); the data for the preceding quarter should be entered in columns D through H (labelled "Previous quarter").

5.2 On-balance sheet items (panel A)

In panel A for on-balance sheet items, there are four columns for the exposure value of derivatives, securities financing transactions (SFTs) and other assets. The first three columns require, respectively, the accounting value, the gross value, and – for SFT and derivatives only – the counterparty credit risk exposure according to the Basel II framework. The fourth column applies to SFT exposures only and asks for the adjusted gross SFT assets.

5.2.1 Accounting values as reported in the banks' financial statements

Column D (and J) requires data as reported in the banks' financial statements prepared in accordance with the applicable accounting standards. Data in these columns should correspond to figures as reported in the financial statements (considering the regulatory scope of consolidation). These data should be net of specific provisions and valuation adjustments and include the effects of balance sheet offsetting as a result of netting agreements and credit risk mitigation only when permitted under the applicable accounting standards.

¹⁸ References to the Basel II framework include the July 2009 Basel II enhancements.

¹⁹ Since the Basel III monitoring exercise is carried out on a semiannual basis, each exercise will collect data covering the two quarters included in the relevant six-month period.

Derivatives

Rows 10, 11 and 12 collect data on the positive fair values of derivatives, as reported on the bank's financial statement, which may reflect the effect of balance sheet offsetting as a result of netting agreements and credit risk mitigation only when permitted under the applicable accounting standards.

Securities financing transactions (SFT)²⁰

Rows 16 and 17 collect data on the on-balance sheet amounts for SFTs, as reported in accordance with the applicable accounting standards separating out those agent transactions eligible for the exceptional treatment as set out in paragraphs 36 and 37 of the Basel III leverage ratio framework from all other SFT assets. Amounts may reflect the effect of balance sheet offsetting as a result of netting agreements and credit risk mitigation only when permitted under the applicable accounting standards.

5.2.2 Gross values

Column E (and K) requires data to be entered using the sum of accounting values (net of specific provisions and valuation adjustments), assuming no accounting netting or credit risk mitigation effects (ie gross values).²¹ Items that are not eligible for accounting netting or subject to credit risk mitigation should be the same as those reported in column D (and J).

Derivatives

Rows 10, 11 and 12 include gross value of **all** derivative exposure amounts, ²² assuming no accounting netting and no credit risk mitigation effects.

The amount of any derivatives collateral provided other than initial margin for client-cleared derivative transactions with a *qualifying* CCP (QCCP), and eligible cash variation margin as defined in paragraphs 25 and 26 of the Basel III leverage ratio framework, where the provision of that collateral has reduced the value of the balance sheet under the applicable accounting framework should be reported in row 21. Similarly, the receivable assets for eligible cash variation margin *provided* in derivative transactions according to paragraphs 25 and 26 of the Basel III leverage ratio framework, taking into account the clarification regarding the interpretation of the *currency of settlement* requirement as set out in Q1 of the FAQs on the Basel III leverage ratio framework are to be reported in row 22 if the bank is required under the applicable accounting standard to recognise these receivable assets. Initial margin *provided* to the QCCP as a result of client-cleared derivative transactions with a QCCP where the bank acts as a clearing member and exempts such initial margin from the leverage ratio exposure measure in accordance with paragraph 27 of the Basel III leverage ratio framework is to be reported in row 23 (note additional memo items regarding initial margin in panel I of the template).

SFT

Rows 16 and 17 require SFT assets to be reported with no recognition of the accounting netting of (cash) payables against (cash) receivables as currently permitted under the applicable accounting standards separating out those agent transactions eligible for the exceptional treatment as set out in paragraphs 36 and 37 of the Basel III leverage ratio framework from all other SFT assets.

²⁰ SFTs as defined by the Basel II framework include transactions such as repurchase agreements, reverse repurchase agreements, security lending and borrowing, and margin lending transactions, where the value of the transactions depends on the market valuations and the transactions are often subject to margin agreements.

For example, if a bank is permitted to net cash collateral against the net derivatives exposure amount under the applicable accounting standards (as reported in columns D and J), then the bank must take that cash collateral out (ie gross up its exposure amount) for purposes of columns E and K.

Including derivatives that are treated off-balance sheet under the applicable accounting standards.

If the applicable accounting standards require a bank to recognise the security received in an SFT as an asset, the asset amount must be reported in row 24.²³ Where SFTs are treated like a sale of asset under the bank's applicable accounting framework, the exposure amount for this SFT is to be reported in row 25 as if it had been treated like a financing transaction according to subparagraphs (i) and (ii) of paragraph 33 of the Basel III leverage ratio framework.

5.2.3 Counterparty credit risk exposure after applying the regulatory netting standards

Column F (and L) requires reporting of the counterparty credit risk exposure of derivatives²⁴ and SFTs after applying the regulatory netting standards based on the Basel II framework (not the accounting rules for netting as applied under column D (and J)).²⁵ Data should not include any other credit risk mitigation effects.²⁶

Derivatives

In row 9 banks are required to report the replacement cost of their derivative positions gross of cash variation margin and using Basel II netting standards, including positions resulting from paragraph 28 of the Basel III leverage ratio framework. Collateral received should not be netted against the (net) derivatives position.²⁷

If a derivatives transaction is not covered under a qualifying Basel II netting agreement, the derivative exposure amount should be reported on a gross basis, the same as the amount reported in column E (and K).

Row 13 asks for the amount of cash variation margin received and eligible for offsetting against the replacement cost portion of the derivative exposures according to paragraphs 25 and 26 of the Basel III leverage ratio framework, taking into account the clarification regarding the interpretation of the currency of settlement requirement as set out in Q1 of the FAQs on the Basel III leverage ratio framework.

Row 14 asks for the replacement cost portion of exempted trade exposures to a *qualifying* CCP (QCCP) from client-cleared derivatives transactions, where the bank acting as clearing member is not obligated to reimburse the client for any losses suffered due to changes in the value of its transactions in the event the QCCP defaults as set out in paragraph 27 of the Basel III leverage ratio framework.

SFT

For SFT, the counterparty credit risk exposure value is determined as the total fair value amount of securities and cash lent to a counterparty for all transactions included in a qualifying Basel II netting agreement²⁸, less the total fair value amount of cash and securities received from the counterparty for those transactions, floored at zero.²⁹

For example, under US GAAP, a security transferor must recognise a security received in a securities lending transaction as an asset if the transferor has the right to hypothecate the security but has not done so.

Including derivatives that are treated off-balance sheet under the applicable accounting standards.

Banks should always apply Basel II standards for netting (even if they are currently applying the Basel I framework).

This does not relate to the deduction of securities and cash received in a SFT according to paragraph 33(ii) of the Basel III leverage ratio framework.

A net derivatives position is the (positive) difference between positive and negative fair values of derivatives in a netting set.

A qualifying netting agreement is a netting agreement that meets the requirements under paragraphs 173 and 174 of the Basel II framework.

Banks should apply the following part of the formula as set forth in paragraph 33(ii), first bullet of the Basel III leverage ratio framework: $E^*=\max\{0,[(\sum E_i-\sum C_i]\}$.

Where no qualifying Basel II netting agreement is in place, the counterparty exposure value of SFT must be calculated on a transaction by transaction basis (that is, each SFT is treated as its own netting set) as set out in in paragraph 33(ii), second bullet of the Basel III leverage ratio framework.

These amounts have to be reported in rows 16 and 17 separating out those agent transactions eligible for the exceptional treatment as set out in paragraphs 36 and 37 of the Basel III leverage ratio framework from all other SFT assets.

5.2.4 Adjusted gross SFT assets

Row 17 of column G (and M) requires banks to report the adjusted gross SFT asset amounts for all SFTs other than the SFT agent transactions eligible for the exceptional treatment as set out in paragraphs 36 and 37 of the Basel III leverage ratio framework, according to paragraph 33 (i), second bullet of the Basel III leverage ratio framework.

5.2.5 Description of the data

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description
8	D, E, J, K	Derivatives	Non entry cells: Items in rows 10 to 12 provide a breakdown of derivatives and should sum to total derivatives.
8	F, L	Derivatives	Non entry cells: The replacement costs associated with all derivatives transactions as they enter the Basel III leverage ratio exposure measure.
9	F, L	Replacement cost associated with all derivatives transactions (gross of variation margin)	The replacement cost of derivatives using Basel II netting standards, with no recognition of collateral (whether cash o non-cash), see paragraphs 19, 20, 21, 23 and 28 of the Basel III leverage ratio framework. Derivatives traded OTC, on an exchange and through a CCF should all be included.
10	D, E, J, K	Credit derivatives (protection sold)	Positive fair values of written credit derivatives (ie where the bank is providing credit protection to a counterparty). Columns D and J must be reported on a net basis (ie reflecting the effect of netting agreements and credit risk mitigation when permitted under the applicable accounting standards); columns E and K must be reported on a gross basis.
11	D, E, J, K	Credit derivatives (protection bought)	Positive fair values of purchased credit derivatives (ie where the bank is buying credit protection from a counterparty). Columns D and J must be reported on a net basis (ie reflecting the effect of netting agreements and credit risk mitigation when permitted under the applicable accounting standards); columns E and K must be reported on a gross basis.
12	D, E, J, K	Financial derivatives	Positive fair values of financial derivatives (eg interest rates derivatives, FX and gold derivatives, equities derivatives, etc Columns D and J must be reported on a net basis (ie reflecting the effect of netting agreements and credit risk mitigation when permitted under the applicable accounting standards); columns E and K must be reported on a gross basis.

Row	Column	Heading	Description
13	F, L	Eligible cash variation margin offset against derivatives market values	Cash variation margin received eligible for offsetting against the replacement cost portion of the derivatives exposures according to paragraphs 25 and 26 of the Basel III leverage ratio framework, taking into account the clarification regarding the interpretation of the currency of settlement requirement as set out in Q1 of the FAQs on the Basel III leverage ratio framework.
14	F, L	Exempted CCP leg of client- cleared trade exposures (replacement costs)	The replacement cost portion of exempted trade exposures to a QCCP from client-cleared derivatives transactions, where the bank acting as clearing member is not obligated to reimburse the client for any losses suffered due to changes in the value of its transactions in the event that the QCCP defaults, see paragraph 27 of the Basel III leverage ratio framework.
15	D, E, J, K	Securities financing transactions	Non entry cells: Items in rows 16 and 17 provide a breakdown of SFTs and should sum to total SFTs.
15	F, L	Securities financing transactions	Non entry cells: Sum of counterparty credit risk exposure of SFT covered and not covered by eligible netting agreements, see paragraph 33(ii) of the Basel III leverage ratio framework. SFT traded OTC, on an exchange and through a CCP should all be included.
15	G, M	Securities financing transactions	Non entry cells: Sum of the adjusted gross SFT assets, see paragraph 33(i), second bullet of the Basel III leverage ratio framework. SFT traded OTC, on an exchange and through a CCP should
			all be included.
16	D, E, J, K	SFT agent transactions eligible for the exceptional treatment	Only SFT agent transactions where the bank acting as agent provides an indemnity or guarantee to a customer or counterparty that is limited to the difference between the value of the security or cash the customer has lent and the value of collateral the borrower has provided are eligible for this exceptional treatment, see paragraphs 36 and 37 of the Basel III leverage ratio framework.
			Columns D and J must be reported net of specific provisions and valuation adjustments and include the effects of netting agreements and credit risk mitigation only as per the relevant accounting standards.
			Columns E and K must be reported with no recognition of accounting netting of (cash) payables against (cash) receivables as permitted under relevant accounting standards.
			The securities lent in a SFT that remain recognised on the balance sheet must not be included here but in row 19.
			The value of securities received in a SFT that are recognised as an asset under the applicable accounting standard must be reported in row 24.
			The securities lent in a SFT that are derecognised due to a sales accounting transaction must not be included here but in row 25.
			SFT traded OTC, on an exchange and through a CCP should all be included.
16	F, L	SFT agent transactions eligible for the exceptional treatment	The exposure measure of eligible SFT agent transactions calculated by applying subparagraph (ii) of paragraph 33 of the Basel III leverage ratio framework.

Row	Column	Heading	Description
17	D, E, J, K	Other SFTs	SFTs other than SFT agent transactions reported in row 16. Columns D and J must be reported net of specific provision and valuation adjustments and include the effects of netting agreements and credit risk mitigation only as per the relevant accounting standards.
			Columns E and K must be reported with no recognition of accounting netting of (cash) payables against (cash) receivables as permitted under relevant accounting standards.
			The securities lent in a SFT that remain recognised on the balance sheet must not be included here but in row 19. The value of securities received in a SFT that are recognised as an asset under the applicable accounting standard must be reported in row 24.
			The securities lent in a SFT that are derecognised due to a sales accounting transaction must not be included here but in row 25.
			SFT traded OTC, on an exchange and through a CCP should all be included.
17	F, L	Other SFTs	The counterparty credit risk exposure of all SFTs other than SFT agent transactions reported in row 16 calculated according to subparagraph (ii) of paragraph 33 of the Basel III leverage ratio framework.
17	G, M	Other SFTs	The adjusted gross SFT assets of all SFTs other than SFT agent transactions reported in row 16 calculated according to subparagraph (i) of paragraph 33 of the Basel III leverage ratio framework.
18	E, K	Other assets	Non entry cells: Other assets as adjusted for the purposes of the Basel III leverage ratio.
19	D, E, J, K	Accounting other assets	Any other assets not specifically identified in any of the row 8 to 17 above (ie any other accounting assets not included under derivatives or SFT items, eg accounting receivables for cash variation margin provided where recognised under operative accounting framework, liquid assets as defined under the liquidity coverage ratio, failed and unsettled transactions). This includes any instrument (including cash) borrowed or lent through an SFT when it is reported on the accounting balance sheet.
20	E, K	Adjustments to accounting other assets for the purposes of the leverage ratio	Non entry cells: adjustments to accounting other assets for the purposes of the Basel III leverage ratio.
21	E, K	Grossed-up assets for derivatives collateral provided	The amount of any derivatives collateral provided where the provision of that collateral has reduced the value of the balance sheet assets under the applicable accounting framework, see paragraph 24 of the Basel III leverage ratio framework. However, initial margin for client-cleared derivative transactions with a qualifying CCP (QCCP) and eligible cash variation margin, as defined in paragraphs 25 and 26 of the Basel III leverage ratio framework, must not b included.

Row	Column	Heading	Description
22	E, K	Receivables for eligible cash variation margin provided in derivatives transactions	The receivables for eligible cash variation margin <i>provided</i> in derivatives transactions if the bank is required, under the applicable accounting standards, to recognise these receivables as an asset, see paragraphs 25 and 26 of the Basel III leverage ratio framework, taking into account the clarification regarding the interpretation of the currency of settlement requirement as set out in Q1 of the FAQs on the Basel III leverage ratio framework. The amount reported must also be included in the accounting other assets reported in row 19.
23	E, K	Exempted CCP leg of client- cleared trade exposures (initial margin)	The initial margin portion of exempted trade exposures to a QCCP from client-cleared derivatives transactions, where the bank acting as clearing member is not obligated to reimburse the client for any losses suffered due to changes in the value of its transactions in the event that the QCCP defaults, see paragraph 27 of the Basel III leverage ratio framework, taking into account the clarification regarding the interpretation of the currency of settlement requirement as set out in Q1 of the FAQs on the Basel III leverage ratio framework. The amount reported should also be included in the accounting other assets reported in row 19.
24	E, K	Securities received in a SFT that are recognised as an asset	Securities received in a SFT that are recognised as an asset under the applicable accounting standards and therefore included in row 19, see paragraph 33 of the Basel III leverage ratio framework.
25	E, K	Adjustments for SFT sales accounting transactions	The value of securities lent in a SFT that are derecognised due to a sales accounting transaction, see paragraph 34 of the Basel III leverage ratio framework.
26	E, K	Fiduciary assets	Fiduciary assets that are included in row 19 and that meet the IAS 39 criteria for derecognition and, where applicable, IFRS 10 for deconsolidation, see footnote 4 to paragraph 15 of the Basel III leverage ratio framework.
27	D, E, F, G, J, K, L, M	Totals	This is a non-data entry row.
29	F, G, L, M	Memo item: SFT exposures to QCCPs from client-cleared transactions	The SFT exposures to QCCPs from client-cleared SFT transactions, where the bank acting as clearing member is not obligated to reimburse the client for any losses suffered due to changes in the value of its transactions in the event that the QCCP defaults. These exposures must be included in rows 16 and 17.
31	E, K	Check row	This is a non-data entry row. It checks that the sum of single values included in the accounting other assets is lower or equal to the accounting other assets.

5.3 Derivatives and off-balance sheet items (panel B)

In panel B for derivatives and off-balance sheet items, there are three columns. The first two columns apply to derivatives solely and require, respectively, the potential future exposure (PFE) assuming no netting or credit risk mitigation, and the PFE with Basel II netting standards. The third column refers to both derivatives and off-balance sheet items and requires the notional values of those exposures.

The PFE and notional amounts excluded from panel B according to paragraph 27 of the Basel III leverage ratio framework must be reported in row 43.

5.3.1 Potential future exposure of derivatives measured using the current exposure method without the effect of Basel II netting

Column D (and J) requires potential future exposure of all derivatives, irrespective of whether or not they are centrally cleared, as well as exposures arising from the application of paragraph 28 of the Basel III leverage ratio framework, measured using the current exposure method (CEM) without the effect of Basel II netting.³⁰ Data in these columns only include the add-on for potential future exposure, since the total replacement cost is already captured in the on-balance sheet panel A. Data on the add-on for derivatives having a negative fair value (thus not reported in panel A) should be included as well.

When compiling the separate line items referred to as "Credit derivatives protection sold" the following criteria should be applied: For sold CDS subject to close out, the full text of paragraph 3 of the Annex of the Basel III leverage ratio framework should be applied; therefore, the add-on should be capped at unpaid premiums. For sold CDS not subject to close out, the treatment provided by the footnote in paragraph 3 of the Annex of the Basel III leverage ratio framework should not be applied and the add-on of 5% or 10% – depending on the nature (qualifying or non-qualifying) of the reference obligation – should always be calculated.³¹

Paragraph 3 of the Annex of the Basel III leverage ratio framework should be applied to all credit derivatives, whether they are included in the banking book or in the trading book.

Data should be reported gross of any netting agreement and credit risk mitigation effect (in line with the criteria for compiling column E (and K) in panel A). All banks should calculate the potential future exposure using the current exposure method, even if they do not apply such a method under the counterparty credit risk framework. For derivatives traded on an exchange or through a CCP the current exposure method is always applied, irrespectively of whether or not an exposure value of zero for counterparty credit risk is attributed under the Basel II framework.

Banks may choose to not include the individual add-on amount relating to a written credit derivative which is not offset by purchased protection with the characteristics described in Section 5.6, letter (c) of the present instructions.

5.3.2 Potential future exposure of derivatives with the effect of the Basel II netting

Column E (and K) requires potential future exposure of derivatives with the effect of the Basel II netting as set out in paragraphs 8 to 11 of the Annex of the Basel III leverage ratio framework. As noted above, banks should always apply the CEM netting standards as defined in the Basel II framework, irrespective of their actual approach to credit risk. Data should not include any credit risk mitigation effect other than the said Basel II netting.

The add-on for credit derivatives should be calculated according to the full text of paragraph 3 of the Annex of the Basel III leverage ratio framework, including the footnote. This implies that the add-on of sold CDS subject to close out should be capped at unpaid premiums, while the add-on for sold CDS not subject to close out should not be included.

Paragraph 3 of the Annex of the Basel III leverage ratio framework should be applied to all credit derivatives, whether they are included in the banking book or in the trading book.

See also Annex IV of the Basel II framework.

The footnote in paragraph 3 of the Annex of the Basel III leverage ratio framework states the following: "The protection seller of a credit default swap shall only be subject to the add-on factor where it is subject to closeout upon the insolvency of the protection buyer while the underlying is still solvent. Add-on should then be capped to the amount of unpaid premiums."

Banks may choose not to include the individual add-on amount relating to a written credit derivative which is not offset by purchased protection following the criteria described in Section 5.6, letter (c) of the present instructions.³²

When calculating the add-on for netted transactions (A_{Net} in the formula in paragraph 10 of Annex of the Basel III leverage ratio framework) and irrespective of the treatment of the collateral by the applicable accounting standards, banks must not recognise the collateral received in the calculation of the net replacement cost.

5.3.3 Notional amounts

Column F (and L) requires banks to report the notional amounts of derivatives and off-balance sheet items.

5.3.4 Description of the data

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description
B1)	Derivatives		
38	E, K	Potential future exposure for derivatives entering the leverage ratio exposure measure	Non entry cell: Provides for the total PFE entering the exposure measure related to derivative transactions according to paragraphs 19 to 28 of the Basel III leverage ratio framework.
39	E, K	Derivatives	Potential future exposure of derivatives when applying the current exposure method and Basel II netting standards.
39	D, F, J, L	Derivatives	Non entry cells: Items in rows 40 to 42 provide a breakdown of derivatives which should sum up to total derivatives.
40	D, F, J, L	Credit derivatives (protection sold)	Potential future exposure with no netting or CRM (columns D and J) or notional amount (columns F and L) for credit derivatives sold subject to close out, including the full treatment set out in paragraph 3 of the Annex of the Basel III leverage ratio framework (capping add-on at unpaid premiums). Where the effective notional amount of written credit
			derivatives is included in the exposure measure and not offset pursuant to paragraph 30 of Basel III leverage ratio framework, banks may choose to set the individual potential future exposure amounts relating to those written credit derivatives to zero.
41	D, F, J, L	Credit derivatives (protection bought)	Potential future exposure with no netting or CRM (columns D and J) or notional amount (columns F and L) of purchased credit derivatives (ie where the bank is buying credit protection from a counterparty)
42	D, F, J, L	Financial derivatives	Potential future exposure with no netting or CRM (columns D and J) or notional amount (columns F and L) of financial derivatives.

In these cases, where effective bilateral netting contracts are in place, and when calculating A_{Net}=0.4*A_{Gross}+0.6*NGR*A_{Gross}-A_{Gross} may be reduced by the individual add-on amounts (ie notionals multiplied by the appropriate add-on factors) which relate to written credit derivatives whose notional values are included as exposures of the Basel III leverage ratio. No adjustments should be made to NGR. Where effective bilateral netting contracts are not in place, the add-on can be set to zero in order to avoid double counting. See paragraph 31 of the Basel III leverage ratio framework.

Row	Column	Heading	Description
43	D, J	Exempted CCP leg of client- cleared trade exposures (potential future exposure)	Potential future exposure using the current exposure method and assuming no netting or CRM associated with exempted CCP leg of client-cleared trade exposures (potential future exposure fulfilling the exemption criteria laid down in paragraph 27 of the Basel III leverage ratio framework).
43	E, F, K, L	Exempted CCP leg of client- cleared trade exposures (potential future exposure)	Potential future exposure of derivatives when applying the current exposure method and Basel II netting standards (columns E and K), or notional amount (columns F and L) for exempted CCP leg of client-cleared trade exposures according to paragraph 27 of the Basel III leverage ratio framework.
B2)	Off-balance	sheet items	
45	F, L	Off-balance sheet items with a 0% CCF in the RSA; of which:	Off-balance sheet items that would be assigned a 0% credit conversion factor as defined in the standardised approach to credit risk in the Basel II framework. That is commitments that are unconditionally cancellable at any time by the bank without prior notice (UCC), or that effectively provide for automatic cancellation due to deterioration in a borrower's creditworthiness (see paragraph 83 of the Basel II framework and the footnote to this paragraph). Note that rows 46 and 47 do not sum up to row 45 since the latter includes commitments that effectively provide for automatic cancellation due to deterioration in a borrower's creditworthiness but that are not UCCs.
46	F, L	Unconditionally cancellable credit cards commitments	Credit cards commitments that are unconditionally cancellable at any time by the bank without prior notice (UCC) that would receive a 0% CCF under the standardised approach to credit risk. Credit card commitments that effectively provide for automatic cancellation due to deterioration in a borrower's creditworthiness but that are not UCC should not be included in this row.
47	F, L	Other unconditionally cancellable commitments	Other commitments that are unconditionally cancellable at any time by the bank without prior notice, that would receive a 0% CCF under the standardised approach to credit risk. Commitments that effectively provide for automatic cancellation due to deterioration in a borrower's creditworthiness but that are not UCC should not be included in this row.
48	F, L	Off-balance sheet items with a 20% CCF in the RSA	Off-balance sheet items that would be assigned a 20% credit conversion factor as defined in the standardised approach to credit risk (see paragraphs 83 and 85 of the Basel II framework and the footnote to paragraph 83).
49	F, L	Off-balance sheet items with a 50% CCF in the RSA	Off-balance sheet items that would be assigned a 50% credit conversion factor as defined in the standardised approach to credit risk (see paragraphs 83, 84(ii) and 84(iii) of the Basel II framework). This includes liquidity facilities and other commitments to securitisations incorporating the changes according to the Enhancements. That is the CCF for all eligible liquidity facilities in the securitisation framework is 50% regardless of the maturity.

Row	Column	Heading	Description
50	F, L	Off-balance sheet items with a 100% CCF in the RSA	Off-balance sheet items that would be assigned a 100% credit conversion factor as defined in the standardised approach to credit risk (see paragraphs 83(i), 83 (ii), 84 and 84(i) of the Basel II framework. This includes liquidity facilities and other commitments to securitisations incorporating the changes according to the Enhancements.
51	F, L	Total off-balance sheet items	This is a non-data entry row.
53	F, L	Check row	This is a non-data entry row. It checks that the unconditionally cancellable commitments do not exceed the off-balance sheet items with a 0% CCF.

5.4 On- and off-balance sheet items – additional breakdown of exposures (panel C)

Panel C provides an additional breakdown for on- and off-balance sheet exposures belonging to the banking book, according to the risk weights applied under the Basel II framework.³³

Banks adopting the standardised approach for credit risk should report each exposure according to the regulatory risk weight as provided by the Basel II framework (under the standardised approach or the securitisation framework). For banks adopting the internal ratings-based approach, for exposure (other than those for which specific regulatory risk weights are provided for – eg specialised lending exposures under the supervisory slotting criteria approach, securitisations exposures with an external credit assessment, equity exposures under the simple risk weight method, etc) belonging to each borrower grade, the risk weight should be derived by dividing the risk weighted exposure obtained from the risk-weight formula or the supervisory formula (for credit risk or securitisations exposures, respectively) by the EAD after recognition of eligible credit risk mitigation techniques. Under the internal ratings-based approach, exposures classified as in default should be excluded from the rows 60 to 67 and included in row 68.

Exposures deducted from the regulatory capital should be considered as being applied a 1250% risk weight. 36

The exposure value of on-balance sheet items (columns D and J) should correspond to the solvency-based value under the Basel II framework,³⁷ after recognition of eligible credit risk mitigation techniques (eg EAD for the internal ratings-based approach). Off-balance sheet items (columns E and K)

Transactions subject to the treatment for counterparty credit risk (see Annex 4 of the Basel II framework) should be included irrespective of whether they are classified in the banking or in the trading book.

For exposures supported by credit risk mitigation techniques implying the substitution of the risk weighting of the counterparty with the risk weighting of the guarantee (eg financial collateral under the simple approach), banks should refer to the risk weight after the substitution effect.

Banks currently adopting the Basel I framework should refer to the risk weights currently applied for the calculation of the credit risk capital requirement; for reporting purposes, the exposures should be mapped to the risk weights buckets as provided in this panel.

Deductions from the capital base under Annex 1a part C of the Basel II framework as well as regulatory adjustments under paragraphs 66 to 88 of the Basel III standards should not be included in panel C. Exposures for which the Basel II framework allows the option of being deducted or risk weighted (eg certain securitisation exposures) should be included in panel C with a 1250% risk weight even if they are deducted from the capital base.

³⁷ Or under the Basel I framework, if currently applied by a bank, in which case the bank should correspondingly apply the Basel I standards for netting.

should be reported as for their notional value multiplied by the regulatory CCF³⁸ under the Basel II framework.

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description
59	D, E, J, K	Total on- and off-balance sheet exposures belonging to the banking book (breakdown according to the effective risk weight):	This is a non-data entry row.
60	D, E, J, K	= 0%	Exposures with effective risk weight of 0%.
61	D, E, J, K	> 0 and ≤ 12%	Exposures with effective risk weights exceeding 0% but not more than 12%.
62	D, E, J, K	> 12 and ≤ 20%	Exposures with effective risk weights exceeding 12% but not more than 20%.
63	D, E, J, K	> 20 and ≤ 50%	Exposures with effective risk weights exceeding 20% but not more than 50%.
64	D, E, J, K	> 50 and ≤ 75%	Exposures with effective risk weights exceeding 50% but not more than 75%.
65	D, E, J, K	> 75 and ≤ 100%	Exposures with effective risk weights exceeding 75 but not more than 100%.
66	D, E, J, K	> 100 and ≤ 425%	Exposures with effective risk weights exceeding 100% but not more than 425%.
67	D, E, J, K	> 425 and ≤ 1250%	Exposures with effective risk weights exceeding 425% but not more than 1250%.
68	D, E, J, K	Defaulted exposures under the IRB approach	Exposures classified as in default under the internal ratings-based approach.

5.5 Reconciliation (panel D)

Panel D on reconciliation is a summary table that seeks to ensure the data is entered correctly and consistently. The reconciliation is between total accounting balance sheet exposures and total exposures after the effects of accounting netting (and other credit risk mitigation effects), or the effects of derivatives treated off-balance sheet under the applicable accounting standards and recognised with their gross values in panel A, have been eliminated. The adjusted non-netted values will provide a consistent comparison of exposures across accounting standards.

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description
74	D	Accounting total assets, previous quarter	Total assets following the relevant accounting balance sheet (considering the regulatory consolidation). The figure should be the same as the total value in cell D27.
74	J	Accounting total assets, reporting date	This is a non-data entry row. Total assets following the relevant accounting balance sheet (considering the regulatory consolidation) should be entered in cell C58 of the "General Info" worksheet. The figure should be the same as the total value in cell J27.

The applicable CCF should be based on the approach to credit risk used by the bank (eg standardised approach or internal ratings-based approach).

Row	Column	Heading	Description
75	D, J	Check row	This is a non-data entry row. It checks that the total assets figure in panel D is the same as reported in panel A.
76	D, J	Reverse out on-balance sheet netting	Enter the amount of on-balance sheet netting (ie netting of loans against deposits) following the relevant accounting standards. This figure should equal the difference between the gross (column E (and K)) and the netted figures (column D (and J)) in panel A for the other assets (row 19).
77	D, J	Reverse out derivatives netting and other derivatives adjustments	Enter the amount of derivatives netting and the gross value of derivatives that are treated off-balance sheet and therefore included in column E (and K), where applicable, following the relevant accounting standards. This figure should equal the difference between the gross (column E (and K)) and the accounting figures (column D (and J)) in panel A for derivatives (row 8).
78	D, J	Reverse out SFT netting	Enter the amount of netting related to SFT following the relevant accounting standards. This figure should equal the difference between the gross (column E (and K)) and the netted figures (column D (and J)) in panel A for SFT (row 15).
79	D, J	Reverse out other netting and other adjustments	Adjustment to the accounting other assets for the purpose of the Basel III leverage ratio. This should correspond to row 20.
80	D, J	Totals	This is a non-data entry row.
81	D, J	Check row	This is a non-data entry row. It checks that the total assets figure calculated in row 80 is the same as the total of gross values in panel A (row 27).

5.6 Adjusted notional exposures for written credit derivatives (panel E)

In panel E for the additional treatment for written credit derivatives exposure³⁹, there are three columns.

- (a) Column D (and J) requires the effective notional amounts⁴⁰ for written credit derivatives, capped at maximum potential loss as defined in paragraph 30 of the Basel III leverage ratio framework.
- (b) Column E (and K) requires the effective notional amounts capped at maximum potential loss, for credit derivatives bought on the same reference name as the written credit derivatives.
- (c) Column F (and L) requires the effective notional amounts capped at maximum potential loss, for credit derivatives bought on the same reference name, where in the addition the maturity of the protection bought is equal to or greater than the maturity of the protection sold.

Reference names are considered the same if the conditions in footnote 14 of the Basel III leverage ratio framework are met.

Assuming Bank A has sold credit protection on \$100 of Corporate X debt for five years, and purchased credit protection on the same debt through the following transactions: (i) \$40 for five years; (ii) \$40 for two years; (iii) \$20 for six months, and assuming Bank A has not entered in other credit derivatives transactions, the notional amounts of credit protection written and purchased under the criteria described above are the following:

Both credit derivatives belonging to the banking book and to the trading book should be reported.

That is reflecting the true exposure of contracts that are leveraged or otherwise enhanced by the structure of the transaction as provided in footnote 13 of the Basel III leverage framework.

	Capped notional amount	Capped notional amount (same reference name)	Capped notional amount (same reference name with no maturity mismatch)
Credit derivatives (protection sold)	100		
Credit derivatives (protection bought)	100	100	40

Row	Column	Heading	Description
87	D, J	Credit derivatives:	This is a non-data entry row.
88	D, J	Credit derivatives (protection sold)	Capped notional value of written credit derivatives (ie where the bank is providing credit protection to a counterparty) as set out in paragraph 30 of the Basel III leverage ratio framework.
89	D, J	Credit derivatives (protection bought)	Capped notional value of purchased credit derivatives (ie where the bank is buying credit protection from a counterparty) as set out in paragraph 30 of the Basel III leverage ratio framework.
89	E, K	Credit derivatives (protection bought)	Capped notional value of purchased credit derivatives (ie where the bank is buying credit protection from a counterparty) as set out in paragraph 30 of the Basel III leverage ratio framework, on the same underlying reference names as those credit derivatives written by the bank as defined in footnote 14 of the Basel III leverage ratio framework. Hence, the value should not be greater than the value entered in cell D89 (and J89) for each reference name.
89	F, L	Credit derivatives (protection bought)	Capped notional value of purchased credit derivatives (ie where the bank is buying credit protection from a counterparty) on the same underlying reference names as those credit derivatives written by the bank, where the maturity of the purchased protection is equal to or greater than the maturity of the sold protection. Hence, the value should not be greater than the value entered in cell E89 (and K89) for each reference name.
90	E, F, K, L	Credit derivatives (protection sold less protection bought)	This is a non-data entry row. It calculates the difference between written and purchased credit derivatives on the same underlying reference names, for each of the two hypotheses for the offsetting as described above.
92	D, J	Check row	This is a non-data entry row. It checks that the notional amount of written credit derivatives is the same as or less than that in panel B.
93	D, J	Check row	This is a non-data entry row. It checks that the notional amount of purchased credit derivatives is the same as or less than that in panel B.
94	D, E, F, J, K, L	Check row	This is a non-data entry row. It checks that the notional amount of purchased credit derivatives for each of the two hypotheses for the offsetting as described above is consistently filled-in.

5.7 Calculation of the Basel III leverage ratio (panel F)

Panel F provides with the calculation of the Basel III leverage ratio, on the basis of the exposures data reported in the "Leverage Ratio" worksheet as well as of other relevant data as reported in the "General Info" and the "DefCap" worksheets (ie Tier 1 capital, regulatory adjustments).

TI (II '		1			
The following	i tahle nrovides	a description	of the data	to he en	tered in each row.
THE TOHOWHING					

Row	Column	Heading	Description
100	D, J	Tier 1 capital	This is a non-data entry row. It includes the amount of Basel III Tier 1 capital as reported in cell E71 of the "General Info" worksheet (numerator of the Basel III leverage ratio).
101	D, J	Total exposures	This is a non-data entry row. It calculates the total exposures to be included in the denominator of the Basel III leverage ratio (before the deduction of regulatory adjustments).
101	E, K	Data complete	This is a non-data entry row. It checks that all required exposure amounts entering the Basel III leverage ratio calculation are reported in previous panels.
102	D, J	Regulatory adjustments	This is a non-data entry row. It includes the amount of regulatory adjustments from Tier 1 as reported in the "DefCap" worksheet.
103	D, J	Total exposures for the calculation of the leverage ratio	This is a non-data entry row. It calculates the total exposures to be used for calculating the Basel III leverage ratio.
104	D, J	Basel III leverage ratio	This is a non-data entry row. It calculates the Basel III leverage ratio on the basis of the previous values.

5.8 Alternative methods for derivative exposures (panel G)

Panel G requests for additional data on alternative methods for calculating the Basel III leverage ratio exposure value for derivative transactions. These methods are represented by the *Standardised approach* for measuring counterparty credit risk exposures as published by the Committee in March 2014 (hereafter referred to as "SA-CCR without modification" and the "SA-CCR document"), ⁴¹ and a modified version of the SA-CCR according to the criteria laid down in the current instructions (hereafter referred to as "modified SA-CCR").

The scope of derivatives transactions for the calculations in panel G is the same as the one used for the calculation of the Basel III leverage ratio exposure measure (ie *excluding* the CCP-leg of clearing members' client-cleared trade exposures to a QCCP as set out in paragraph 27 of the Basel III leverage ratio framework).

5.8.1 Counterparty credit risk exposure according to the SA-CCR without modification

Column D (and J) requires the counterparty credit risk exposure of derivative transactions already included in panels A and B to be calculated according to the SA-CCR without modification.

Replacement cost

In row 110 banks are required to report the replacement cost of their derivative transactions calculated according to paragraphs 130 to 145 of the SA-CCR document.

Basel Committee on Banking Supervision, *The standardised approach for measuring counterparty credit risk exposures*, March 2014, www.bis.org/publ/bcbs279.htm.

Further rows 111 to 114 request data on respectively:

- (a) the impact on the replacement cost of the bank's derivative transactions of respectively the:
 - (i) collateral *provided* by the bank and included in rows 18 or 21 (cells D111 and J111); and
 - (ii) non-cash collateral *provided* by the bank and included in rows 18 or 21 (cells D112 and J112);
- (b) receivables for cash collateral provided and taken into account in C or NICA according to the SA-CCR document (cells D113 and J113);
- (c) collateral *provided* that is both included in rows 18 or 21 and taken into account in C or NICA according to the SA-CCR document (cells D114 and J114).

Potential future exposure

In row 117 banks are required to report the potential future exposure of their derivative transactions calculated according to paragraphs 146 to 187 of the SA-CCR document.

5.8.2 Counterparty credit risk exposure according to a modified SA-CCR

Columns E and K require the counterparty credit risk exposure of derivative transactions already included in panels A and B to be calculated according to a modified version of the SA-CCR according to the criteria laid down below.

Replacement cost

Cells E110 and K 110 are non-entry cells (relevant data are gathered from panel A).

Potential future exposure

Data on PFE add-on have to be reported according to two alternative criteria:

- in row 117 banks are requested to calculate and report the PFE of all their derivative transactions, including those for which variation margin is exchanged, by applying the maturity factors for unmargined transactions provided in the SA-CCR document (see paragraphs 154, 155 and 164 of the SA-CCR document). Furthermore, the PFE multiplier must be set to one for all derivative transactions. All other criteria provided by the SA-CCR document, including the methodology for offsetting long and short positions within each hedging set and for aggregating the trade-level add-ons within each asset class have to be applied. This row is the the sum of
 - PFE of all margined netting sets as requested in row 115, where banks are required to calculate the PFE of all their derivative transactions, including those for which variation margin is exchanged, by applying the maturity factors for unmargined transactions provided in the SA-CCR document (see paragraphs 154, 155 and 164); furthermore, the PFE multiplier must be set to one for all derivative transactions and all other criteria provided by the SA-CCR document, including the methodology for offsetting long and short positions within each hedging set and for aggregating the trade-level add-ons within each asset class have to be applied;
 - PFE of all *unmargined* netting sets as requested in row 116, where banks are required to calculate the PFE of all their derivative transactions, including those for which variation margin is exchanged, by applying the maturity factors for unmargined transactions provided in the SA-CCR document (see paragraphs 154, 155 and 164); furthermore, the PFE multiplier must be set to one for all derivative transactions and all other criteria provided by the SA-CCR document, including the methodology for

offsetting long and short positions within each hedging set and for aggregating the trade-level add-ons within each asset class have to be applied.

• in row 118 banks are requested to calculate and report the sum of gross add-ons without any recognition of netting and by applying the maturity factors for unmargined transactions throughout. To calculate the gross add-ons, banks are requested, for each derivative transaction, to multiply (i) the absolute value of the supervisory delta (paragraph 159 of the SA-CCR document); (ii) the adjusted notional amount (paragraphs 157 and 158 of the SA-CCR document); (iii) the supervisory factor (paragraph 183 of the SA-CCR document); and (iv) the unmargined maturity factor (first bullet of paragraph 164 of the SA-CCR document). The resulting trade-level gross add-ons are then summed up. This calculation is equivalent to replacing all the aggregation formulas specified in paragraphs 168 to 179 of the SA-CCR document by simple gross sums.

5.8.3 Description of the data

Row	Column	Heading	Description
110	D, J	Replacement cost (RC)	Replacement cost (RC) of derivative transactions calculated according to paragraphs 130 to 145 of the SA-CCR document.
110	E, K	Replacement cost (RC)	Non-entry cells (relevant data are gathered from panel A).
111	D, J	Impact on the RC of collateral provided by the bank and included in rows 18 or 21	The difference between un-modified RC under SA-CCR, which includes collateral provided by the bank, and the RC without taking into account any collateral provided by the bank that is already included in rows 18 or 21. This row is introduced to correct for double counting of collateral provided.
112	D, J	Impact on the RC of non- cash collateral provided by the bank and included in rows 18 or 21	Similar to row 111 but restricting the impact to non-cash collateral. That is, the difference between RC under SA-CCR which includes collateral provided by the bank, and the RC without taking into account any non-cash collateral provide by the bank that is already included in rows 18 or 21.
113	D, J	Receivables for cash collateral provided and taken into account in C or NICA	Receivables for cash collateral provided by the bank that is both included in row 18 and taken into account in the calculation of C or NICA in SA-CCR. The eligibility criteria of the Basel III leverage ratio framework are not applied, ie it includes any cash collateral or cash variation margin whether it meets the criteria in paragraph .25 or not.
114	D, J	Collateral provided that is both included in rows 18 or 21 and taken into account in C or NICA	The gross value of all collateral provided by the bank that is both included in row 18 or 21 and taken into account in the calculation of C or NICA in SA-CCR.
115	E, K	Potential future exposure of all margined netting sets w/o collateral	Potential future exposure of derivative transactions for margined netting sets, calculated by applying the SA-CCR with the modification that the PFE multiplier has to be set to one. Therefore, formulas for margined transactions may be applied to derivative transactions as appropriate.
116	E, K	Potential future exposure of all unmargined netting sets w/o collateral	Potential future exposure of derivative transactions for unmargined netting sets, calculated by applying the SA-CCI with the modification that the PFE multiplier has to be set to one.

Row	Column	Heading	Description
117	D, J	Potential future exposure:	Potential future exposure of derivative transactions calculated according to paragraphs 146 to 187 of the SA-CCR document.
117	E, K	Potential future exposure:	Non-entry cell. It is the sum of the potential future exposure of all margined and unmargined netting sets without collateral as requested in rows 115 and 116. Potential future exposure of derivative transactions calculated by applying the SA-CCR with the following modifications: • formulas for unmargined transactions must be applied to all derivative transactions, including those for which variation margin is exchanged; • the PFE multiplier has to be set to one.
118	E, K	Sum of trade-level gross add-ons	Potential future exposure of derivative transactions calculated: • by multiplying, for each individual derivative transaction, (i) the absolute value of the supervisory delta (paragraph 159 of the SA-CCR document); (ii) the adjusted notional amount (paragraphs 157 and158 of the SA-CCR document); (iii) the supervisory factor (paragraph 183 of the SA-CCR document); and (iv) the unmargined maturity factor (first bullet of paragraph 164 of the SA-CCR document); and then • by summing up the resulting trade-level add-ons.
119	D, E, J, K	Check row	This is a non-data entry row. It checks that the potential future exposure amounts with netting are not greater than the sum of trade-level gross add-ons.

5.9 Alternative currency criteria for eligible cash variation margin in derivative transactions (panel H)

Panel H requests for additional data on alternative interpretations of the currency criterion as set out in paragraph 25(iii) of the Basel III leverage ratio framework for eligible cash variation margin *received* or *posted* in relation to derivative transactions (see paragraphs 25 and 26 of the Basel III leverage ratio framework).

The data included in rows 13 and 22 of panel A (where applicable under the operative accounting framework) of the template should reflect the interpretation given in Q1 of the FAQs on the Basel III leverage ratio framework. According to the said FAQ Q1 the variation margin in every currency specified in the derivative contract, governing qualifying master netting agreement, or the credit support annex to the qualifying master netting agreement meets the currency criterion.

Panel H requests data on two alternative currency criteria:

- **Currency criterion 1**: for cash variation margin to be eligible, it must offset derivative market values whose associated cash flows will be settled in the same currency (currency of settlement). Hence, for each cash variation margin received (posted) in any currency X, banks have to determine whether the netting set contains a positive (negative) derivative market value in this given currency X. If it is the case, the cash variation margin meets the currency criterion.
- **Currency criterion 2**: for cash variation margin to be eligible, the cash flows of all derivative contracts in the netting set must be settled in the currency of the variation margin. This implies

that for netting sets involving derivatives to be settled in more than one currency, no variation margin is eligible.

Row	Column	Heading	Description
126	D, J	Eligible cash variation margin received and offset against market values of derivative transactions (criterion 1)	Same as row 13 but applying stricter currency criterion 1 specified above (as opposed to the interpretation specified in Q1 of the FAQs on the Basel III leverage ratio framework)
127	D, J	Eligible cash variation margin received and offset against market values of derivative transactions (criterion 2)	Same as row 13 but applying stricter currency criterion 2 specified above (as opposed to the interpretation specified in Q1 of the FAQs on the Basel III leverage ratio framework)
128	D, J	Check row	This is a non-data entry row. It checks that eligible cash variation margin received and offset against the market values of the derivative transactions under the eligibility criterion 1 cannot be greater than under the criterion specified in Q1 of the FAQs on the Basel III leverage ratio framework and as reported in row 13.
129	D, J	Check row	This is a non-data entry row. It checks that eligible cash variation margin received and offset against the market values of the derivative transactions under the eligibility criterion 2 cannot be greater than under the criterion specified in Q1 of the document FAQs on the Basel III leverage ratio framework and as reported in row 13.
130	D, J	Check row	This is a non-data entry row. It checks that eligible cash variation margin received and offset against the market values of the derivative transactions under the eligibility criterion 2 cannot be greater than under the criterion 1.
132	D, J	Receivables for eligible cash variation margin provided in derivatives transactions (criterion 1)	Same as row 22, where applicable under the operative accounting framework, but applying stricter currency criterion 1 specified above (as opposed to the interpretation specified in Q1 of the FAQs on the Basel III leverage ration framework).
133	D, J	Receivables for eligible cash variation margin provided in derivatives transactions (criterion 2)	Same as row 22, where applicable under the operative accounting framework, but applying stricter currency criterion 2 specified above (as opposed to the interpretation specified in Q1 of the FAQs on the Basel III leverage ration framework).
134	D, J	Check row	This is a non-data entry row. It checks that the amount of receivables for eligible cash variation margin provided for derivative transactions under criterion 1 cannot be greater than under the criterion specified in Q1 of the FAQs on the Basel III leverage ratio framework and as reported in row 22
135	D, J	Check row	This is a non-data entry row. It checks that the amount of receivables for eligible cash variation margin provided for derivative transactions under criterion 2 cannot be greater than under the criterion specified in Q1 of the FAQs on the Basel III leverage ratio framework and as reported in row 22
136	D, J	Check row	This is a non-data entry row. It checks that that the amount of receivables for eligible cash variation margin provided for derivative transactions under criterion 2 cannot be greater than under criterion 1.

5.10 Memo items related to initial margin for centrally cleared derivative transactions (panel I)

Panel I requests additional data *on initial margin* that a **clearing member** (CM) bank collects from its clients for cleared derivative transactions. **By construction**, a **bank that is not a CM or a bank that is a CM but does not engage in clearing of client derivative transactions should report zero amounts in all the cells of panel I.**

Row	Column	Heading	Description
142	D, J	Initial margin in the form of securities that a bank receives from clients for cleared derivative transactions	These cells include initial margin in the form of securities tha a bank receives from clients for cleared transactions.
143	D, J	a. Amount of the cash initial margin that the bank passes on to an account in the name of the CCP	Report the amount of the cash initial margin that the bank passes on to a QCCP (eg deposits into a QCCP account). Provide additional details of this amount in rows 144 and 145.
144	D, J	i. Amount of the cash initial margin that remains on the bank's balance sheet	Report the amount of the cash initial margin that the bank passes on to a QCCP, but which continues to remain on the bank's balance sheet.
145	D, J	ii. Amount of the cash initial margin that is off the bank's balance sheet but continues to create an off-balance sheet exposure of the bank	Report the amount of the cash initial margin that the bank passes on to a QCCP and which is off the bank's balance sheet, but continues to create an off-balance sheet exposure of the bank. For example, this may be the case if a bank provides a guarantee to the client with regard to the value of the initial margin that is passed on to the QCCP.
146	D, J	b. Amount of the cash initial margin that is segregated from the bank's other assets	Report the amount of the initial margin (that the bank does not pass on to a QCCP) that is segregated from the bank's other assets. That is, the contractual provisions relating to this segregated initial margin mean that it may not be used, pledged or re-hypothecated by the bank for its own busines purposes. However, such segregated margin may be used in accordance with the applicable customer protection rules, subject to the prior agreement with the clearing client.
147	D, J	c. Amount of the cash initial margin that is not segregated from the bank's other assets	Report the portion of the initial margin (that the bank does not pass on to a QCCP) that is not segregated from the bank's other assets, as described in row 146.
148	D, J	Initial margin in the form of cash that a bank receives from clients for centrally cleared derivative transactions	This is a non-data entry row. It calculates the full amount of initial margin in the form of cash that a bank receives from clients for cleared transactions, regardless whether such initial margin is included on the bank's balance sheet. The breakout of this amount is included in rows 143, 146 and 147.
149	D, J	a. Initial margin in the form of securities that a bank includes in its total Basel III leverage ratio exposure measure	Report the amount of initial margin in the form of securities that a bank includes in its total Basel III leverage ratio exposure measure.
150	D, J	b. Initial margin in the form of cash that a bank includes in its total Basel III leverage ratio exposure measure	Report the amount of initial margin in the form of cash that a bank includes in its total Basel III leverage ratio exposure measure.

Row	Column	Heading	Description
151	D, J	Initial margin that a bank includes in its total Basel III leverage ratio exposure measure	This is a non-data entry row. It calculates the amount of the initial margin received from the bank's clients for cleared transactions (in the form of cash or securities) that the bank includes in its total Basel III leverage ratio exposure measure. The breakout of this amount is included rows 149 and 150.

5.11 Business model categorisation (panel J)

Panel J provides additional data for the purposes of the categorisation of business models. The definitions for the line items correspond as far as possible with those provided in the Basel II framework (cross references as provided below).

Row	Column	Heading	Description
157	J	Total exposures; of which:	This is a non-data entry row. Rows 158, 161 and 187 provide a breakdown of total exposures.
158	J	Total trading book exposures; of which:	This is a non-data entry row. Items in rows 159 and 160 provide a breakdown of the Basel III leverage ratio exposure amount for exposures that meet the definition in paragraphs 685 to 689(iii) of the Basel II framework.
159	J	Derivatives, SFTs	Basel III leverage ratio exposure amount for derivatives and SFT that belong to the trading book according to paragraphs 685 to 689(iii) of the Basel II framework.
160	J	Other trading book exposures	Basel III leverage ratio exposure amount for instruments that belong to the trading book according to paragraphs 685 to 689(iii) of the Basel II framework other than derivatives and SFT.
161	J	Total banking book exposures; of which:	This is a non-data entry row. Items in rows 162 to 164 provide a breakdown of the Basel III leverage ratio exposure amount for all exposures that do not meet the definition in paragraphs 685 to 689(iii) of the Basel II framework.
162	J	Derivatives, SFTs	Basel III leverage ratio exposure amount for derivatives and SFT.
163	J	Investments in covered bonds	Basel III leverage ratio exposure amount for covered bonds.
164	J	Other banking book exposures; of which:	This is a non-data entry row. Items in rows 165, 172, 173, 178 and 184 provide a breakdown of the Basel III leverage exposure amount of banking book exposures other than derivatives, SFT and covered bonds.
165	J	Sovereigns; of which:	This is a non-data entry row. Basel III leverage ratio exposure amount for exposures which meet the definition in paragraph 229 of the Basel II framework, as well as Basel III leverage ratio exposures that meet the definition of claims on domestic PSEs and of exposures to MDBs in paragraph 230 of the Basel II framework. Items in rows 166, 170 and 171 provide a breakdown of the sovereign exposures.
166	J	Public sector entities (PSEs); of which:	Basel III leverage ratio exposure amount for exposures to PSEs referred to in paragraphs 229 and 230 of the Basel II framework.
167	J	PSE guaranteed by central government	Basel III leverage ratio exposure amount for PSE exposures guaranteed by central government (of which item, also to be included in row 166).

Row	Column	Heading	Description
168	J	PSEs not guaranteed by central government but treated as a sovereign under paragraph 229 of the Basel II framework	Basel III leverage ratio exposure amount for PSEs not guaranteed by central government but treated as a sovereign under paragraph 229 of the Basel II framework (o which item, also to be included in row 166).
169	J	Check row	This is a non-data entry row. It checks that the sum of the exposure amounts in rows 167 and 168 is smaller than or equal the amount of total PSE exposures.
170	J	MDBs	Basel III leverage ratio exposure amount for exposures to MDBs referred to in paragraphs 229 and 230 of the Basel II framework.
171	J	Other sovereign exposures	Basel III leverage ratio exposure amount for sovereigns exposures, excluding exposures to PSEs and MDBs.
172	J	Banks	Basel III leverage ratio exposure amount for exposures whice meet the definition in paragraph 230 of the Basel II framework, excluding exposures to PSEs and MDBs.
173	J	Retail exposures; of which:	This is a non-data entry row. Items in rows 174 to 177 provide a breakdown of Basel III leverage ratio exposure amount for exposures which meet the definition in paragraphs 231 to 234 of the Basel II framework.
174	J	Residential real estate exposures	Basel III leverage ratio exposure amount for exposures whice meet the definition in the second bullet of paragraph 231 of the Basel II framework.
175	J	SME exposures	Basel III leverage ratio exposure amount for exposures whice meet the definition in the third bullet of paragraph 231 and in paragraph 232 of the Basel II framework.
176	J	Qualifying revolving retail exposures	Basel III leverage ratio exposure amount for exposures which meet the definition in paragraph 234 of the Basel II framework.
177	J	Other retail exposures	Basel III leverage ratio exposure amount for retail exposures other than residential real estate, SME and qualifying revolving retail exposures.
178	J	Corporate ; of which:	This is a non-data entry row. Items in rows 179 and 180 provide a breakdown of Basel III leverage ratio exposure amount for exposures which meet the definition in paragraphs 218 to 228 of the Basel II framework.
179	J	Financial	Basel III leverage ratio exposure amount for corporate exposures which meet the definition in paragraph 102 of th Basel III framework, excluding exposures to banks.
180	J	Non-financial; of which:	This is a non-data entry row. Items in rows 181 to 183 provide a breakdown of non-financial exposures.
181	J	SME exposures	Basel III leverage ratio exposure amount for exposures which meet the definition in paragraph 273 of the Basel II framework excluding exposures that meet the definition in paragraphs 231, third bullet, and 232.
182	J	Commercial real estate	Basel III leverage ratio exposure amount for commercial rea estate exposures which meet the definition in paragraphs 219 to 228 of the Basel II framework.
183	J	Other corporate non- financial	Basel III leverage ratio exposure amount for non-financial corporate exposures which meet the definition in paragraph 219 to 228 of the Basel II framework, other than SME and commercial real estate exposures.

Row	Column	Heading	Description
184	J	Other exposures (eg equity and other non-credit obligation assets); of which:	Basel III leverage ratio exposure amount for banking book exposures other than sovereigns, banks, retail and corporate exposures.
185	J	Securitisation exposures	Basel III leverage ratio exposure amount for securitisation exposures (of which item, also to be included in row 184).
186	J	Check row	This is a non-data entry row. It checks that the exposure amount for securitisation exposures is smaller than or equal the amount of total other exposures.
187	J	Exposure amounts resulting from the additional treatment for credit derivatives	Basel III leverage ratio exposure amount for capped notional amounts for credit derivatives (panel E).
188	J	Check row	This is a non-data entry row. It checks that total in row 157 equals total exposures in panels A, B and E.
190	J	Memo item: Trade finance exposures	Basel III leverage ratio exposure amount for issued and confirmed import and export letters of credit which are short-term and self-liquidating, and similar transactions. Trade finance exposures should also be included in one of the rows 158 to 185.

Banks should report all exposure values consistent with the calculations for the purposes of the Basel III leverage ratio in the rest of this worksheet. As a result, row 157 should equal total exposures in panels A, B and E. Unless mentioned otherwise, the input rows in this panel are mutually exclusive. Rows 157, 158, 161, 164, 165, 173, 178 and 180 are non-data entry rows, and rows 169, 186 and 188 include checks.

5.12 EU-specific (panel K)

This panel should only be completed by banks in the European Union.

The data item described below refers to the same derivative exposures reported in the main Basel III leverage ratio reporting template of the Basel III monitoring exercise. However, instead of applying the Current Exposure Method (CEM) of the Basel II framework, institutions shall apply the *Original Exposure Method* (OEM) as set forth in Article 275 of the *Capital Requirements Regulation* (CRR), as published in the *Official Journal of the European Union* in June 2013, to determine the values reported in cell J196. Institutions that do not use the OEM shall leave cell J196 blank.

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description
196	J	Exposure value when applying the Original Exposure Method	This cell provides the leverage ratio exposure value of derivatives calculated according to the Original Exposure Method set forth in Article 275 of the CRR as published in the Official Journal of the European Union in June 2013.
197	J	Check row	This is a non-data entry row. It checks that J196 does not exceed L39.

6. Liquidity

This chapter of the Instructions regards the LCR and NSFR. The data collection is predominantly aimed at monitoring the LCR as specified in *Basel III: The Liquidity Coverage Ratio and liquidity risk monitoring*

tools, published by the Committee in January 2013 and the NSFR as specified in *Basel III: The Net Stable Funding Ratio*, published by the Committee in October 2014. These documents are referred to in the remainder of this chapter as the "Basel III LCR standards" and "Basel III NSFR standards", respectively.

Purpose of this exercise is to collect information that enables the Committee to monitor banks migration towards compliance with the LCR and NSFR as specified in the Basel III LCR standards and Basel III NSFR standards, respectively.

The liquidity data are collected in two templates: one for the LCR and one for the NSFR. The template for the LCR is built up the same way as the LCR section in the Basel III LCR standards.

All specifications and criteria specified in the Basel III LCR standards and the Basel III NSFR standards apply. The instructions indicate which paragraph of these documents the data requested refer to. If the instruction contradicts these documents, the standards overrule the instructions. Where the instructions provide further specification on the requested data beyond the standards, however, these instructions should be followed.

The template should be filled in on a consolidated basis following the existing scope of application set out in Part I (Scope of Application) of the Basel II framework (Basel III LCR standards paragraph 164). Consistent with all other worksheets, data for the "LCR" and "NSFR" worksheets should be reported in the most convenient currency. The currency which has been used should be recorded in the "General Info" worksheet (see Section 2.2).

6.1 Liquidity coverage ratio (LCR)

The LCR has two components:

- (a) The value of the stock of high-quality liquid assets (HQLA) in stressed conditions (see subsection 6.1.1 below); and
- (b) Total net cash outflows, calculated according to the scenario parameters set by the supervisors. The term "total net cash outflows" is defined as "total expected cash outflows" (see sub-section 6.1.2 below) minus "total expected cash inflows" (see sub-section 6.1.3 below) in the specified stress scenario for the subsequent 30 calendar days (the stressed period).

6.1.1 Liquid assets (panel A)

Operational requirements (paragraphs 28 to 40 in the Basel III LCR standards): All assets in the stock are subject to the following operational requirements. These operational requirements are designed to ensure that the stock of HQLA is managed in such a way that the bank can, and is able to demonstrate that it can, immediately use the stock of assets as a source of contingent funds that is available for the bank to convert into cash through outright sale or repo, to fill funding gaps between cash inflows and outflows at any time during the 30 day stress period, with no restriction on the use of the liquidity generated.

All assets in the stock should be unencumbered, per the definition below. Banks should exclude from the stock those assets that, although meeting the definition of "unencumbered" specified below, the bank would not have the operational capability to monetise to meet outflows during the stressed period. Operational capability to monetise assets requires having procedures and appropriate systems in place, including providing the function noted below with access to all necessary information to execute monetisation of any asset at any time. Monetisation of the asset must be executable, from an operational perspective, in the standard settlement period for the asset class in the relevant jurisdiction.

All assets accounted for in this section should be under the control of the function charged with managing the liquidity of the bank (eg the treasurer), meaning the function has the continuous authority, and legal and operational capability, to monetise any asset in the stock. Control must be evidenced either by maintaining assets in a separate pool managed by the function with the sole intent

for use as a source of contingent funds, or by demonstrating that the function can monetise the asset at any point in the 30 day stress period and that the proceeds of doing so are available to the function throughout the 30 day stress period without directly conflicting with a stated business or risk management strategy. For example, an asset should not be included in the stock if the sale of that asset, without replacement throughout the 30 day period, would remove a hedge that would create an open risk position in excess of internal limits.

A bank is permitted to hedge the market risk associated with ownership of the stock of liquid assets and still include the assets in the stock. If it chooses to hedge the market risk, the bank should take into account (in the market value applied to each asset) the cash outflow that would arise if the hedge were to be closed out early (in the event of the asset being sold).

In accordance with Principle 9 of the *Sound Principles* a bank "should monitor the legal entity and physical location where collateral is held and how it may be mobilised in a timely manner". Specifically it should have a policy in place that identifies legal entities, geographical locations, currencies and specific custodial or bank accounts where HQLA are held. In addition the bank should determine whether any such assets should be excluded for operational reasons and, therefore, have the ability to determine the composition of its stock on a daily basis.

Qualifying HQLA that are held to meet statutory liquidity requirements at the legal entity or sub-consolidated level (where applicable) may only be included in the stock at the consolidated level to the extent that the related risks (as measured by the legal entity's or sub-consolidated group's net cash outflows in the LCR) are also reflected in the consolidated LCR. Any surplus of HQLA held at the legal entity can only be included in the consolidated stock if those assets would also be freely available to the consolidated (parent) entity in times of stress.

In assessing whether assets are freely transferable for regulatory purposes, banks should be aware that assets may not be freely available to the consolidated entity due to regulatory, legal, tax, accounting or other impediments. Assets held in legal entities without market access should only be included to the extent that they can be freely transferred to other entities that could monetise the assets.

In certain jurisdictions, large, deep and active repo markets do not exist for eligible asset classes, and therefore such assets are likely to be monetised through outright sale. In these circumstances, a bank should exclude from the stock of HQLA those assets where there are impediments to sale, such as large fire-sale discounts which would cause it to breach minimum solvency requirements, or requirements to hold such assets, including, but not limited to, statutory minimum inventory requirements for market-making.

Banks should not include in the stock of HQLA any assets, or liquidity generated from assets, they have received under right of rehypothecation, if the beneficial owner has the contractual right to withdraw those assets during the 30 day stress period.

Assets received as collateral for derivatives transactions that are not segregated and legally able to be rehypothecated may be included in the stock of HQLA provided that the bank records an appropriate outflow for the associated risks as set out in the Basel III LCR standards paragraph 116.

As part of the stock, the liquid assets cannot be counted as cash inflows even if they mature within 30 days (ie no double-counting is allowed).

Definition of unencumbered: free of legal, regulatory, contractual or other restrictions on the ability of the bank to liquidate, sell, transfer, or assign the asset. An asset in the stock should not be pledged by the bank (either explicitly or implicitly) to secure, collateralise or credit-enhance any transaction, nor be designated to cover operational costs (such as rents and salaries). However, assets that the bank received as collateral in reverse repo and securities financing transactions can be considered as part of the stock if they are held at the bank, have not been rehypothecated, and are legally and contractually available for the bank's use. In addition, assets which qualify for the stock of HQLA that have been prepositioned or deposited with, or pledged to, the central bank or a public sector

entity (PSE) but have not been used to generate liquidity may be included in the stock. If a bank has deposited, pre-positioned or pledged Level 1, Level 2 and other assets in a collateral pool and no specific securities are assigned as collateral for any transactions, it may assume that assets are encumbered in order of increasing liquidity value in the LCR, ie assets ineligible for the LCR are assigned first, followed by Level 2B, then other Level 2 and finally Level 1. This determination must be made in compliance with any requirements, such as concentration or diversification, of the central bank or PSE.

Criteria of liquid assets: To qualify as a "high quality liquid asset", assets should be liquid in markets during a time of stress and, with the exception of Level 2B assets, ideally be central bank eligible. Such assets should generally possess the fundamental and market-related characteristics specified in paragraphs 24(i) and 24(ii) of the Basel III LCR standards. Securities that can be included in the stock of HQLA should meet the following common criteria (note that additional security-specific criteria are included in the individual line item descriptions):

- they should neither be issued by, nor be an obligation of, a financial institution ⁴² or any of its affiliated entities (except in the case of covered bonds and RMBS which should not be issued by the bank itself or any of its affiliated entities);
- they should be traded in large, deep and active repo or cash markets characterised by a low level of concentration;
- they should have a proven record as a reliable source of liquidity in the markets (repo or sale) even during stressed market conditions; and
- with the exception of Level 2B assets, they should ideally be central bank eligible.

Row	Heading	Description	Basel III LCR standards reference
A)a)	Level 1 assets		
6	Coins and banknotes	Coins and banknotes currently held by the bank that are immediately available to meet obligations. Deposits placed at, or receivables from, other institutions should be reported in the inflows section.	50(a)
7	Total central bank reserves; of which:	Total amount held in central bank reserves (including required reserves) including banks' overnight deposits with the central bank, and term deposits with the central bank that: (i) are explicitly and contractually repayable on notice from the depositing bank; or (ii) that constitute a loan against which the bank can borrow on a term basis or on an overnight but automatically renewable basis (only where the bank has an existing deposit with the relevant central bank). Other term deposits with central banks are not eligible for the stock of HQLA; however, if the term expires within 30 days, the term deposit could be considered as an inflow (reported in line 305).	50(b), footnote 12
8	part of central bank reserves that can be drawn in times of stress	Total amount held in central bank reserves and overnight and term deposits at the same central bank (as reported in line 7) which can be drawn down in times of stress. Amounts required to be installed in the central bank reserves within 30 days should be reported in line 166 of the outflows section. Please refer to the instructions from your supervisor for the specification of this item.	50(b), footnote 13

Financial institutions, in this context, include banks, securities firms and insurance companies.

⁴³ Central bank eligibility alone is not a sufficient basis for determining which assets qualify as HQLA.

Row	Heading	Description	Basel III LCR standards reference
Securiti	es with a 0% risk weight:		
11	issued by sovereigns	Marketable debt securities issued by sovereigns, receiving a 0% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 53).	50(c)
12	guaranteed by sovereigns	Marketable debt securities guaranteed by sovereigns, receiving a 0% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 53).	50(c)
13	issued or guaranteed by central banks	Marketable debt securities issued or guaranteed by central banks, receiving a 0% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 53).	50(c)
14	issued or guaranteed by PSEs	Marketable debt securities issued or guaranteed by public sector entities, receiving a 0% risk weight under the standardised approach to credit risk of the Basel II framework (paragraphs 57 and 58).	50(c)
15	issued or guaranteed by BIS, IMF, ECB and European Community or MDBs	Marketable debt securities issued or guaranteed by the Bank for International Settlements, the International Monetary Fund, the European Central Bank (ECB) and European Community. or multilateral development banks (MDBs), receiving a 0% risk weight under the standardised approach to credit risk of the Basel II framework (paragraphs 56 and 59).	50(c)
For nor	-0% risk-weighted sovereigns:		
17	sovereign or central bank debt securities issued in domestic currency by the sovereign or central bank in the country in which the liquidity risk is taken or in the bank's home country	Debt securities issued by the sovereign or central bank in the domestic currency of that country, that are not eligible for inclusion in line items 11 or 13 because of the non-0% risk weight of that country.	50(d)
		Banks are only permitted to include debt issued by sovereigns or central banks of their home jurisdictions or, to the extent of the liquidity risk taken in other jurisdictions, of those jurisdictions.	
18	domestic sovereign or central bank debt securities issued in foreign currencies, up to the amount of the bank's stressed net cash outflows in that specific foreign currency stemming from the bank's operations in the jurisdiction where the bank's liquidity risk is being taken	Debt securities issued by the domestic sovereign or central bank in foreign currencies (that are not eligible for inclusion in line items 11 or 13 because of the non-0% risk weight), up to the amount of the bank's stressed net cash outflows in that specific foreign currency stemming from the bank's operations in the jurisdiction where the bank's liquidity risk is being taken.	50(e)
Total Le	evel 1 assets:		
19	Total stock of Level 1 assets	Total outright holdings of Level 1 assets plus all borrowed securities of Level 1 assets	49
20	Adjustment to stock of Level 1 assets	Adjustment to the stock of Level 1 assets for purpose of calculating the caps on Level 2 and Level 2B assets.	Annex 1
21	Adjusted amount of Level 1 assets	Adjusted amount of Level 1 assets used for the purpose of calculating the adjustment to the stock of HQLA due to the cap on Level 2 assets in line item 50 and the cap on Level 2B assets in line item 49.	Annex 1

Row	Heading	Description	Basel III LCR standards reference
A)b)	Level 2A assets		'
Securiti	es with a 20% risk weight:		
25	issued by sovereigns	Marketable debt securities issued by sovereigns, receiving a 20% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 53), satisfying all the conditions listed in paragraph 52(a) of the Basel III LCR standards, and not included in lines 17 or 18.	52(a)
26	guaranteed by sovereigns	Marketable debt securities guaranteed by sovereigns, receiving a 20% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 53), satisfying all the conditions listed in paragraph 52(a) of the Basel III LCR standards.	52(a)
27	issued or guaranteed by central banks	Marketable debt securities issued or guaranteed by central banks, receiving a 20% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 53), satisfying all the conditions listed in paragraph 52(a) of the Basel III LCR standards, and not included in lines 17 or 18.	52(a)
28	issued or guaranteed by PSEs	Marketable debt securities issued or guaranteed by PSEs, receiving a 20% risk weight under the standardised approach to credit risk of the Basel II framework (paragraphs 57 and 58), satisfying all the conditions listed in paragraph 52(a) of the Basel III LCR standards.	52(a)
29	issued or guaranteed by MDBs	Marketable debt securities issued or guaranteed by multilateral development banks, receiving a 20% risk weight under the standardised approach to credit risk of the Basel II framework (paragraph 59), satisfying all the conditions listed in paragraph 52(a) of the Basel III LCR standards.	52(a)
Non-fin	ancial corporate bonds:		
30	rated AA- or better	Non-financial corporate bonds (including commercial paper) (i) having a long-term credit assessment by a recognised ECAI of at least AA- or in the absence of a long term rating, a short-term rating equivalent in quality to the long-term rating or (ii) not having a credit assessment by a recognised ECAI but are internally rated as having a probability of default (PD) corresponding to a credit rating of at least AA-, satisfying the conditions listed in paragraph 52(b) of the Basel III LCR standards.	52(b)
Covered	d bonds (not self-issued):		
31	rated AA- or better	Covered bonds, not self-issued, (i) having a long-term credit assessment by a recognised ECAI of at least AA- or in the absence of a long term rating, a short-term rating equivalent in quality to the long-term rating or (ii) not having a credit assessment by a recognised ECAI but are internally rated as having a probability of default (PD) corresponding to a credit rating of at least AA-, satisfying the conditions listed in paragraph 52(b) of the Basel III LCR standards.	52(b)
Total Le	evel 2A assets:		
32	Total stock of Level 2A assets	Total outright holdings of Level 2A assets plus all borrowed securities of Level 2A assets, after applying haircuts	52(a),(b)
33	Adjustment to stock of Level 2A assets	Adjustment to the stock of Level 2A assets for purpose of calculating the caps on Level 2 and Level 2B assets.	Annex 1

Row	Heading	Description	Basel III LCR standards reference
34	Adjusted amount of Level 2A assets	Adjusted amount of Level 2A assets used for the purpose of calculating the adjustment to the stock of HQLA due to the cap on Level 2 assets in line item 50 and the cap on Level 2B assets in line item 49.	Annex 1

A)c) Level 2B assets

Please refer to the instructions from your supervisor for the specification of items in the Level 2B assets subsection

In choosing to include any Level 2B assets in Level 2, national supervisors are expected to ensure that (i) such assets fully comply with the qualifying criteria set out Basel III LCR standards, paragraph 54; and (ii) banks have appropriate systems and measures to monitor and control the potential risks (eg credit and market risks) that banks could be exposed to in holding these assets.

Holuling	tnese assets.		
37	Residential mortgage backed securities (RMBS), rated AA or better	RMBS that satisfy all of the conditions listed in paragraph 54(a) of the Basel III LCR standards.	54(a)
38	Non-financial corporate bonds, rated BBB- to A+	Non-financial corporate debt securities (including commercial paper) rated BBB- to A+ that satisfy all of the conditions listed in paragraph 54(b) of the Basel III LCR standards.	54(b)
39	Non-financial common equity shares	Non-financial common equity shares that satisfy all of the conditions listed in paragraph 54(c) of the Basel III LCR standards.	54(c)
40	Sovereign or central bank debt securities, rated BBB- to BBB+	Sovereign or central bank debt securities, rated BBB- to BBB+, that are not already included in lines 17 or 18, per FAQ 3(a) in Basel Committee on Banking Supervision, Frequently Asked Questions on Basel III's January 2013 Liquidity Coverage Ratio, April 2014, www.bis.org/publ/bcbs284.htm.	BCBS FAQ 3(a)
Total Le	evel 2B assets:		
41	Total stock of Level 2B RMBS assets	Total outright holdings of Level 2B RMBS assets plus all borrowed securities of Level 2B RMBS assets, after applying haircuts.	54(a)
42	Adjustment to stock of Level 2B RMBS assets	Adjustment to the stock of Level 2B RMBS assets for purpose of calculating the caps on Level 2 and Level 2B assets.	Annex 1
43	Adjusted amount of Level 2B RMBS assets	Adjusted amount of Level 2B RMBS assets used for the purpose of calculating the adjustment to the stock of HQLA due to the cap on Level 2 assets in line item 50 and the cap on Level 2B assets in line item 49.	Annex 1
44	Total stock of Level 2B non- RMBS assets	Total outright holdings of Level 2B non-RMBS assets plus all borrowed securities of Level 2B non-RMBS assets, after applying haircuts.	54(b),(c)
45	Adjustment to stock of Level 2B non-RMBS assets	Adjustment to the stock of Level 2B non-RMBS assets for purpose of calculating the caps on Level 2 and Level 2B assets.	Annex 1
46	Adjusted amount of Level 2B non-RMBS assets	Adjusted amount of Level 2B non-RMBS assets used for the purpose of calculating the adjustment to the stock of HQLA due to the cap on Level 2 assets in line item 50 and the cap on Level 2B assets in line item 49.	Annex 1
47	Adjusted amount of Level 2B (RMBS and non-RMBS) assets	Sum of adjusted amount of Level 2B RMBS assets and adjusted amount of Level 2B non-RMBS assets	Annex 1
49	Adjustment to stock of HQLA due to cap on Level 2B assets	Adjustment to stock of HQLA due to 15% cap on Level 2B assets.	47, Annex 1

Row	Heading	Description	Basel III LCR standards reference
50	Adjustment to stock of HQLA due to cap on Level 2 assets	Adjustment to stock of HQLA due to 40% cap on Level 2 assets.	51, Annex 1
A)d)	Total stock of HQLA		
53	Total stock of HQLA	Total stock of HQLA after taking haircuts and the adjustment for the caps on Level 2 and Level 2B assets into account.	
57	Assets held at the entity level, but excluded from the consolidated stock of HQLA	Any surplus of liquid assets held at the legal entity that is excluded (ie not reported in lines above) from the consolidated stock because of reasonable doubts that they would be freely available to the consolidated (parent) entity in times of stress. Eligible liquid assets that are held by a legal entity being consolidated to meet its local LCR requirements (where applicable) can be included in the consolidated LCR to the extent that such liquid assets are used to cover the total net cash outflows of that entity, notwithstanding that the assets are subject to liquidity transfer restrictions. If the liquid assets held in excess of the total net cash outflows of the legal entity are not transferable, such surplus liquidity should be excluded from the standard and reported in this line. For practical reasons, the liquidity transfer restrictions to be accounted for in the consolidated ratio are confined to existing restrictions imposed under applicable laws, regulations and supervisory requirements. Banks should report the market value of Level 1 assets excluded in column D, the market value of Level 2A assets excluded in column E, the market value of Level 2B RMBS assets excluded in column F and the market value of Level 2B non-RMBS assets excluded in column G.	36–37, 171– 172
58	of which, can be included in the consolidated stock by the time the standard is implemented	Any assets reported in row 57 but which the bank believes will, through management actions executed prior to the implementation date of the standard, meet the eligibility requirements for the stock of liquid assets.	
60	Assets excluded from the stock of HQLA due to operational restrictions	Level 1 and Level 2 assets held by the bank that are not included in the stock of HQLA (ie not reported in lines above), because of the operational restrictions noted in paragraphs 31-34 and 38-40 of the Basel III LCR standards. Banks should report the market value of Level 1 assets excluded in column D, the market value of Level 2A assets excluded in column E, the market value of Level 2B RMBS assets excluded in column F and the market value of Level 2B non-RMBS assets excluded in column G.	31–34, 38–40
61	of which, can be bought back into the qualifying stock by the time the standard is implemented	Any assets reported in row 60 but which the bank believes will, through management actions executed prior to the implementation date of the standard, meet the eligibility requirements for the stock of liquid assets.	

Row	Heading	Description	Basel III LCR
			standards
			reference

A)e) Treatment for jurisdictions with insufficient HQLA

Please refer to the instructions from your supervisor for the specification of this subsection.

Some jurisdictions may not have sufficient supply of Level 1 assets (or both Level 1 and Level 2 assets) in their domestic currency to meet the aggregate demand of banks with significant exposures in this currency (note that an insufficiency in Level 2 assets alone does not qualify for the alternative treatment). To address this situation, the Committee has developed alternative treatments for the holdings in the stock of HQLA, which are expected to apply to a limited number of currencies and jurisdictions.

Eligibility for such alternative treatment will be judged on the basis of qualifying criteria set out in Annex 2 of the Basel III LCR standards and will be determined through an independent peer review process overseen by the Committee. The purpose of this process is to ensure that the alternative treatments are only used when there is a true shortfall in HQLA in the domestic currency relative to the needs in that currency.

There are three potential options for this treatment (line items 68 to 72). **If your supervisor intends to adopt this treatment, it is expected that they provide specific instructions to the banks under its supervision for reporting the relevant information under the option it intends to use.** To avoid double-counting, if an asset has already been included in the eligible stock of HQLA, it should not be reported under these options.

Option 1 - Contractual committed liquidity facilities from the relevant central bank, with a fee

These facilities should not be confused with regular central bank standing arrangements. In particular, these facilities are contractual arrangements between the central bank and the commercial bank with a maturity date which, at a minimum, falls outside the 30-day LCR window. Further, the contract must be irrevocable prior to maturity and involve no ex-post credit decision by the central bank.

Such facilities are only permissible if there is also a fee for the facility which is charged regardless of the amount, if any, drawn down against that facility and the fee is set so that banks which claim the facility line to meet the LCR, and banks which do not, have similar financial incentives to reduce their exposure to liquidity risk. That is, the fee should be set so that the net yield on the assets used to secure the facility should not be higher than the net yield on a representative portfolio of Level 1 and Level 2 assets, after adjusting for any material differences in credit risk.

68	Option 1 – Contractual	Only include the portion of facility that is secured by available	58
	committed liquidity facilities	collateral accepted by the central bank, after haircut specified	
	from the relevant central	by the central bank. Please refer to the instructions from	
	bank	your supervisor for the specification of this item.	

Option 2 - Foreign currency HQLA to cover domestic currency liquidity needs

For currencies that do not have sufficient HQLA, supervisors may permit banks that evidence a shortfall of HQLA in the domestic currency (which would match the currency of the underlying risks) to hold HQLA in a currency that does not match the currency of the associated liquidity risk, provided that the resulting currency mismatch positions are justifiable and controlled within limits agreed by their supervisors.

To account for foreign exchange risk associated with foreign currency HQLA used to cover liquidity needs in the domestic currency, such liquid assets should be subject to a minimum haircut of 8% for major currencies that are active in global foreign exchange markets. For other currencies, supervisors should increase the haircut to an appropriate level on the basis of historical (monthly) exchange rate volatilities between the currency pair over an extended period of time. If the domestic currency is formally pegged to another currency under an effective mechanism, the haircut for the pegged currency can be lowered to a level that reflects the limited exchange rate risk under the peg arrangement. Haircuts for foreign currency HQLA used under Option 2 would apply only to HQLA in excess of a threshold specified by supervisors which is not greater than 25% that are used to cover liquidity needs in the domestic currency.

70	Level 1 assets	Subject to the limit mentioned above, the aggregate amount of the excess of Level 1 assets in a given foreign currency or currencies that can be used to cover the associated liquidity need of the domestic currency. Please refer to the instructions from your supervisor for the specification of this item.	59
71	Level 2 assets	Subject to the limit mentioned above, the aggregate amount of the excess of Level 2 assets in a given foreign currency or currencies that can be used to cover the associated liquidity need of the domestic currency. Please refer to the instructions from your supervisor for the specification of this item.	59

			1
Row	Heading	Description	Basel III LCR standards reference
			reference
Option	3 – Additional use of Level 2 as	ssets with a higher haircut	
principl that evi- hold ad 5% high	es and criteria, but where there a dence a shortfall of liquid assets ditional Level 2A assets in the sto ner than the 15% haircut applicat	ch there are insufficient Level 1 assets, as determined by the qualifier sufficient Level 2A assets. In this case, supervisors may choose to in the domestic currency (to match the currency of the liquidity rislock. These additional Level 2A assets should be subject to a minimulate to Level 2A assets that are included in the 40% cap. Any Level 21 cap of 15%, regardless of the amount of other Level 2 assets held.	o allow banks k incurred) to um 20% – ie B assets held
72	Option 3 – Additional use of Level 2 assets with a higher haircut	Assets reported in lines 25 to 31 that are not counted towards the regular stock of HQLA because of the cap on Level 2 assets. Please refer to the instructions from your supervisor for the specification of this item.	62
Total u	sage of alternative treatment		
73	Total usage of alternative treatment (post-haircut) before applying the cap	Sum of the usage of alternative treatment should be equal to total outright holdings and all borrowed securities under different options. Please refer to the instructions from your supervisor for the specification of this item.	
74	Cap on usage of alternative treatment	Please refer to the instructions from your supervisor for the specification of this item.	
75	Total usage of alternative treatment (post-haircut) after applying the cap	The lower of the cap and eligible alternative treatment (post-haircut) before applying the cap. Please refer to the instructions from your supervisor for the specification of this item.	
A)f)	Total stock of HQLA plus us	age of alternative treatment	
78	Total stock of HQLA plus	Sum of stock of HQLA and usage of alternative treatment after	

6.1.2 Outflows, Liquidity Coverage Ratio (LCR) (panel B1)

cap.

usage of alternative

treatment

This section calculates the total expected cash outflows in the LCR stress scenario for the subsequent 30 calendar days. They are calculated by multiplying the outstanding balances of various categories or types of liabilities and off-balance sheet commitments by the rates at which they are expected to run off or to be drawn down (Basel III LCR standards paragraph 69).

Where there is potential that an item could be reported in multiple outflow categories, (eg committed liquidity facilities granted to cover debt maturing within the 30 calendar day period), a bank only has to assume up to the maximum contractual outflow for that product (Basel III LCR standards paragraph 72).

Row	Heading	Description	Basel III LCR
			standards
			reference

a) Retail deposit run-off

Retail deposits are defined as deposits placed with a bank by a natural person. Deposits from legal entities, sole proprietorships and partnerships are captured in wholesale deposit categories. Retail deposits reported in lines 88 to 108 include demand deposits and term deposits maturing in or with a notice period up to 30 days.

Term deposits with a residual contractual maturity greater than 30 days which may be withdrawn within 30 days without entailing a significant withdrawal penalty materially greater than the loss of interest, should be considered to mature within the 30-day horizon and should also be included in lines 88 to 108 as appropriate. If a portion of the term deposit can be withdrawn without incurring such a penalty, only that portion should be treated as a demand deposit. The remaining balance of the deposit should be treated as a term deposit.

Notes, bonds and other debt securities sold exclusively to the retail market and held in retail accounts can be reported in the appropriate retail deposit category (Basel III LCR standards paragraph 110). To be treated in this manner, it is not sufficient that the debt instruments are specifically designed and marketed to retail customers. Rather there should be limitations placed such that those instruments cannot be bought and held by parties other than retail customers.

Per paragraph 76 of the Basel III LCR standards, an "effective deposit insurance scheme" refers to a scheme (i) that guarantees that it has the ability to make prompt payouts, (ii) for which the coverage is clearly defined and (iii) of which public awareness is high. The deposit insurer in an effective deposit insurance scheme has formal legal powers to fulfil its mandate and is operationally independent, transparent and accountable. A jurisdiction with an explicit and legally binding sovereign deposit guarantee that effectively functions as deposit insurance can be regarded as having an effective deposit insurance scheme.

84	Total retail deposits; of which	Total retail deposits as defined above.	73–84
85	Insured deposits; of which:	The portion of retail deposits that are fully insured by an effective deposit insurance scheme.	75–78
86	in transactional accounts; of which:	Total insured retail deposits in transactional accounts (eg accounts where salaries are automatically credited).	75, 78
87	eligible for a 3% run-off rate; of which:	The amount of insured transactional retail deposits that are in jurisdictions where the supervisor chooses to apply a 3% run-off rate given the deposits are fully insured by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards. Please refer to the instructions from your supervisor for the specification of these items.	78
88	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 87, the amount that are in the reporting bank's home jurisdiction.	78
89	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 87, the amount that are not in the reporting bank's home jurisdiction.	78
90	eligible for a 5% run-off rate; of which:	The amount of insured transactional retail deposits that are in jurisdictions where the supervisor does not choose to apply a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	75
91	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 90, the amount that are in the reporting bank's home jurisdiction.	75
92	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 90, the amount that are not in the reporting bank's home jurisdiction.	75
93	in non-transactional accounts with established relationships that make deposit withdrawal highly unlikely; of which:	Total insured retail deposits in non-transactional accounts where the customer has another relationship with the bank that would make deposit withdrawal highly unlikely.	75, 78

Row	Heading	Description	Basel III LCR standards reference
94	eligible for a 3% run-off rate; of which:	The amount of insured non-transactional established relationship retail deposits that are in jurisdictions where the supervisor chooses to apply a 3% run-off rate given the deposits are fully insured by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards. Please refer to the instructions from your supervisor for the specification of these items.	78
95	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 94, the amount that are in the reporting bank's home jurisdiction.	78
96	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 94, the amount that are not in the reporting bank's home jurisdiction.	78
97	eligible for a 5% run-off rate; of which:	The amount of insured non-transactional established relationship retail deposits that are in jurisdictions where the supervisor does not choose to apply a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	75
98	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 97, the amount that are in the reporting bank's home jurisdiction.	75
99	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 97, the amount that are not in the reporting bank's home jurisdiction.	75
100	in non-transactional and non-relationship accounts	Insured retail deposits in non-transactional accounts where the customer does not have another relationship with the bank that would make deposit withdrawal highly unlikely.	79
101	Uninsured deposits	The portion of retail deposits that are non-maturing or mature within 30 days that are not fully insured by an effective deposit insurance scheme (ie all retail deposits not reported in lines 88 to 100, excluding any deposits included in lines 103 to 105).	79
102	Additional deposit categories with higher run-off rates as specified by supervisor	Other retail deposit categories, as defined by the supervisor. These amounts should not be included in the lines above.	79
103	Category 1	As defined by supervisor	79
104	Category 2	As defined by supervisor	79
105	Category 3	As defined by supervisor	79
106	Term deposits (treated as having >30 day remaining maturity); of which	Retail deposits with a residual maturity or withdrawal notice period greater than 30 days where the depositor has no legal right to withdraw deposits within 30 days, or where early withdrawal results in a significant penalty that is materially greater than the loss of interest.	82–84
107	With a supervisory run-off rate	As defined by supervisor.	84
108	Without supervisory run-off rate	All other term retail deposits treated as having > 30 day remaining maturity as defined in line 106.	82

b) Unsecured wholesale funding run-off

Unsecured wholesale funding is defined as liabilities and general obligations that are raised from non-natural persons (ie legal entities, including sole proprietorships and partnerships) and are **not** collateralised by legal rights to specifically designated assets owned by the borrowing institution in the case of bankruptcy, insolvency, liquidation or resolution, excluding derivatives.

Wholesale funding included in the LCR is defined as all funding that is callable within the LCR's 30-day horizon or that has its earliest possible contractual maturity date within this horizon (such as maturing term deposits and unsecured debt securities) as well as funding with an undetermined maturity. This includes all funding with options that are exercisable at the investor's discretion within the 30-day horizon. It also includes funding with options exercisable at the

Row	Heading	Description	Basel III LCR
			standards
			reference

bank's discretion where the bank's ability not to exercise the option is limited for reputational reasons. In particular, where the market expects certain liabilities to be redeemed before their legal final maturity date and within the 30-day horizon, such liabilities should be included in the appropriate outflows category.

Small business customers

Unsecured wholesale funding provided by small business customers consists of deposits and other extensions of funds made by non-financial small business customers. "Small business customers" are defined in line with the definition of loans extended to small businesses in paragraph 231 of the Basel II framework that are managed as retail exposures and are generally considered as having similar liquidity risk characteristics to retail accounts, provided the total aggregated funding raised from the small business customer is less than €1 million (on a consolidated basis where applicable) (Basel III LCR standards paragraph 90).

"Aggregated funding" means the gross amount (ie not netting any form of credit extended to the legal entity) of all forms of funding (eg deposits or debt securities or similar derivative exposure for which the counterparty is known to be a small business customer) (Basel III LCR standards footnote 41).

Applying the limit on a consolidated basis means that where one or more small business customers are affiliated with each other, they may be considered as a single creditor such that the limit is applied to the total funding received by the bank from this group of customers (Basel III LCR standards footnote 41).

Where a bank does not have any exposure to a small business customer that would enable it to use the definition under paragraph 231 of the Basel II framework, the bank may include such a deposit in this category provided that the total aggregate funding raised from the customer is less than €1 million (on a consolidated basis where applicable) and the deposit is managed as a retail deposit. This means that the bank treats such deposits in its internal risk management systems consistently over time and in the same manner as other retail deposits, and that the deposits are not individually managed in a way comparable to larger corporate deposits.

Term deposits from small business customers with a residual contractual maturity of greater than 30 days which can be withdrawn within 30 days without a significant withdrawal penalty materially greater than the loss of interest should be considered to fall within the 30-day horizon and should also be included in lines 117 to 137 as appropriate. If a portion of the term deposit can be withdrawn without incurring such a penalty, only that portion should be treated as a demand deposit. The remaining balance of the deposit should be treated as a term deposit.

112	Total unsecured wholesale funding		85–111
113	Total funding provided by small business customers; of which:	Total small business customer deposits as defined above.	89–92
114	Insured deposits; of which:	The portion of deposits or other forms of unsecured wholesale funding which are provided by non-financial small business customers and are non-maturing or mature within 30 days that are fully insured by an effective deposit insurance scheme.	89, 75–78
115	in transactional accounts; of which:	Total insured small business customer deposits in transactional accounts (eg accounts where salaries are paid out from).	89, 75, 78
116	eligible for a 3% run-off rate; of which:	The amount of insured transactional small business customer deposits that are in jurisdictions where the supervisor chooses to apply a 3% run-off rate given the deposits are fully insured by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards. Please refer to the instructions from your supervisor for the specification of these items.	89, 78
117	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 116, the amount that are in the reporting bank's home jurisdiction.	89, 78
118	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 116, the amount that are not in the reporting bank's home jurisdiction.	89, 78
119	eligible for a 5% run-off rate; of which:	The amount of insured transactional small business customer deposits that are in jurisdictions where the supervisor does not choose to apply a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	89, 75

Row	Heading	Description	Basel III LCR standards reference
120	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 119, the amount that are in the reporting bank's home jurisdiction.	89, 75
121	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 119, the amount that are not in the reporting bank's home jurisdiction.	89, 75
122	in non-transactional accounts with established relationships that make deposit withdrawal highly unlikely; of which:	Total insured small business customer deposits in non-transactional accounts where the customer has another relationship with the bank that would make deposit withdrawal highly unlikely.	89, 75, 78
123	eligible for a 3% run-off rate; of which:	The amount of insured non-transactional established relationship small business customer deposits that are in jurisdictions where the supervisor chooses to apply a 3% runoff rate given the deposits are fully insured by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards. Please refer to the instructions from your supervisor for the specification of these items.	89, 78
124	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 123, the amount that are in the reporting bank's home jurisdiction.	89, 78
125	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 123, the amount that are not in the reporting bank's home jurisdiction.	89, 78
126	eligible for a 5% run-off rate; of which:	The amount of insured non-transactional established relationship small business customer deposits that are in jurisdictions where the supervisor does not choose to apply a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	89, 75
127	are in the reporting bank's home jurisdiction	Of the deposits referenced in line 126, the amount that are in the reporting bank's home jurisdiction.	89, 75
128	are not in the reporting bank's home jurisdiction	Of the deposits referenced in line 126, the amount that are not in the reporting bank's home jurisdiction.	89, 75
129	in non-transactional and non-relationship accounts	Insured small business customer deposits in non-transactional accounts, where the customer does not have another relationship with the bank that would make deposit withdrawal highly unlikely.	89, 79
130	Uninsured deposits	The portion of small business customer deposits that are non-maturing or mature within 30 days, that are not fully insured by an effective deposit insurance scheme (ie all small business customer deposits not reported in lines 117 to 129, excluding any reported in lines 132 to 134).	89, 79
131	Additional deposit categories with higher run-off rates as specified by supervisor	Other small business customer deposits, as defined by supervisor. Amounts in these categories should not be included in the lines above.	89, 79
132	Category 1	As defined by supervisor.	89, 79
133	Category 2	As defined by supervisor.	89, 79
134	Category 3	As defined by supervisor.	89, 79
135	Term deposits (treated as having >30 day maturity); of which:	Small business customer deposits with a residual maturity or withdrawal notice period of greater than 30 days where the depositor has no legal right to withdraw deposits within 30 days, or if early withdrawal is allowed, would result in a significant penalty that is materially greater than the loss of interest.	92, 82-84

Row	Heading	Description	Basel III LCR standards reference
136	With a supervisory run-off rate	As defined by supervisor.	92, 84
137	Without supervisory run-off rate	All other term small business customer deposits treated as having > 30 day remaining maturity as defined in line 135.	92, 82

Unsecured wholesale funding generated by clearing, custody and cash management activities ("operational deposits"):

Reported in lines 140 to 154 are portions of deposits and other extensions of funds from financial and non-financial wholesale customers (excluding deposits less than €1 million from small business customers which are reported in lines 117 to 137) generated out of clearing, custody and cash management activities ("operational deposits"). These funds may receive a 25% run-off factor only if the customer has a substantive dependency with the bank and the deposit is required for such activities.

Qualifying activities in this context refer to clearing, custody or cash management activities that meet the following criteria:

- The customer is reliant on the bank to perform these services as an independent third party intermediary in order to fulfil its normal banking activities over the next 30 days. For example, this condition would not be met if the bank is aware that the customer has adequate back-up arrangements.
- These services must be provided under a legally binding agreement to institutional customers.
- The termination of such agreements shall be subject either to a notice period of at least 30 days or significant switching costs (such as those related to transaction, information technology, early termination or legal costs) to be borne by the customer if the operational deposits are moved before 30 days.

Qualifying operational deposits generated by such an activity are ones where:

- The deposits are by-products of the underlying services provided by the banking organisation and not sought out in the wholesale market in the sole interest of offering interest income.
- The deposits are held in specifically designated accounts and priced without giving an economic incentive to the customer (not limited to paying market interest rates) to leave any excess funds on these accounts. In the case that interest rates in a jurisdiction are close to zero, it would be expected that such accounts are non-interest bearing.

Any excess balances that could be withdrawn and would still leave enough funds to fulfil these clearing, custody and cash management activities do not qualify for the 25% factor. In other words, only that part of the deposit balance with the service provider that is proven to serve a customer's operational needs can qualify as stable. Excess balances should be treated in the appropriate category for non-operational deposits. If banks are unable to determine the amount of the excess balance, then the entire deposit should be assumed to be excess to requirements and, therefore, considered non-operational.

Deposits arising out of correspondent banking or from the provision of prime brokerage services (as defined in Basel III LCR standards footnote 42) should not be reported in these lines rather as non-operational deposits in lines 157 to 164 as appropriate (Basel III LCR standards paragraph 99) and lines 170 and 172, respectively.

A clearing relationship, in this context, refers to a service arrangement that enables customers to transfer funds (or securities) indirectly through direct participants in domestic settlement systems to final recipients. Such services are limited to the following activities: transmission, reconciliation and confirmation of payment orders; daylight overdraft, overnight financing and maintenance of post-settlement balances; and determination of intra-day and final settlement positions. (Basel III LCR standards, paragraph 101)

A custody relationship, in this context, refers to the provision of safekeeping, reporting, processing of assets or the facilitation of the operational and administrative elements of related activities on behalf of customers in the process of their transacting and retaining financial assets. Such services are limited to the settlement of securities transactions, the transfer of contractual payments, the processing of collateral, and the provision of custody related cash management services. Also included are the receipt of dividends and other income, client subscriptions and redemptions. Custodial services can furthermore extend to asset and corporate trust servicing, treasury, escrow, funds transfer, stock transfer and agency services, including payment and settlement services (excluding correspondent banking), and depository receipts. (Basel III LCR standards, paragraph 102)

A cash management relationship, in this context, refers to the provision of cash management and related services to customers. Cash management services, in this context, refers to those products and services provided to a customer to manage its cash flows, assets and liabilities, and conduct financial transactions necessary to the customer's ongoing operations. Such services are limited to payment remittance, collection and aggregation of funds, payroll administration, and control over the disbursement of funds. (Basel III LCR standards, paragraph 103)

Row	Heading	Description	Basel III LCR standards reference
138	Total operational deposits; of which:	The portion of unsecured operational wholesale funding generated by clearing, custody and cash management activities as defined above.	93–104
139	provided by non-financial corporates	Such funds provided by non-financial corporates. Funds from small business customers that meet the requirements outlined in paragraphs 90 and 91 of the Basel III LCR standards should not be reported here but are subject to lower run-off rates in rows 117 to 130.	93–104
140	insured, with a 3% run-off rate	The portion of such funds provided by non-financial corporates that are fully covered by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards and are in jurisdictions where the supervisor chooses to prescribe a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
141	insured, with a 5% run-off rate	The portion of such funds provided by non-financial corporates that are fully covered by an effective deposit insurance scheme but that are not prescribed a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
142	uninsured	The portion of such funds provided by non-financial corporates that are not fully covered by an effective deposit insurance scheme.	93–103
143	provided by sovereigns, central banks, PSEs and MDBs	Such funds provided by sovereigns, central banks, PSEs and multilateral development banks.	93–104
144	insured, with a 3% run-off rate	The portion of such funds provided by sovereigns, central banks, PSEs and multilateral development banks that are fully covered by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards and are in jurisdictions where the supervisor chooses to prescribe a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
145	insured, with a 5% run-off rate	The portion of such funds provided by sovereigns, central banks, PSEs and multilateral development banks that are fully covered by an effective deposit insurance scheme but that are not prescribed a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
146	uninsured	The portion of such funds provided by sovereigns, central banks, PSEs and multilateral development banks that are not fully covered by an effective deposit insurance scheme.	93–103
147	provided by banks	Such funds provided by banks.	93–104
148	insured, with a 3% run-off rate	The portion of such funds provided by banks that are fully covered by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards and are in jurisdictions where the supervisor chooses to prescribe a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104

Row	Heading	Description	Basel III LCR standards reference
149	insured, with a 5% run-off rate	The portion of such funds provided by banks that are fully covered by an effective deposit insurance scheme but that are not prescribed a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
150	uninsured	The portion of such funds provided by banks that are not fully covered by an effective deposit insurance scheme.	93–103
151	provided by other financial institutions and other legal entities	Such funds provided by financial institutions (other than banks) and other legal entities.	93–104
152	insured, with a 3% run-off rate	The portion of such funds provided by financial institutions (other than banks) and other legal entities that are fully covered by an effective deposit insurance scheme that meets the conditions outlined in paragraph 78 of the Basel III LCR standards and are in jurisdictions where the supervisor chooses to prescribe a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
153	insured, with a 5% run-off rate	The portion of such funds provided by financial institutions (other than banks) and other legal entities that are fully covered by an effective deposit insurance scheme but that are not prescribed a 3% run-off rate. Please refer to the instructions from your supervisor for the specification of these items.	104
154	uninsured	The portion of such funds provided by financial institutions (other than banks) and other legal entities that are not fully covered by an effective deposit insurance scheme.	93–103

Non-operational deposits in lines 157 to 164 include all deposits and other extensions of unsecured funding not included under operational deposits in lines 140 to 154, excluding notes, bonds and other debt securities, covered bond issuance or repo and secured funding transactions (reported below). Deposits arising out of correspondent banking or from the provision of prime brokerage services (as defined in the Basel III LCR standards, footnote 42) should **not** be included in these lines (Basel III LCR standards, paragraph 99).

Customer cash balances arising from the provision of prime brokerage services, including but not limited to the cash arising from prime brokerage services as identified in Basel III LCR standards, paragraph 99, should be considered separate from any required segregated balances related to client protection regimes imposed by national regulations, and should not be netted against other customer exposures included in this standard. These offsetting balances held in segregated accounts are treated as inflows in Basel III LCR standards, paragraph 154 and should be excluded from the stock of HQLA (Basel III LCR standards, paragraph 111).

155	Total non-operational deposits; of which	The portion of unsecured wholesale funding not considered as "operational deposits" as defined above.	105–109
156	provided by non-financial corporates; of which:	Total amount of such funds provided by non-financial corporates.	107–108
157	where entire amount is fully covered by an effective deposit insurance scheme	Amount of such funds provided by non-financial corporates where the entire amount of the deposit is fully covered by an effective deposit insurance scheme.	108
158	where entire amount is not fully covered by an effective deposit insurance scheme	Amount of such funds provided by non-financial corporates where the entire amount of the deposit is not fully covered by an effective deposit insurance scheme.	107
159	provided by sovereigns, central banks, PSEs and MDBs; of which:	Such funds provided by sovereigns, central banks (other than funds to be reported in line item 166), PSEs, and multilateral development banks.	107-108
160	where entire amount is fully covered by an effective deposit insurance scheme	Amount of such funds provided by sovereigns, central banks, PSEs and MDBs where the entire amount of the deposit is fully covered by an effective deposit insurance scheme.	108

Row	Heading	Description	Basel III LCR standards reference
161	where entire amount is not fully covered by an effective deposit insurance scheme	Amount of such funds provided by sovereigns, central banks, PSEs and MDBs where the entire amount of the deposit is not fully covered by an effective deposit insurance scheme.	107
162	provided by members of institutional networks of cooperative (or otherwise named) banks	An institutional network of cooperative (or otherwise named) banks is a group of legally autonomous banks with a statutory framework of cooperation with common strategic focus and brand where specific functions are performed by central institutions or specialised service providers. Central institutions or specialised central service providers of such networks should report in this line the amount of deposits placed by network member institutions (that are not reported in line items 149 or 150 and that are) (a) due to statutory minimum deposit requirements which are registered at regulators or (b) in the context of common task sharing and legal, statutory or contractual arrangements so long as both the bank that has received the monies and the bank that has deposited participate in the same institutional network's mutual protection scheme against illiquidity and insolvency of its members. Deposits from network member institutions that are neither included in line items 149 or 150, nor placed for purposes as referred to in letters (a) and (b) above, are to be reported in line items 163 or 164. Banks that are not the central institutions or specialised central service provider of such network should report zero in this line.	105
163	provided by other banks	Such funds provided by other banks, not reported in line 162.	109
164	provided by other financial institutions and other legal entities	Such funds provided by financial institutions other than banks and by other legal entities not included in the categories above. Funding from fiduciaries, beneficiaries, conduits and special purpose vehicles and affiliated entities should also be reported here.	109
bond is retail), i	sold exclusively in the retail mark n which case the instruments can	ssued by the bank are included in line 165 regardless of the holder ket and held in retail accounts (including small business customers a be reported in the appropriate retail or small business customer do 137, respectively. Outflows on covered bonds should be reported	treated as eposit
165	Unsecured debt issuance	Outflows on notes, bonds and other debt securities, excluding on bonds sold exclusively to the retail or small business customer markets, and excluding outflows on covered bonds.	110
166	Additional balances required to be installed in central bank reserves	Amounts to be installed in the central bank reserves within 30 days. Funds reported in this line should not be included in line 160 or 161. Please refer to the instructions from your supervisor for the specification of this item.	Extension of 50(b)
169	Of the non-operational deposits reported above, amounts that could be considered operational in nature but per the standards have been excluded from receiving the operational deposit treatment due to:		

Row	Heading	Description	Basel III LCR standards reference
170	correspondent banking activity	Amounts in accounts with a clearing, custody or cash management relationship but which have been excluded from the operational deposit category because the account is a correspondent banking account.	99, footnote 42
		Correspondent banking refers to arrangements under which one bank (correspondent) holds deposits owned by other banks (respondents) and provides payment and other services in order to settle foreign currency transactions (eg so-called nostro and vostro accounts used to settle transactions in a currency other than the domestic currency of the respondent bank for the provision of clearing and settlement of payments).	
172	prime brokerage services	Amounts in accounts with a clearing, custody or cash management relationship but which have been excluded from the operational deposit category because the account holder is a prime brokerage client of the reporting institution. Prime brokerage is a package of services offered to large active investors, particularly hedge funds.	99, footnote 42
174	excess balances in operational accounts that could be withdrawn and would leave enough funds to fulfil the clearing, custody and cash management activities	Amounts in accounts with a clearing, custody or cash management relationship but which have been excluded from the operational deposit category because these funds are excess balances and could be withdrawn and would leave enough funds to fulfil the clearing, custody and cash management activities.	96

c) Secured funding run-off

Secured funding is defined as those liabilities and general obligations that are collateralised by legal rights to specifically designated assets owned by the borrowing institution in the case of bankruptcy, insolvency, liquidation or resolution. In this section any transaction in which the bank has received a collateralised loan in cash, such as repo transactions, expiring within 30 days should be reported. Collateral swaps where the bank receives a collateralised loan in the form of other assets than cash, should not be reported here, but in panel C below.

Additionally, collateral lent to the bank's customers to effect short positions should be treated as a form of secured funding. A customer short position in this context describes a transaction where a bank's customer sells a security it does not own, and the bank subsequently obtains the same security from internal or external sources to make delivery into the sale. Internal sources include the bank's own inventory of collateral as well as rehypothecatable Level 1 or Level 2 collateral held in other customer margin accounts. The contingent risk associated with non-contractual obligations where customer short positions are covered by other customers' collateral that does not qualify as Level 1 or Level 2 should be reported in line 264. External sources include collateral obtained through a securities borrowing, reverse repo, or like transaction.

If the bank has deposited both liquid and non-liquid assets in a collateral pool and no assets are specifically assigned as collateral for the secured transaction, the bank may assume for this monitoring exercise that the assets with the lowest liquidity get assigned first: assets that are not eligible for the stock of liquid assets are assumed to be assigned first. Only once all those assets are fully assigned should Level 2B assets be assumed to be assigned, followed by Level 2A assets. Only once all Level 2 assets are assigned should Level 1 assets be assumed to be assigned.

A bank should report all outstanding secured funding transactions with remaining maturities within the 30 calendar day stress horizon, including customer short positions that do not have a specified contractual maturity. The amount of funds raised through the transaction should be reported in column D ("amount received"). The value of the underlying collateral extended in the transaction should be reported in column E ("market value of extended collateral"). Both values are needed to calculate the caps on Level 2 and Level 2B assets and both should be calculated at the date of reporting, not the trade or settlement date of the transaction.

Please refer to the instructions from your supervisor for the specification of items related to Level 2B assets in this subsection.

Row	Heading	Description	Basel III LCR standards reference
178	Transactions conducted with the bank's domestic central bank; of which:	In column D: Amount raised in secured funding or repo transactions with the bank's domestic central bank that mature within 30 days. In column E: The market value of the collateral extended on these transactions.	114–115
179	Backed by Level 1 assets; of which:	In column D: Amount raised in secured funding or repo transactions with the bank's domestic central bank that mature within 30 days and are backed by Level 1 assets. In column E: The market value of the Level 1 asset collateral extended on these transactions.	114–115
180	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 179, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 1 assets where these assets would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (if they were not already securing the particular transaction in question), because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 1 asset collateral extended on these transactions.	114–115
182	Backed by Level 2A assets; of which:	In column D: Amount raised in secured funding or repo transactions with the bank's domestic central bank that mature within 30 days and are backed by Level 2A assets. In column E: The market value of the Level 2A asset collateral extended on these transactions.	114–115
183	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 182, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2A assets where these assets would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2A asset collateral extended on these transactions.	114–115
185	Backed by Level 2B RMBS assets; of which:	In column D: Amount raised in secured funding or repo transactions with the bank's domestic central bank that mature within 30 days and are backed by Level 2B RMBS assets. In column E: The market value of the Level 2B RMBS asset collateral extended on these transactions.	114–115

Row	Heading	Description	Basel III LCR standards reference
186	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 185, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2B RMBS assets where these assets would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2B RMBS asset collateral extended on these transactions.	114–115
188	Backed by Level 2B non- RMBS assets; of which:	In column D: Amount raised in secured funding or repo transactions with the bank's domestic central bank that mature within 30 days and are backed by Level 2B non-RMBS assets. In column E: The market value of the Level 2B non-RMBS asset collateral extended on these transactions.	114–115
189	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 188, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2B non-RMBS assets where these assets would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2B non-RMBS asset collateral extended on these transactions.	114–115
191	Backed by other assets	In column D: Amount raised on secured funding or repo transactions with the bank's domestic central bank that mature within 30 days and are backed by all other assets (ie other than Level 1 or Level 2 assets). In column E: The market value of the other asset collateral extended on these transactions.	114–115
192	Transactions not conducted with the bank's domestic central bank and backed by Level 1 assets; of which:	In column D: Amount raised in secured funding or repo transactions that are not conducted with the bank's domestic central bank and that mature within 30 days and are backed by Level 1 assets. In column E: The market value of the Level 1 asset collateral extended on these transactions.	114–115
193	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 192, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 1 assets where these assets would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (if they were not already securing the particular transaction in question), because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 1 asset collateral extended on these transactions.	114–115

Row	Heading	Description	Basel III LCR standards reference
195	Transactions not conducted with the bank's domestic central bank and backed by Level 2A assets; of which:	In column D: Amount raised in secured funding or repo transactions that are not conducted with the bank's domestic central bank and that mature within 30 days and are backed by Level 2A assets. In column E: The market value of the Level 2A asset collateral extended on these transactions.	114–115
196	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 195, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2A assets where these assets would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2A asset collateral extended on these transactions.	114–115
198	Transactions not conducted with the bank's domestic central bank and backed by Level 2B RMBS assets; of which:	In column D: Amount raised in secured funding or repo transactions that are not conducted with the bank's domestic central bank and that mature within 30 days and are backed by Level 2B RMBS assets. In column E: The market value of the Level 2B RMBS asset collateral extended on these transactions.	114–115
199	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 198, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2B RMBS assets where these assets would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2B RMBS asset collateral extended on these transactions.	114–115
201	Transactions not conducted with the bank's domestic central bank and backed by Level 2B non-RMBS assets; of which:	In column D: Amount raised in secured funding or repo transactions that are not conducted with the bank's domestic central bank and that mature within 30 days and are backed by Level 2B non-RMBS assets. In column E: The market value of the Level 2B non-RMBS asset collateral extended on these transactions.	114–115
202	Counterparties are domestic sovereigns, MDBs or domestic PSEs with a 20% risk weight; of which:	In column D: Secured funding transactions with domestic sovereign, multilateral development banks or domestic PSEs that are backed by Level 2B non-RMBS assets. PSEs that receive this treatment should be limited to those that are 20% or lower risk weighted. In column E: The market value of collateral extended on these transactions.	114–115

Row	Heading	Description	Basel III LCR standards reference
203	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 202, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2B non-RMBS assets where these assets would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2B non-RMBS asset collateral extended on these transactions.	114–115
205	Counterparties are not domestic sovereigns, MDBs or domestic PSEs with a 20% risk weight; of which:	In column D: Secured funding transactions with counterparties other than domestic sovereign, multilateral development banks or domestic PSEs with a 20% risk weight that are backed by Level 2B non-RMBS assets. In column E: The market value of collateral extended on these transactions.	114–115
206	Transactions involving eligible liquid assets	In column D: Of the amount reported in line 205, that which is raised in secured funding or repo transactions that mature within 30 days and are backed by Level 2B non-RMBS assets where these assets would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (if they were not already securing the particular transaction in question) because: (i) they would be held unencumbered; and (ii) they would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column E: The market value of the Level 2B non-RMBS asset collateral extended on these transactions.	114–115
208	Transactions not conducted with the bank's domestic central bank and backed by other assets (non-HQLA); of which:	In column D: Amount raised in secured funding or repo transactions that are not conducted with the bank's domestic central bank and that mature within 30 days and are backed by other assets (non-HQLA). In column E: The market value of the other (non-HQLA) asset collateral extended on these transactions.	114–115
209	Counterparties are domestic sovereigns, MDBs or domestic PSEs with a 20% risk weight; of which:	In column D: Secured funding transactions with domestic sovereign, multilateral development banks or domestic PSEs that are backed by other assets (non-HQLA). PSEs that receive this treatment should be limited to those that are 20% or lower risk weighted. In column E: The market value of collateral extended on these transactions.	114–115
210	Counterparties are not domestic sovereigns, MDBs or domestic PSEs with a 20% risk weight; of which:	In column D: Secured funding transactions with counterparties other than domestic sovereign, multilateral development banks or PSEs that are backed by other assets (non-HQLA). In column E: The market value of collateral extended on these transactions.	114–115

Row	Heading	Description	Basel III LCR standards reference
d)	Additional requirements		
214	Derivatives cash outflow	Banks should calculate, in accordance with their existing valuation methodologies, expected contractual derivative cash inflows and outflows. Cash flows may be calculated on a net basis (ie inflows can offset outflows) by counterparty, only where a valid master netting agreement exists. The sum of all net cash outflows should be reported here. The sum of all net cash inflows should be reported in line 316. Banks should exclude from such calculations those liquidity requirements that would result from increased collateral needs due to market value movements (to be reported in line 222) or falls in value of collateral posted (reported in line 217 and line 218). Options should be assumed to be exercised when they are 'in the money' to the option buyer. Where derivative payments are collateralised by HQLA, cash outflows should be calculated net of any corresponding cash or collateral inflows that would result, all other things being equal, from contractual obligations for cash or collateral to be provided to the bank, if the bank is legally entitled and operationally capable to re-use the collateral in new cash raising transactions once the collateral is received. This is in line with the principle that banks should not double count liquidity inflows and outflows. Note that cash flows do not equal the marked-to-market value, since the marked-to-market value also includes estimates for contingent inflows and outflows and may include cash flows that occur beyond the 30-day horizon. It is generally expected that a positive amount would be provided for both this line item and line 316 for institutions engaged in derivatives transactions.	116, 117
215	Increased liquidity needs related to downgrade triggers in derivatives and other financing transactions	The amount of collateral that would need to be posted for or contractual cash outflows generated by any downgrade up to and including a 3-notch downgrade of the bank's long-term credit rating. Triggers linked to a bank's short-term rating should be assumed to be triggered at the corresponding long-term rating in accordance with published ratings criteria. The impact of the downgrade should consider impacts on all types of margin collateral and contractual triggers which change rehypothecation rights for non-segregated collateral.	118
216	Increased liquidity needs related to the potential for valuation changes on posted collateral securing derivative and other transactions:		119
217	Cash and Level 1 assets	Current market value of relevant collateral posted as margin for derivatives and other transactions that, if they had been unencumbered, would have been eligible for inclusion in line items 6 to 18.	

Row	Heading	Description	Basel III LCR standards reference
218	For other collateral (ie all non-Level 1 collateral)	Current market value of relevant collateral posted as margin for derivatives and other transactions other than those included in line item 217 (all non-Level 1 collateral). This amount should be calculated net of collateral received on a counterparty basis (provided that the collateral received is not subject to restrictions on reuse or rehypothecation). Any collateral that is in a segregated margin account can only be used to offset outflows that are associated with payments that are eligible to be offset from that same account.	
219	Increased liquidity needs related to excess nonsegregated collateral held by the bank that could contractually be called at any time by the counterparty	The amount of non-segregated collateral that the reporting institution currently has received from counterparties but could under legal documentation be recalled because the collateral is in excess of that counterparty's current collateral requirements.	120
220	Increased liquidity needs related to contractually required collateral on transactions for which the counterparty has not yet demanded the collateral be posted	The amount of collateral that is contractually due from the reporting institution, but for which the counterparty has not yet demanded the posting of such collateral.	121
221	Increased liquidity needs related to contracts that allow collateral substitution to non-HQLA assets	The amount of HQLA collateral that can be substituted for non-HQLA without the bank's consent that has been received to secure transactions and that has not been segregated (eg otherwise included in HQLAs, as secured funding collateral or in other bank operations).	122
222	Increased liquidity needs related to market valuation changes on derivative or other transactions	Any potential liquidity needs deriving from full collateralisation of mark-to-market exposures on derivative and other transactions. Unless its national supervisor has provided other instructions, banks should calculate any outflow generated by increased needs related to market valuation changes by identifying the largest absolute net 30-day collateral flow realised during the preceding 24 months, where the absolute net collateral flow is based on both realised outflows and inflows. Inflows and outflows of transactions executed under the same master netting agreement can be treated on a net basis.	123
223	Loss of funding on ABS and other structured financing instruments issued by the bank, excluding covered bonds	Balances of term asset-backed securities and other structured financing instruments, excluding covered bonds (which should be reported in line 228), issued by the bank that mature in 30 days or less. To the extent that sponsored conduits/SPVs are required to be consolidated under liquidity requirements, their assets and liabilities should be taken into account.	124

Row	Heading	Description	Basel III LCR standards reference
224	Loss of funding on ABCP, conduits, SIVs and other such financing activities; of which:	All funding on asset-backed commercial paper, conduits, securities investment vehicles and other such financing facilities maturing or returnable within 30 days. Banks having structured financing facilities that include the issuance of short-term debt instruments, such as asset backed commercial paper, should report the potential liquidity outflows from these structures. These include, but are not limited to, (i) the inability to refinance maturing debt, and (ii) the existence of derivatives or derivative-like components contractually written into the documentation associated with the structure that would allow the "return" of assets in a financing arrangement, or that require the original asset transferor to provide liquidity, effectively ending the financing arrangement ("liquidity puts") within the 30-day period. Where the structured financing activities are conducted through a special purpose entity (such as a special purpose vehicle, conduit or SIV), the bank should, in determining the HQLA requirements, look through to the maturity of the debt instruments issued by the entity and any embedded options in financing arrangements that may potentially trigger the "return" of assets or the need for liquidity, irrespective of whether or not the SPV is consolidated.	125
225	debt maturing ≤ 30 days	Portion of the funding specified in line 224 maturing within 30 days.	125
226	with embedded options in financing arrangements	Portion of the funding specified in line 224 not maturing within 30 days but with embedded options that could reduce the effective maturity of the debt to 30 days or less.	125
227	other potential loss of such funding	Portion of the funding specified in line 224 that is not included in line 225 or 226.	125
228	Loss of funding on covered bonds issued by the bank	Balances of covered bonds, issued by the bank that mature in 30 days or less.	124

Credit and liquidity facilities are defined as explicit contractual agreements or obligations to extend funds at a future date to retail or wholesale counterparties. For the purpose of the standard, these facilities only include contractually irrevocable ("committed") or conditionally revocable agreements to extend funds in the future (Basel III LCR standards, paragraph 126).

Unconditionally revocable facilities that are unconditionally cancellable by the bank (in particular, those without a precondition of a material change in the credit condition of the borrower) are excluded from this section and should be reported in lines 254 to 266, as appropriate (Basel III LCR standards, paragraph 126).

The currently undrawn portion of these facilities should be reported. The reported amount may be net of any HQLAs eligible for the stock of HQLAs, if the HQLAs have already been posted as collateral by the counterparty to secure the facilities or that are contractually obliged to be posted when the counterparty will draw down the facility (eg a liquidity facility structured as a repo facility), if the bank is legally entitled and operationally capable to re-use the collateral in new cash raising transactions once the facility is drawn, and there is no undue correlation between the probability of drawing the facility and the market value of the collateral. The collateral can be netted against the outstanding amount of the facility to the extent that this collateral is not already counted in the stock of HQLAs (Basel III LCR standards, paragraph 127).

A liquidity facility is defined as any committed, undrawn back-up facility that would be utilised to refinance the debt obligations of a customer in situations where such a customer is unable to rollover that debt in financial markets (eg pursuant to a commercial paper programme, secured financing transactions, obligations to redeem units, etc).

The amount of a commitment to be treated as a liquidity facility is the amount of the currently outstanding debt issued by the customer (or proportionate share, if a syndicated facility) maturing within a 30 day period that is backstopped by the facility. The portion of a liquidity facility that is backing debt that does not mature within the 30-day window is excluded from the scope of the definition of a facility. Any additional capacity of the facility (ie the remaining commitment) would be treated as a committed credit facility and should be reported as such.

General working capital facilities for corporate entities (eg revolving credit facilities in place for general corporate and/or working capital purposes) will not be classified as liquidity facilities, but as credit facilities.

Row	Heading	Description	Basel III LCR standards reference
vehicles to finan For that maturin liquidity	, for example SPEs (as defined in ce the banks own assets, should portion of financing programs t g or have liquidity puts that may	s provided to hedge funds, money market funds and special purpo the Basel III LCR standards, paragraph 125) or conduits, or other v be captured in their entirety as a liquidity facility and reported in li hat are captured in the Basel III LCR standards, paragraphs 124 and be exercised in the 30-day horizon), banks that are providers of as count the maturing financing instrument and the liquidity facility f	ehicles used ne 239. I 125 (ie are ssociated
229	Undrawn committed credit and liquidity facilities to retail and small business customers	Balances of undrawn committed credit and liquidity facilities extended by the bank to natural persons and small business customers, as defined above.	131(a)
230	Undrawn committed credit facilities to		
231	non-financial corporates	Balances of undrawn committed credit facilities extended by the bank to non-financial institution corporations (excluding small business customers). The amount reported in this line should also include any 'additional capacity' of liquidity facilities (as defined above) provided to non-financial corporates.	131(b)
232	sovereigns, central banks, PSEs and MDBs	Balances of undrawn committed credit facilities extended by the bank to sovereigns, central banks, PSEs, multilateral development banks and any other entity not included in other drawdown categories. The amount reported in this line should also include any 'additional capacity' of liquidity facilities (as defined above) provided to sovereigns, central banks, PSEs, multilateral development banks.	131(b)
233	Undrawn committed liquidity facilities to		
234	non-financial corporates	The amount of undrawn committed liquidity facilities should be the amount of currently outstanding debt (or proportionate share if a syndicated facility) issued by non-financial institution corporations (excluding small business customers) maturing within a 30 day period that is backstopped by the facility. Any 'additional capacity' of liquidity facilities (as defined above) provided to non-financial corporates should not be reported here, rather should be reported in line 231.	131(c)
235	sovereigns, central banks, PSEs and MDBs	The amount of undrawn committed liquidity facilities should be the amount of currently outstanding debt (or proportionate share if a syndicated facility) issued by sovereigns, central banks, PSEs, or multilateral development banks maturing within a 30 day period that is backstopped by the facility. Any 'additional capacity' of liquidity facilities (as defined above) provided to sovereigns, central banks, PSEs, or multilateral development banks should not be reported here, rather should be reported in line 232.	131(c)
236	Undrawn committed credit and liquidity facilities provided to banks subject to prudential supervision	Balances of undrawn committed credit and liquidity facilities extended to banks that are subject to prudential supervision.	131(d)

Row	Heading	Description	Basel III LCR standards reference
237	Undrawn committed credit facilities provided to other FIs	Balances of undrawn committed credit facilities extended by the bank to other financial institutions (including securities firms, insurance companies, fiduciaries and beneficiaries). The amount reported in this line should also include any 'additional capacity' of liquidity facilities (as defined above) provided to other financial institutions (including securities firms, insurance companies, fiduciaries and beneficiaries).	131(e)
238	Undrawn committed liquidity facilities provided to other FIs	The amount of undrawn committed liquidity facilities should be the amount of currently outstanding debt (or proportionate share if a syndicated facility) issued by to other financial institutions (including securities firms, insurance companies, fiduciaries and beneficiaries) maturing within a 30 day period that is backstopped by the facility. Any 'additional capacity' of liquidity facilities (as defined above) provided to other financial institutions (including securities firms, insurance companies, fiduciaries and beneficiaries) should not be reported here, rather should be reported in line 237.	131(f)
239	Undrawn committed credit and liquidity facilities to other legal entities	Balances of undrawn committed credit and liquidity facilities extended to other legal entities, including hedge funds, money market funds and special purpose funding vehicles, for example SPEs (as defined in the Basel III LCR standards, paragraph 125) or conduits, or other vehicles used to finance the banks own assets (not included in lines 229 to 238).	131(g)
Other c	ontractual obligations to extend	funds	ı
241	Other contractual obligations to extend funds to:	Any contractual lending obligations not captured elsewhere in the standard.	132-133
242	financial institutions	Any contractual lending obligations to financial institutions not captured elsewhere.	132
243	retail clients	The full amount of contractual obligations to extend funds to retail clients within the next 30 calendar days (not netted for the assumed roll-over on the inflows in line 302).	133
244	small business customers	The full amount of contractual obligations to extend funds to small business customers within the next 30 calendar days (not netted for the assumed roll-over on the inflows in line 303).	133
245	non-financial corporates	The full amount of contractual obligations to extend funds to non-financial corporate clients within the next 30 calendar days (not netted for the assumed roll-over on the inflows in line 304).	133
246	other clients	The full amount of contractual obligations to extend funds to other clients within the next 30 calendar days (not netted for the assumed roll-over on the inflows in line 310).	133
247	retail, small business customers, non-financials and other clients	The amounts of contractual obligations to extend funds to retail, small business customers, non-financial corporate and other clients within the next 30 calendar days (lines 243 to 246) are added up in this line. The roll-over of funds that is implicitly assumed in the inflow section (lines 302, 303, 304 and 310) are then subtracted. If the result is positive, it is included here as an outflow in column H. Otherwise, the outflow included here is zero.	133

Row	Heading	Description	Basel III LCR
			standards
			reference

Other contingent funding obligations (treatment determined by national supervisor)

These contingent funding obligations may be either contractual or non-contractual and are not lending commitments. Non-contractual contingent funding obligations include associations with, or sponsorship of, products sold or services provided that may require the support or extension of funds in the future under stressed conditions. Non-contractual obligations may be embedded in financial products and instruments sold, sponsored, or originated by the institution that can give rise to unplanned balance sheet growth arising from support given for reputational risk considerations (Basel III LCR standards, paragraph 135). Stressed conditions in this context refer to the scenario as described in paragraph 19 of the Basel III LCR standards. Banks should report the full amount of any exposure and national supervisors should set appropriate outflow rates for their jurisdictions.

254	Non-contractual obligations related to potential liquidity draws from joint ventures or minority investments in entities	Non contractual contingent funding obligations related to potential liquidity draws from joint ventures or minority investments in entities, which are not consolidated per paragraph 164 of the Basel III LCR standards, where there is the expectation that the bank will be the main liquidity provider when the entity is in need of liquidity. The amount included should be calculated in accordance with the methodology agreed by the bank's supervisor. Please refer to the	137
		instructions from your supervisor for the specification of this item.	
255	Unconditionally revocable "uncommitted" credit and liquidity facilities	Balances of undrawn credit and liquidity facilities where the bank has the right to unconditionally revoke the undrawn portion of these facilities.	140
256	Trade-finance related obligations (including guarantees and letters of credit)	Trade finance instruments consist of trade-related obligations directly underpinned by the movement of goods or the provision of services. Amounts to be reported here include items such as: outstanding documentary trade letters of credit, documentary and clean collection, import bills, and export bills; and outstanding guarantees directly related to trade finance obligations, such as shipping guarantees. Lending commitments, such as direct import or export financing for non-financial corporate firms, are excluded from this treatment and reported in lines 229 to 239.	138, 139
257	Guarantees and letters of credit unrelated to trade finance obligations	The outstanding amount of letters of credit issued by the bank and guarantees unrelated to trade finance obligations described in line 256.	140
258	Non-contractual obligations:		
259	Debt-buy back requests (incl related conduits)	Potential requests for debt repurchases of the bank's own debt or that of related conduits, securities investment vehicles and other such financing facilities. In case debt amounts qualify for both line 259 and line 263, please enter them in just one of these lines.	140
260	Structured products	Structured products where customers anticipate ready marketability, such as adjustable rate notes and variable rate demand notes (VRDNs).	140
261	Managed funds	Managed funds that are marketed with the objective of maintaining a stable value such as money market mutual funds or other types of stable value collective investment funds etc.	140
262	Other non-contractual obligations	Any other non-contractual obligation not entered above.	140

Row	Heading	Description	Basel III LCR standards reference
263	Outstanding debt securities with remaining maturity > 30 days	For issuers with an affiliated dealer or market maker, there may be a need to include an amount of the outstanding debt securities (unsecured and secured, term as well as short term) having maturities greater than 30 calendar days, to cover the potential repurchase of such outstanding securities. In case debt amounts qualify for both line 259 and line 263, please enter them in just one of these lines.	140
264	Non contractual obligations where customer short positions are covered by other customers' collateral	Amount of contingent obligations related to instances where banks have internally matched client assets against other clients' short positions where the collateral does not qualify as Level 1 or Level 2, and the bank may be obligated to find additional sources of funding for these positions in the event of client withdrawals. Instances where the collateral qualifies as Level 1 or Level 2 should be reported in the appropriate line of the secured funding section (lines 192 to 206).	140
265	Bank outright short positions covered by a collateralised securities financing transaction	Amount of the bank's outright short positions that are being covered by collateralised securities financing transactions. Such short positions are assumed to be maintained throughout the 30-day period and receive a 0% outflow. The corresponding collateralised securities financing transactions that are covering such short positions should be reported in lines 291 to 296 or 406 to 430.	147
266	Other contractual cash outflows (including those related to unsecured collateral borrowings and uncovered short positions)	Any other contractual cash outflows within the next 30 calendar days should be captured in this standard, such as such as outflows to cover unsecured collateral borrowings, uncovered short positions, dividends or contractual interest payments, with explanation given in an accompanying note to your supervisor as to what comprises the amounts included in this line. This amount should exclude outflows related to operating costs.	141, 147

6.1.3 Inflows, Liquidity Coverage Ratio (LCR) (panel B2)

Row	Heading	Description	Basel III LCR
			standards
			reference

Total expected contractual cash inflows are calculated by multiplying the outstanding balances of various categories of contractual receivables by the rates at which they are expected to flow in under the scenario up to an aggregate cap of 75% of total expected cash outflows (Basel III LCR standards, paragraph 69).

Items must not be double counted – if an asset is included as part of the "stock of HQLA" (ie the numerator), the associated cash inflows cannot also be counted as cash inflows (ie part of the denominator) (Basel III LCR standards, paragraph 72).

When considering its available cash inflows, the bank should only include contractual inflows (including interest payments) from outstanding exposures that are fully performing and for which the bank has no reason to expect a default within the 30-day time horizon (Basel III LCR standards, paragraph 142). Pre-payments on loans (not due within 30 days) should not be included in the inflows.

Contingent inflows are not included in total net cash outflows (Basel III LCR standards, paragraph 142).

a) Secured lending including reverse repos and securities borrowing

Secured lending is defined as those loans that the bank has extended and are collateralised by legal rights to specifically designated assets owned by the borrowing institution, which the bank use or rehypothecate for the duration of the loan, and for which the bank can claim ownership to in the case of default by the borrower. In this section any transaction in which the bank has extended a collateralised loan in cash, such as reverse repo transactions, expiring within 30 days should be reported. Collateral swaps where the bank has extended a collateralised loan in the form of other assets than

Row	Heading	Description	Basel III LCR
			standards
			reference

cash, should not be reported here, but in panel C below.

A bank should report all outstanding secured lending transactions with remaining maturities within the 30 calendar day stress horizon. The amount of funds extended through the transaction should be reported in column D ("amount extended"). The value of the underlying collateral received in the transactions should be reported in column E ("market value of received collateral"). Both values are needed to calculate the caps on Level 2 and Level 2B assets and both should be calculated at the date of reporting, not the date of the transaction. Note that if the collateral received in the form of Level 1 or Level 2 assets is not rehypothecated and is legally and contractually available for the bank's use it should be reported in the appropriate lines of the stock of HQLA section (lines 11 to 40) as well as in this subsection (see paragraph 31 of the Basel III LCR standards).

Please refer to the instructions from your supervisor for the specification of items related to Level 2B assets in this subsection.

274	Reverse repo and other secured lending or securities borrowing transactions maturing ≤ 30 days	All reverse repo or securities borrowing transactions maturing within 30 days, in which the bank has extended cash and obtained collateral.	145–146
275	Of which collateral is not reused (ie is not rehypothecated) to cover the reporting institution's outright short positions	Such transactions in which the collateral obtained is not reused (ie is not rehypothecated) to cover the reporting institution's outright short positions. If the collateral is re-used, the transactions should be reported in lines 291 to 296.	145–146
276	Transactions backed by Level 1 assets	All such transactions in which the bank has obtained collateral in the form of Level 1 assets. These transactions are assumed to roll-over in full, not giving rise to any cash inflows. In column D: The amounts extended in these transactions. In column E: The market value of the Level 1 collateral received in these transactions.	145–146
277	Transactions involving eligible liquid assets	Of the transactions backed by Level 1 assets, those where the collateral obtained is reported in panel Aa of the "LCR" worksheet as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column D: The amounts extended in these transactions. In column E: The market value of the Level 1 collateral received in these transactions.	145–146
279	Transactions backed by Level 2A assets; of which:	All such transactions in which the bank has obtained collateral in the form of Level 2A assets. These are assumed to lead to a 15% cash inflow due to the reduction of funds extended against the collateral. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2A collateral received in these transactions.	145–146
280	Transactions involving eligible liquid assets	Of the transactions backed by Level 2A assets, those where the collateral obtained is reported in panel Ab of the "LCR" worksheet as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2A collateral received in these transactions.	145–146

Row	Heading	Description	Basel III LCR standards reference
282	Transactions backed by Level 2B RMBS assets; of which:	All such transactions in which the bank has obtained collateral in the form of Level 2B RMBS assets. These are assumed to lead to a 25% cash inflow due to the reduction of funds extended against the collateral. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2B RMBS collateral received in these transactions.	145–146
283	Transactions involving eligible liquid assets	Of the transactions backed by Level 2B RMBS assets, those where the collateral obtained is reported in panel Ac of the "LCR" worksheet as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2B RMBS collateral received in these transactions.	145–146
285	Transactions backed by Level 2B non-RMBS assets; of which:	All such transactions in which the bank has obtained collateral in the form of Level 2B non-RMBS assets. These are assumed to lead to a 50% cash inflow due to the reduction of funds extended against the collateral. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2B non-RMBS collateral received in these transactions.	145–146
286	Transactions involving eligible liquid assets	Of the transactions backed by Level 2B non-RMBS assets, those where the collateral obtained is reported in panel Ac of the "LCR" worksheet as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2B non-RMBS collateral received in these transactions.	145–146
288	Margin lending backed by non-Level 1 or non-Level 2 collateral	Collateralised loans extended to customers for the purpose of taking leveraged trading positions ("margin loans") made against non-HQLA collateral. These are assumed to lead to a 50% cash inflow. In column D: The amounts extended in these transactions. In column E: The market value of the collateral received in these transactions.	145–146
289	Transactions backed by other collateral	All such transactions (other than those reported in line 288) in which the bank has obtained collateral in another form than Level 1 or Level 2 assets. These are assumed not to roll over and therefore lead to a 100% cash inflow. In column D: The amounts extended in these transactions. In column E: The market value of the collateral received in these transactions.	145–146

Row	Heading	Description	Basel III LCR standards reference
290	Of which collateral is re-used (ie is rehypothecated) to cover the reporting institution's outright short positions	If the collateral obtained in these transactions is re-used (ie rehypothecated) to cover the reporting institution's outright short positions that could be extended beyond 30 days, it should be assumed that the transactions will be rolled-over and will not give rise to any cash inflows. This reflects the need to continue to cover the short position or to repurchase the relevant securities. Institutions should only report reverse repo amounts in these cells where it itself is short the collateral. If the collateral is not re-used, the transaction should be reported in lines 275 to 289.	145–146
291	Transactions backed by Level 1 assets	All such transactions in which the bank has obtained collateral in the form of Level 1 assets. In column D: The amounts extended in these transactions. In column E: The market value of the Level 1 collateral received in these transactions.	145–146
292	Transactions backed by Level 2A assets	All such transactions in which the bank has obtained collateral in the form of Level 2A assets. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2A collateral received in these transactions.	145–146
293	Transactions backed by Level 2B RMBS assets	All such transactions in which the bank has obtained collateral in the form of Level 2B RMBS assets. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2B RMBS collateral received in these transactions.	145–146
294	Transactions backed by Level 2B non-RMBS assets	All such transactions in which the bank has obtained collateral in the form of Level 2B non-RMBS assets. In column D: The amounts extended in these transactions. In column E: The market value of the Level 2B non-RMBS collateral received in these transactions.	145–146
295	Margin lending backed by non-Level 1 or non-Level 2 collateral	Collateralised loans extended to customers for the purpose of taking leveraged trading positions ("margin loans") made against non-HQLA collateral. In column D: The amounts extended in these transactions. In column E: The market value of the collateral received in these transactions.	145–146
296	Transactions backed by other collateral	All such transactions (other than those reported in line 295) in which the bank has obtained collateral in another form than Level 1 or Level 2 assets. In column D: The amounts extended in these transactions. In column E: The market value of collateral received in these transactions.	145–146

Row	Heading	Description	Basel III LCR
			standards
			reference

b) Other inflows by counterparty

Contractual inflows (including interest payments and instalments) due in \leq 30 days from fully performing loans, not reported in lines 276 to 296. These include maturing loans that have already been agreed to roll over. The agreed roll-over should also be reported in lines 242 to 246 as appropriate.

Inflows should only be taken at the latest possible date, based on the contractual rights available to counterparties. For revolving credit facilities, this assumes that the existing loan is rolled over and that any remaining balances are treated in the same way as a committed facility according to Basel III LCR standards, paragraph 131.

Inflows from loans that have no specific maturity (ie have non-defined or open maturity) should not be included; therefore, no assumptions should be applied as to when maturity of such loans would occur. An exception to this, as noted below, would be minimum payments of principal, fee or interest associated with an open maturity loan, provided that such payments are contractually due within 30 days.

302	Retail customers	All payments (including interest payments and instalments) from retail customers on fully performing loans not reported in	153
		lines 276 to 296 that are contractually due within the 30-day horizon. Only contractual payments due should be reported, eg required minimum payments of principal, fee or interest, and not total loan balances of undefined or open maturity.	
303	Small business customers	All payments (including interest payments and instalments) from small business customers on fully performing loans not reported in lines 276 to 296 that are contractually due within the 30-day horizon. Only contractual payments due should be reported, eg required minimum payments of principal, fee or interest, and not total loan balances of undefined or open maturity.	153
304	Non-financial corporates	All payments (including interest payments and instalments) from non-financial corporates on fully performing loans not reported in lines 276 to 296 that are contractually due within the 30-day horizon. Only contractual payments due should be reported, eg required minimum payments of principal, fee or interest, and not total loan balances of undefined or open maturity.	154
305	Central banks	All payments (including interest payments and instalments) from central banks on fully performing loans. Central bank reserves (including required reserves) including banks' overnight deposits with the central bank, and term deposits with the central bank that: (i) are explicitly and contractually repayable on notice from the depositing bank; or (ii) that constitute a loan against which the bank can borrow on a term basis or on an overnight but automatically renewable basis (only where the bank has an existing deposit with the relevant central bank), should be reported in lines 7 or 8 and not here. If the term of other deposits (not included in lines 7 or 8) expires within 30 days, it should be included in this line.	154
306	Financial institutions, of which	All payments (including interest payments and instalments) from financial institutions on fully performing loans not reported in lines 276 to 296 that are contractually due within the 30-day horizon. Only contractual payments due should be reported, eg required minimum payments of principal, fee or interest, and not total loan balances of undefined or open maturity.	154
307	operational deposits	All deposits held at other financial institutions for operational activities, as outlined in the Basel III LCR standards, paragraphs 93 to 104, such as for clearing, custody, and cash management activities.	156

Row	Heading	Description	Basel III LCR standards reference
308	deposits at the centralised institution of an institutional network that receive 25% run-off	For banks that belong to a cooperative network as described in paragraphs 105 and 106 of the Basel III LCR standards, this item includes all (portions of) deposits (not included in line item 307) held at the centralised institution in the cooperative banking network that are placed (a) due to statutory minimum deposit requirements which are registered at regulators, or (b) in the context of common task sharing and legal, statutory or contractual arrangements. These deposits receive a 25% runoff at the centralised institution.	157
309	all payments on other loans and deposits due in ≤ 30 days	All payments (including interest payments and instalments) from financial institutions on fully performing unsecured and secured loans, that are contractually due within the 30-day horizon, and the amount of deposits held at financial institutions that is or becomes available within 30 days, and that are not included in lines 307 or 308. Banks may also recognise in this category inflows from the release of balances held in segregated accounts in accordance with regulatory requirements for the protection of customer trading assets, provided that these segregated balances are maintained in Level 1 or Level 2 assets. This inflow should be calculated in line with the treatment of other related outflows and inflows covered in this standard.	154
310	Other entities	All payments (including interest payments and instalments) from other entities (including sovereigns, multilateral development banks, and PSEs) on fully performing loans that are contractually due within 30 days, not included in lines 302 to 309.	154

	ther cash inflows rivatives cash inflow	Banks should calculate, in accordance with their existing valuation methodologies, expected contractual derivative cash inflows and outflows. Cash flows may be calculated on a net basis (ie inflows can offset outflows) by counterparty, only where a valid master netting agreement exists. The sum of all net cash inflows should be reported here. The sum of all net cash outflows should be reported in line 214. Banks should exclude from such calculations those liquidity requirements that would result from increased collateral needs due to market value movements (to be reported in line 222) or	158, 159
316 Der	rivatives cash inflow	valuation methodologies, expected contractual derivative cash inflows and outflows. Cash flows may be calculated on a net basis (ie inflows can offset outflows) by counterparty, only where a valid master netting agreement exists. The sum of all net cash inflows should be reported here. The sum of all net cash outflows should be reported in line 214. Banks should exclude from such calculations those liquidity requirements that would result from increased collateral needs due to market value movements (to be reported in line 222) or	158, 159
		falls in value of collateral posted (reported in line 217 and line 218). Options should be assumed to be exercised when they are 'in the money' to the option buyer. Where derivatives are collateralised by HQLA, cash inflows should be calculated net of any corresponding cash or contractual collateral outflows that would result, all other things being equal, from contractual obligations for cash or collateral to be posted by the bank, given these contractual obligations would reduce the stock of HQLA. This is in line with the principle that banks should not double count liquidity inflows and outflows. Note that cash flows do not equal the marked-to-market value, since the marked-to-market value also includes estimates for contingent inflows and outflows and may include cash flows that occur beyond the 30-day horizon. It is generally expected that a positive amount would be provided for both this line item and line 214 for institutions engaged in derivatives transactions.	
sec day	ntractual inflows from curities maturing ≤ 30 ys and not included ywhere above	Contractual inflows from securities, including certificates of deposit, maturing ≤ 30 days that are not already included in any other item of the LCR framework, provided that they are fully performing (ie no default expected). Level 1 and Level 2 securities maturing within 30 days should be included in the stock of liquid assets in panel A, provided that they meet all operational and definitional requirements outlined in the Basel III LCR standards.	155
infl	her contractual cash lows	Any other contractual cash inflows to be received ≤ 30 days that are not already included in any other item of the LCR framework. Inflow percentages should be determined as appropriate for each type of inflow by supervisors in each jurisdiction. Cash inflows related to non-financial revenues are not to be included, since they are not taken into account in the calculation of LCR. Any non-contractual contingent inflows should not be reported, as they are not included in the LCR. Please provide your supervisor with an explanatory note on any amounts included in this line.	160
ensure a min	prevent banks from relying so nimum level of HQLA holding	olely on anticipated inflows to meet their liquidity requirement, and gs, the amount of inflows that can offset outflows is capped at 75% the standard. This requires that a bank must maintain a minimum a	6 of total
		he standard. This requires that a bank must maintain a minimum a net cash outflows (Basel III LCR standards, paragraph 144).	mount of
324 Cap	p on cash inflows	The cap on cash inflows is equal to 75% of total cash outflows.	69, 144

Row	Heading	Description	Basel III LCR standards reference
325	Total cash inflows after applying the cap	The amount of total cash inflows after applying the cap is the lower of the total cash inflows before applying the cap and the level of the cap.	69, 144

6.1.4 Collateral swaps (panel C)

Any transaction maturing within 30 days in which non-cash assets are swapped for other non-cash assets, should be reported in this panel. "Level 1 assets" in this section refers to Level 1 assets other than cash. Please refer to the instructions from your supervisor for the specification of items related to Level 2B assets in this subsection.

Row	Heading	Description	Basel III LCR standards reference
330	Collateral swaps maturing ≤ 30 days	Any transaction maturing within 30 days in which non-cash assets are swapped for other non-cash assets.	48, 113, 146, Annex 1
331	Of which the borrowed assets are not re-used (ie are not rehypothecated) to cover short positions	Such transactions in which the collateral obtained is not reused (ie is not rehypothecated) in transactions to cover short positions. If the collateral is re-used, the transaction should be reported in lines 406 to 430.	48, 113, 146, Annex 1
332	Level 1 assets are lent and Level 1 assets are borrowed; of which:	Such transactions in which the bank has swapped Level 1 assets (lent) for other Level 1 assets (borrowed).	48, 113, 146, Annex 1
333	Involving eligible liquid assets	Of the transactions where Level 1 assets are lent and Level 1 assets are borrowed, those where: (i) the Level 1 collateral borrowed is reported in panel Aa of the "LCR" worksheet (which should also be reported in E333), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 1 collateral lent would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (which is the value that should be reported in D333), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
335	Level 1 assets are lent and Level 2A assets are borrowed; of which:	Such transactions in which the bank has swapped Level 1 assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
336	Involving eligible liquid assets	Of the transactions where Level 1 assets are lent and Level 2A assets are borrowed, those where: (i) the Level 2A collateral borrowed is reported in panel Ab of the "LCR" worksheet (which should also be reported in E336), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 1 collateral lent would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (which is the value that should be reported in D336), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
338	Level 1 assets are lent and Level 2B RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 1 assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
339	Involving eligible liquid assets	Of the transactions where Level 1 assets are lent and Level 2B RMBS assets are borrowed, those where: (i) the Level 2B RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E339), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 1 collateral lent would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (which is the value that should be reported in D339), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
341	Level 1 assets are lent and Level 2B non-RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 1 assets (lent) for Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
342	Involving eligible liquid assets	Of the transactions where Level 1 assets are lent and Level 2B non-RMBS assets are borrowed, those where: (i) the Level 2B non-RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E342), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 1 collateral lent would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (which is the value that should be reported in D342), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
344	Level 1 assets are lent and other assets are borrowed; of which:	Such transactions in which the bank has swapped Level 1 assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
345	Involving eligible liquid assets	Of the transactions where Level 1 assets are lent and other assets are borrowed, those where: (a) the Level 1 collateral lent would otherwise qualify to be reported in panel Aa of the "LCR" worksheet (value to be reported in D345), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards); and (b) the collateral borrowed is non-Level 1 and non-Level 2 assets (which is the value that should be reported in E345).	48, 113, 146, Annex 1
347	Level 2A assets are lent and Level 1 assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2A assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
348	Involving eligible liquid assets	Of the transactions where Level 2A assets are lent and Level 1 assets are borrowed, those where: (i) the Level 1 collateral borrowed is reported in panel Aa of the "LCR" worksheet (which should also be reported in E348), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2A collateral lent would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (which is the value that should be reported in D348), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
350	Level 2A assets are lent and Level 2A assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2A assets (lent) for other Level 2A assets (borrowed).	48, 113, 146, Annex 1
351	Involving eligible liquid assets	Of the transactions where Level 2A assets are lent and Level 2A assets are borrowed, those where: (i) the Level 2A collateral borrowed is reported in panel Ab of the "LCR" worksheet (which should also be reported in E351), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2A collateral lent would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (which is the value that should be reported in D351), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
353	Level 2A assets are lent and Level 2B RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2A assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
354	Involving eligible liquid assets	Of the transactions where Level 2A assets are lent and Level 2B RMBS assets are borrowed, those where: (i) the Level 2B RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E354), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2A collateral lent would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (which is the value that should be reported in D354), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
356	Level 2A assets are lent and Level 2B non-RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2A assets (lent) for other Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
357	Involving eligible liquid assets	Of the transactions where Level 2A assets are lent and Level 2B non-RMBS assets are borrowed, those where: (i) the Level 2B non-RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E357), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2A collateral lent would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (which is the value that should be reported in D357), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
359	Level 2A assets are lent and other assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2A assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1
360	Involving eligible liquid assets	Of the transactions where Level 2A assets are lent and other assets are borrowed, those where: (i) the Level 2A collateral lent would otherwise qualify to be reported in panel Ab of the "LCR" worksheet (which is the value that should be reported in D360), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards); and (ii) the collateral borrowed is non-Level 1 and non-Level 2 assets (which is the value that should be reported in E360).	48, 113, 146, Annex 1
362	Level 2B RMBS assets are lent and Level 1 assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
363	Involving eligible liquid assets	Of the transactions where Level 2B RMBS assets are lent and Level 1 assets are borrowed, those where: (i) the Level 1 collateral borrowed is reported in panel Aa of the "LCR" worksheet (which should also be reported in E363), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D363), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
365	Level 2B RMBS assets are lent and Level 2A assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1
366	Involving eligible liquid assets	Of the transactions where Level 2B RMBS assets are lent and Level 2A assets are borrowed, those where: (i) the Level 2A collateral borrowed is reported in panel Ab of the "LCR" worksheet (which should also be reported in E366), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D366), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
368	Level 2B RMBS assets are lent and Level 2B RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
369	Involving eligible liquid assets	Of the transactions where Level 2B RMBS assets are lent and Level 2B RMBS assets are borrowed, those where: (i) the Level 2B RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E369), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D369), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
371	Level 2B RMBS assets are lent and Level 2B non-RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for other Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
372	Involving eligible liquid assets	Of the transactions where Level 2B RMBS assets are lent and Level 2B non-RMBS assets are borrowed, those where: (i) the Level 2B non-RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E372), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D372), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
374	Level 2B RMBS assets are lent and other assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1
375	Involving eligible liquid assets	Of the transactions where Level 2B RMBS assets are lent and other assets are borrowed, those where: (i) the Level 2B RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D375), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards); and (ii) the collateral borrowed is non-Level 1 and non-Level 2 assets (which is the value that should be reported in E375).	48, 113, 146, Annex 1
377	Level 2B non-RMBS assets are lent and Level 1 assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
378	Involving eligible liquid assets	Of the transactions where Level 2B non-RMBS assets are lent and Level 1 assets are borrowed, those where: (i) the Level 1 collateral borrowed is reported in panel Aa of the "LCR" worksheet (which should also be reported in E378), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B non-RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D378), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
380	Level 2B non-RMBS assets are lent and Level 2A assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
381	Involving eligible liquid assets	Of the transactions where Level 2B non-RMBS assets are lent and Level 2A assets are borrowed, those where: (i) the Level 2A collateral borrowed is reported in panel Ab of the "LCR" worksheet (which should also be reported in E381), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B non-RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D381), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
383	Level 2B non-RMBS assets are lent and Level 2B RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
384	Involving eligible liquid assets	Of the transactions where Level 2B non-RMBS assets are lent and RMBS assets are borrowed, those where: (i) the RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E384), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B non-RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D384), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
386	Level 2B non-RMBS assets are lent and Level 2B non- RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for other Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
387	Involving eligible liquid assets	Of the transactions where Level 2B non-RMBS assets are lent and Level 2B non-RMBS assets are borrowed, those where: (i) the Level 2B non-RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E387), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the Level 2B non-RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D387), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards).	48, 113, 146, Annex 1
389	Level 2B non-RMBS assets are lent and other assets are borrowed; of which:	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
390	Involving eligible liquid assets	Of the transactions where Level 2B non-RMBS assets are lent and other assets are borrowed, those where: (i) the Level 2B non-RMBS collateral lent would otherwise qualify to be reported in panel Ac of the "LCR" worksheet (which is the value that should be reported in D390), if they were not already securing the particular transaction in question (ie would be unencumbered and would meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards); and (ii) the collateral borrowed is non-Level 1 and non-Level 2 assets (which is the value that should be reported in E390).	48, 113, 146, Annex 1
392	Other assets are lent and Level 1 assets are borrowed; of which:	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
393	Involving eligible liquid assets	Of the transactions where other assets are lent and Level 1 assets are borrowed, those where: (i) the Level 1 collateral borrowed is reported in panel Aa of the "LCR" worksheet (which should also be reported in E393), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the collateral lent is non-Level 1 and non-Level 2 assets (which is the value that should be reported in D393).	48, 113, 146, Annex 1
395	Other assets are lent and Level 2A assets are borrowed; of which:	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1
396	Involving eligible liquid assets	Of the transactions where other assets are lent and Level 2A assets are borrowed, those where: (i) the Level 2A collateral borrowed is reported in panel Ab of the "LCR" worksheet (which should also be reported in E396), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the collateral lent is non-Level 1 and non-Level 2 assets (which is the value that should be reported in D396).	48, 113, 146, Annex 1
398	Other assets are lent and Level 2B RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
399	Involving eligible liquid assets	Of the transactions where other assets are lent and Level 2B RMBS assets are borrowed, those where: (i) the Level 2B RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E399), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the collateral lent is non-Level 1 and non-Level 2 assets (which is the value that should be reported in D399).	48, 113, 146, Annex 1
401	Other assets are lent and Level 2B non-RMBS assets are borrowed; of which:	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	
402	Involving eligible liquid assets	Of the transactions where other assets are lent and Level 2B non-RMBS assets are borrowed, those where: (i) the Level 2B non-RMBS collateral borrowed is reported in panel Ac of the "LCR" worksheet (which should also be reported in E402), as the assets meet the operational requirements for HQLA as specified in paragraphs 28 to 40 of the Basel III LCR standards; and (ii) the collateral lent is non-Level 1 and non-Level 2 assets (which is the value that should be reported in D402).	48, 113, 146, Annex 1
404	Other assets are lent and other assets are borrowed	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1
405	Of which the borrowed assets are re-used (ie are rehypothecated) in transactions to cover short positions	If the collateral obtained in these transactions is re-used (ie rehypothecated) to cover short positions that could be extended beyond 30 days, it should be assumed that the transactions will be rolled-over and will not give rise to any cash inflows. This reflects the need to continue to cover the short position or to repurchase the relevant securities. If the collateral is not re-used, the transaction should be reported in lines 332 to 404.	48, 113, 146, Annex 1
406	Level 1 assets are lent and Level 1 assets are borrowed	Such transactions in which the bank has swapped Level 1 assets (lent) for other Level 1 assets (borrowed).	48, 113, 146, Annex 1
407	Level 1 assets are lent and Level 2A assets are borrowed	Such transactions in which the bank has swapped Level 1 assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1
408	Level 1 assets are lent and Level 2B RMBS assets are borrowed	Such transactions in which the bank has swapped Level 1 assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
409	Level 1 assets are lent and Level 2B non-RMBS assets are borrowed	Such transactions in which the bank has swapped Level 1 assets (lent) for Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
410	Level 1 assets are lent and other assets are borrowed	Such transactions in which the bank has swapped Level 1 assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1
411	Level 2A assets are lent and Level 1 assets are borrowed	Such transactions in which the bank has swapped Level 2A assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
412	Level 2A assets are lent and Level 2A assets are borrowed	Such transactions in which the bank has swapped Level 2A assets (lent) for other Level 2A assets (borrowed).	48, 113, 146, Annex 1
413	Level 2A assets are lent and Level 2B RMBS assets are borrowed	Such transactions in which the bank has swapped Level 2A assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
414	Level 2A assets are lent and Level 2B non-RMBS assets are borrowed	Such transactions in which the bank has swapped Level 2A assets (lent) for Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
415	Level 2A assets are lent and other assets are borrowed	Such transactions in which the bank has swapped Level 2A assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1

Row	Heading	Description	Basel III LCR standards reference
416	Level 2B RMBS assets are lent and Level 1 assets are borrowed	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
417	Level 2B RMBS assets are lent and Level 2A assets are borrowed	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1
418	Level 2B RMBS assets are lent and Level 2B RMBS assets are borrowed	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for other Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
419	Level 2B RMBS assets are lent and Level 2B non-RMBS assets are borrowed	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
420	Level 2B RMBS assets are lent and other assets are borrowed	Such transactions in which the bank has swapped Level 2B RMBS assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1
421	Level 2B non-RMBS assets are lent and Level 1 assets are borrowed	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
422	Level 2B non-RMBS assets are lent and Level 2A assets are borrowed	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1
423	Level 2B non-RMBS assets are lent and Level 2B RMBS assets are borrowed	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
424	Level 2B non-RMBS assets are lent and Level 2B non- RMBS assets are borrowed	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for other Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
425	Level 2B non-RMBS assets are lent and other assets are borrowed	Such transactions in which the bank has swapped Level 2B non-RMBS assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1
426	Other assets are lent and Level 1 assets are borrowed	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 1 assets (borrowed).	48, 113, 146, Annex 1
427	Other assets are lent and Level 2A assets are borrowed	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 2A assets (borrowed).	48, 113, 146, Annex 1
428	Other assets are lent and Level 2B RMBS assets are borrowed	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 2B RMBS assets (borrowed).	48, 113, 146, Annex 1
429	Other assets are lent and Level 2B non-RMBS assets are borrowed	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for Level 2B non-RMBS assets (borrowed).	48, 113, 146, Annex 1
430	Other assets are lent and other assets are borrowed	Such transactions in which the bank has swapped other assets than Level 1 or Level 2 assets (lent) for other assets than Level 1 or Level 2 assets (borrowed).	48, 113, 146, Annex 1

6.2 Net Stable Funding Ratio (NSFR)

The Net Stable Funding Ratio has been developed to ensure a stable funding profile in relation to the characteristics of the composition of an institution's assets and off-balance sheet activities. A sustainable funding structure is intended to reduce the likelihood that disruptions to a bank's regular sources of funding will erode its liquidity position in a way that would increase the risk of its failure and potentially lead to broader systemic stress. This metric establishes a minimum level of stable funding based on the liquidity characteristics of an institution's on- and off-balance sheet items over a one year horizon.

The NSFR is defined as the ratio of the amount of available stable funding to the amount of required stable funding. *Available* stable funding is defined as the portion of capital and liabilities expected to be reliable over the time horizon considered by the NSFR, which extends to one year. The amount of such funding *required* of a specific institution is a function of the liquidity characteristics and residual maturities of the various assets held by that institution as well as those of its off-balance sheet exposures.

Banks should report their NSFR using the same scope of application as for the Liquidity Coverage Ratio. All references to LCR definitions in the NSFR refer to the definitions in the LCR standard published by the Basel Committee. Supervisors who have chosen to implement a more stringent definition in their domestic LCR rules than those set out in the Basel Committee LCR standard have discretion over whether to apply this stricter definition for the purposes of implementing the NSFR requirements in their jurisdiction.

The template asks banks to allocate their liabilities and capital as reported on their balance sheet to the specific Available Stable Funding (ASF) categories outlined below. Banks should allocate the assets reported on their balance sheet to specific Required Stable Funding (RSF) categories according to:

- (i) their remaining maturity;
- (ii) whether they are unencumbered or encumbered; and,
- (iii) if they are encumbered, the duration of the encumbrance.

Treatment of securities financing transactions

Use of balance sheet and accounting treatments should generally result in banks **excluding**, from their assets, securities which they have borrowed in securities financing transactions (such as reverse repos and collateral swaps) where they do not have beneficial ownership. In contrast, banks should **include** securities they have lent in securities financing transactions (such as repos or collateral swaps) where they retain beneficial ownership.

Banks should also exclude any securities they have received through collateral swaps if these securities do not appear on their balance sheets.

Where banks have encumbered securities in repos or other securities financing transactions, but have retained beneficial ownership and those assets remain on the bank's balance sheet, the bank should allocate such securities to the appropriate RSF category.

Securities financing transactions with a single counterparty may be measured net when calculating the NSFR, provided that the netting conditions set out in Paragraph 33(i) of the Basel III Leverage ratio framework are met.

Treatment of encumbrance

In accordance with the principle that a bank cannot derive liquidity benefit from assets that they have encumbered, banks are required to identify whether specific assets have been encumbered and for what duration. For each category of assets, banks should report in separate lines the balances of encumbered and unencumbered assets in the appropriate column, depending on the residual maturity of the asset.

Assets encumbered for exceptional central bank liquidity operations⁴⁴ where national supervisors and central banks have agreed to a reduced RSF factor (not lower than the RSF factor applied to the equivalent asset that is unencumbered) should report such values separately as described below.

Further details of how encumbrance is to be reported are included at the start of section 6.2.2.

Treatment of derivatives payables and derivatives receivables

A bank will usually have both derivatives liabilities (ie payables) and derivative assets (ie receivables) on its balance sheet. Derivative liabilities are calculated first based on the replacement cost for derivative contracts (obtained by marking to market) where the contract has a negative value. When an eligible bilateral netting contract is in place that meets the conditions as specified in paragraphs 8 and 9 of the annex of the Basel III leverage ratio framework, the replacement cost for the set of derivative exposures covered by the contract will be the net replacement cost. In calculating NSFR derivative liabilities, collateral posted in the form of variation margin in connection with derivatives contracts, regardless of the asset type, must be deducted from the negative replacement cost amount. 45,46

Derivative assets are calculated first based on the replacement cost for derivative contracts (obtained by marking to market) where the contract has a positive value. When an eligible bilateral netting contract is in place that meets the conditions as specified in paragraphs 8 and 9 of the annex of the Basel III leverage ratio framework, the replacement cost for the set of derivative exposures covered by the contract will be the net replacement cost.

In calculating NSFR derivatives assets, collateral received in connection with derivatives contracts may not offset the positive replacement cost amount, regardless of whether or not netting is permitted under the bank's operative accounting or risk-based framework, unless it is received in the form of cash variation margin and meets the conditions as specified in paragraph 25 of the Basel III Leverage ratio framework or further specified in any related FAQ. Any remaining balance sheet liability associated with (a) variation margin received that does not meet the criteria above or (b) initial margin received may not offset derivative assets and should be assigned a 0% ASF factor.

6.2.1 Available stable funding (panel A)

The available amount of stable funding is calculated by first assigning the **carrying value** of an institution's capital and liabilities to the categories below, which are also listed in Table 1, page 6 of the Basel III NSFR standards. ⁴⁸ Carrying value represents the amount at which a liability or equity instrument is recorded before the application of any regulatory deductions, filters or other adjustments and is the amount prior to the application of any ASF factors.

Some amendments have been made to the definitions in the Basel III NSFR standards to take into account the collection of data in maturity buckets.

- Institutions should report all capital and liabilities to the appropriate columns based on maturity.
- In general, exceptional central bank liquidity operations are considered to be non-standard, temporary operations conducted by the central bank in order to achieve its mandate in a period of market-wide financial stress and/or exceptional macroeconomic challenges.
- NSFR derivative liabilities = (derivative liabilities) (total collateral posted as variation margin on derivative liabilities)
- To the extent the bank's accounting framework reflects on balance sheet, in connection with a derivatives contract, an asset associated with collateral posted as variation margin that is deducted from the replacement cost amount for purposes of the NSFR, that asset should not be included in the calculation of a bank's RSF to avoid any double counting.
- NSFR derivative assets = (derivative assets) (cash collateral received as variation margin on derivative assets)
- www.bis.org/bcbs/publ/d295.htm.

- When determining the maturity of an instrument, investors are assumed to redeem a call option at the earliest possible date. For funding with options exercisable at the bank's discretion supervisors should take into account reputational factors that may limit a bank's ability not to exercise the option. ⁴⁹ In particular, where the market expects certain liabilities to be redeemed before their legal final maturity date, banks and supervisors should assume such behaviour for the purpose of the NSFR and include these liabilities in the corresponding ASF category. For long-dated liabilities, only the portion of cash flows falling at or beyond the sixmonth and one-year time horizons should be treated as having an effective residual maturity of six months or more and one year or more, respectively.
- For retail and small business customers the same methodology for determining maturity should be followed in the NSFR as in the LCR.
- Deposits with a fixed term should be allocated to the appropriate maturity bucket; non-maturity (demand) deposits should be reported in the column for less than six months.

Row	Heading	Description	Basel III NSFR standards reference (unless otherwise noted)
6	Tier 1 and 2 capital (Basel III 2022), before the application of capital deductions and excluding the proportion of Tier 2 instruments with residual maturity of less than one year	The total amount of regulatory capital, before the application of capital deductions, as defined in paragraph 49 of the Basel III capital standards, excluding the proportion of Tier 2 instruments with residual maturity of less than one year. Amounts reported here should only include amounts after transitional arrangements have expired under fully implemented Basel III standards (ie as in 2022). Standards governing Tier 1 and Tier 2 capital are described in the Basel III capital standards.	21(a)
8	Capital instruments not included above with an effective residual maturity of one year or more	The total amount of any capital instrument not included in line 6 that has an effective residual maturity of one year or more but excluding any instruments with explicit or embedded options that, if exercised, would reduce the expected maturity to less than one year.	21(b)
9	"Stable" (as defined in the LCR) demand and/or term deposits from retail and small business customers	"Stable" non-maturity (demand) deposits and/or term deposits (as defined in the LCR in paragraphs 75 to 78) provided by retail customers and small business customers. Term deposits, regardless of the residual contractual maturity, which may be withdrawn early without entailing a withdrawal penalty significantly greater than the loss of interest should be reported in the <6 months column.	21(c), 22
11	"Less stable" (as defined in the LCR) demand and/or term deposits from retail and small business customers	"Less stable" (as defined in the LCR in paragraphs 79 to 81) non-maturity (demand) deposits and/or term deposits provided by retail and small business customers. Term deposits, regardless of the residual contractual maturity, which may be withdrawn early without entailing a withdrawal penalty significantly greater than the loss of interest should be reported in the <6 months column.	21(c), 23

This could reflect a case where a bank may imply that it would be subject to funding risk if it did not exercise an option on its own funding.

Row	Heading	Description	Basel III NSFR standards reference (unless otherwise noted)
13	Unsecured funding from non- financial corporates	Unsecured funding, non-maturity deposits and/or term deposits provided by non-financial corporates (excluding small business customers).	21(c), 24(a)
14	Of which is an operational deposit (as defined in the LCR)	Banks should report the portion of unsecured deposits provided by non-financial corporates with operational relationships, as defined in the LCR.	93–104 (Basel III LCR standards)
15	Of which is a non-operational deposit (as defined in the LCR)	Banks should report the portion of unsecured deposits provided by non-financial corporates without operational relationships, as defined in the LCR.	107–108 (Basel III LCR standards)
16	Of which is non-deposit unsecured funding	Banks should report any non-deposit unsecured funding provided by non-financial corporates.	
20	Unsecured funding from central banks	Unsecured funding, non-maturity deposits and/or term deposits provided by central banks.	21(c), 24(b), 24(d), 25(a)
21	Of which is an operational deposit (as defined in the LCR)	Banks should report the portion of unsecured deposits provided by central banks with operational relationships, as defined in the LCR.	93–104 (Basel III LCR standards)
22	Of which is a non-operational deposit (as defined in the LCR)	Banks should report the portion of unsecured deposits provided by central banks without operational relationships, as defined in the LCR.	107–108 (Basel III LCR standards)
23	Of which is non-deposit unsecured funding	Banks should report any non-deposit unsecured funding provided by central banks.	
25	Unsecured funding from sovereigns, PSEs, MDBs and NDBs	Unsecured funding, non-maturity deposits and/or term deposits provided by sovereigns, public sector entities (PSEs), multilateral development banks (MDBs) and national development banks (NDBs). Banks should include in this line unsecured funding received from the Bank for International Settlements, the International Monetary Fund and the European Commission.	21(c), 24(c)
26	Of which is an operational deposit (as defined in the LCR)	Banks should report the portion of unsecured deposits provided by sovereigns, PSEs, MDBs and NDBs with operational relationships, as defined in the LCR.	93–104 (Basel III LCR standards)
27	Of which is a non-operational deposit (as defined in the LCR)	Banks should report the portion of unsecured deposits provided by sovereigns, PSEs, MDBs and NDBs without operational relationships, as defined in the LCR.	107-108 (Basel III LCR standards)
28	Of which is non-deposit unsecured funding	Banks should report any non-deposit unsecured funding provided by sovereigns, PSEs, MDBs and NDBs.	
32	Unsecured funding from other legal entities (including financial corporates and financial institutions)	luding (including term deposits) not reported in rows 13 to 28,	
33	Of which is an operational deposit (as defined in the LCR)	Banks should report the total amount of unsecured deposits provided by other legal entities with operational relationships, as defined in the LCR.	93–104 (Basel III LCR standards)
34	Of which is a non-operational deposit (as defined in the LCR)	Banks should report the total amount of unsecured deposits provided by other legal entities without operational relationships, as defined in the LCR.	109 (Basel III LCR standards)

Row	Heading	Description	Basel III NSFR standards reference (unless otherwise noted)
35	Of which is non-deposit unsecured funding	Banks should report any non-deposit unsecured funding provided by other legal entities (including financial corporates and financial institutions). Banks should report here any non-deposit unsecured funding for which a counterparty cannot be determined (and is thus not reported in lines 16, 23, and/or 28).	
39	Deposits from members of the same cooperative network of banks subject to national discretion as defined in FN 10	In accordance with footnote 10 of the Basel III NSFR standards, this section should only be used to report deposits that exist between banks within the same cooperative network, provided they are either (a) required by law in some jurisdictions to be placed at the central organisation and are legally constrained within the cooperative bank network as minimum deposit requirements, or (b) in the context of common task sharing and legal, statutory or contractual arrangements, so long as the bank that has received the monies and the bank that has deposited participate in the same institutional network's mutual protection scheme against illiquidity and insolvency of its members. If deposits are placed in the context of (a) above, then banks should allocate the total amount of deposits received from members of their institutional network of cooperative banks according to underlying funding source in Panel D (lines 277 to 294 below), and the total balance reported in those lines should equal the balance reported here. If deposits are placed in the context of (b) above, then banks should report balances in this row but do not need to report balances in Panel D (lines 277 to 294 below). Any deposits that are operational deposits according to paragraphs 93 to 104 of the Basel III LCR standards or other deposits from members of their institutional networks of cooperative networks would be reported in line 41.	105(a) (Basel III LCR standards), footnote 10, 21(c)
41	Other deposits from members of a cooperative network of banks	Banks should report any deposits from banks that are members of the same cooperative network of banks that are operational deposits according to paragraphs 93 to 104 of the Basel III LCR standards or other deposits from members of their cooperative networks that are not included in line 39.	
42	Secured borrowings and liabilities (including secured term deposits): of which are from:	The total amount of secured borrowings and liabilities (including term deposits). Secured funding is defined as those liabilities and general obligations that are collateralised by legal rights to specifically designated assets owned by the borrowing institution in the case of bankruptcy, insolvency, liquidation or resolution.	21(c), 24, 25(a)
43	Retail and small business customers	The amount of secured borrowings and liabilities (including term deposits) from retail and small business customers.	
44	Non-financial corporates	The amount of secured borrowings and liabilities (including term deposits) from non-financial corporates.	

Row	Heading	Description	Basel III NSFR standards reference (unless otherwise noted)
45	Central banks	The amount of secured borrowings and liabilities (including term deposits) from central banks.	
46	Sovereigns/PSEs/MDBs/NDBs	The amount of secured borrowings and liabilities (including term deposits) from sovereigns/PSEs and multilateral and national development banks.	
47	Other legal entities (including financial corporates and financial institutions)	The amount of secured borrowings and liabilities (including term deposits) from other legal entities (including financial corporates and financial institutions).	
49	Derivative Liabilities, gross of variation margin posted	Report derivative liabilities based on the replacement cost for derivative contracts (obtained by marking to market) where the contract has a negative value. When an eligible bilateral netting contract is in place that meets the conditions as specified in paragraphs 8 and 9 of the annex of the Basel III leverage ratio framework, the replacement cost for the set of derivative exposures covered by the contract will be the net replacement cost. The value reported here should be gross of variation margin posted. That is, it should represent derivative liabilities prior to the deduction of variation margin posted.	19
50	Of which are derivative liabilities where the counterparty is exempt from BCBS-IOSCO margin requirements; of which:	Report derivative liabilities (as described above), gross of variation margin posted, where the counterparty is exempt from BCBS-IOSCO margin requirements, as laid out in paragraph 2(c) of the BCBS-IOSCO margin requirements for non-centrally cleared derivatives. ⁵⁰	
51	Non-financial entities that are not systemically important	Derivative liabilities with non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	
52	Sovereigns/Central Banks/MDBs/BIS	Derivative liabilities with sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	
54	Total variation margin posted	All collateral posted in the form of variation margin in connection with derivative contracts, regardless of asset type.	
55	Of which is posted to counterparties exempted from BCBS-IOSCO margin requirements; of which:	All collateral posted in the form of variation margin in row 54 above that is posted to counterparties exempt from BCBS-IOSCO margin rules, as laid out in paragraph 2(c) of the BCBS-IOSCO Margin requirements for non-centrally cleared derivatives.	
56	Non-financial entities that are not systemically important	Variation margin posted to non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	
57	Sovereigns/Central Banks/MDBs/BIS	Variation margin posted to sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	

Basel Committee on Banking Supervision and Board of the International Organization of Securities Commissions, *Margin requirements for non-centrally cleared derivatives*, September 2013, www.bis.org/publ/bcbs261.htm.

Row	Heading	Description	Basel III NSFR standards reference (unless otherwise noted)
59	NSFR derivative liabilities (derivative liabilities less total collateral posted as variation margin on derivative liabilities)	Non-entry field. In calculating NSFR derivative liabilities, collateral posted in the form of variation margin in connection with derivatives contracts, regardless of the asset type, is deducted from the negative replacement cost amount or the negative net replacement cost where applicable. 51 52	19, 20, FN 6
60	Total initial margin received	All cash, securities or other assets received as initial margin for all derivative contracts (eg, including any independent amount received in relation to OTC contracts).	
61	Of which, initial margin received in the form of cash	Cash received as initial margin for derivative contracts	
62	Of which, initial margin received in the form of Level 1 securities	Initial margin received, in the form of Level 1 securities for derivative contracts	
63	Of which, initial margin received in the form of all other collateral	Initial margin received, in the form of collateral other than cash or Level 1 securities included above in rows 61 to 62.	
65	Total initial margin received, in the form of any collateral type, according to residual maturity of associated derivative contract(s)	All cash, securities or other assets received as initial margin for derivative contracts. Report initial margin balances in this category according to the residual maturity of the derivative contract(s) directly associated with the initial margin. In the case of pooled collateral, report the maturity of initial margin balances according to the maturity of the derivative contract with the longest term in the applicable netting set that contributes to an initial margin requirement. Contracts that are fully offsetting (ie long and short positions in identical contracts) and do not contribute to an initial margin requirement may be excluded from the determination of maturity. The sum of this category should equal total initial margin received in row 60 above.	
67	Initial margin received, in the form of any collateral type, from counterparties exempt from BCBS-IOSCO margin requirements; of which:	Cash, securities or other assets received as initial margin for derivative contracts from counterparties exempt from BCBS-IOSCO margin requirements for non-centrally cleared derivatives.	
68	Non-financial entities that are not systemically important	Cash, securities or other assets received as initial margin for derivative contracts from non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	

NSFR derivative liabilities = (derivative liabilities) – (total collateral posted as variation margin on derivative liabilities)

To the extent the bank's accounting framework reflects on balance sheet, in connection with a derivatives contract, an asset associated with collateral posted as variation margin that is deducted from the replacement cost amount for purposes of the NSFR, that asset should not be included in the calculation of a bank's RSF to avoid any double counting.

Row	Heading	Description	Basel III NSFR standards reference (unless otherwise noted)
69	Sovereigns/Central Banks/MDBs/BIS	Cash, securities or other assets received as initial margin for derivative contracts from sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	
72	Deferred tax liabilities (DTLs)	The amount of deferred tax liabilities, reported according to the nearest possible date in which such liabilities could be realised.	25(b)
73	Minority interest	The amount of minority interest, reported according to the term of the instrument, usually in perpetuity.	25(b)
74	Trade date payables	The amount of payables arising from purchases of financial instruments, foreign currencies and commodities that (i) are expected to settle within the standard settlement cycle or period that is customary for the relevant exchange or type of transaction, or (ii) have failed to, but are still expected to, settle.	25(d)
75	Interdependent liabilities	National supervisors have discretion in limited circumstances to determine interdependent assets and liabilities in accordance with paragraph 45 of Basel III NSFR standards. Report here liability items which, on the basis of contractual arrangements, are interdependent on corresponding assets report in line 249 below such that: the liability cannot fall due while the asset remains on the balance sheet, the principal payment flows from the asset cannot be used for something other than repaying the liability, and the liability cannot be used to fund other assets. For interdependent items, supervisors may adjust RSF and ASF factors so that they are both 0%, subject to the following criteria: The individual interdependent asset and liability items must be clearly identifiable. The maturity and principal amount of both the liability and its interdependent asset should be the same. The bank is acting solely as a pass-through unit to channel the funding received (the interdependent liability) into the corresponding interdependent asset. The counterparties for each pair of interdependent liabilities and assets should not be the same.	45
76	All other liability and equity categories not included above	All other liabilities of the institution (not otherwise reported in above categories) should be accounted for in this row at their carrying value. The value of short positions and open maturity positions should be reported in the < 6 month column. Note: deductions from capital should not be included in the amount reported in this line item, and should instead be reported according to the instructions in line 247 below.	21(c), 24(d), 25(a), 25(b)

6.2.2 Required stable funding (panel B)

The amount of required stable funding (RSF) is measured using assumptions on the broad characteristics of the liquidity risk profile of an institution's assets and off-balance sheet exposures. The amount of required stable funding is calculated by first assigning the **carrying value** of an institution's assets to the categories below, which are also listed in Table 2, page 11 of the Basel III NSFR standards. The amount assigned to each category is then multiplied by an RSF factor and the total RSF is the sum of the weighted amounts added to the amount of off-balance sheet activity (or potential liquidity exposure) multiplied by its associated RSF factor.

The RSF factor applied to the reported values of each asset or off-balance sheet exposure is intended to approximate the amount of a particular asset that would have to be funded, either because it will be rolled over or because it could not be monetised through sale or used as collateral in a secured borrowing transaction over the course of one year without significant expense. Under the standard, such amounts are expected to be supported by stable funding.

In completing this section of the template banks should allocate the assets recorded on their balance sheet to the appropriate RSF category. For purposes of determining its required stable funding, an institution should (i) include financial instruments, foreign currencies and commodities for which a purchase order has been executed, and (ii) exclude financial instruments, foreign currencies and commodities for which a sales order has been executed, even if such transactions have not been reflected in the balance sheet under a settlement-date accounting model, provided that (i) such transactions are not reflected as derivatives or secured financing transactions in the institution's balance sheet, and (ii) the effects of such transactions will be reflected in the institution's balance sheet when settled.

Treatment of encumbrance

Where indicated, banks should report assets according to:

- (i) whether they are encumbered or unencumbered; and,
- (ii) if they are encumbered, according to the period of encumbrance.
- (iii) In determining encumbrance where it is not tied to specific assets, eg the encumbrance is allocated against a pool of assets that includes different RSF categories, the bank should assume that the highest RSF factor assets are encumbered first.

Where a bank has rehypothecated assets in which it has both positions it owns outright and borrowed positions, a bank should assume it has encumbered the borrowed securities first, unless it has an internal process for making this allocation, or it has applied a different methodology for determining the encumbrance of positions in the LCR. For example, if for the LCR the bank assumes positions held outright are encumbered before borrowed positions in order to recognise inflows from maturing borrowed positions, then the bank must use an equivalent approach for these transactions in the NSFR. For their encumbered assets, banks should first report their value in the appropriate column **according to residual maturity** at the carrying value on the balance sheet, and not the value assigned to it for the purposes of the encumbrance transaction. If the bank is required to over-collateralise transactions, for example due to the application of haircuts, or to achieve a desired credit-rating on a funding instrument, then these excess assets should be reported as encumbered.

The bank should then report that same value **according to the remaining period of encumbrance** in the same column of the appropriate row beneath. Banks should consider whether specific assets have a remaining term of encumbrance period (or residual encumbrance period) that is longer than the maturity of the asset, eg where in practice there is a requirement to encumber additional assets at the contracted maturity date of the currently encumbered asset. For example, if debt is secured on loans of a shorter maturity and the bank will be required to pledge additional collateral to maintain appropriate collateralisation levels, as may be the case with mortgage-backed securities.

For example, if a bank had a non-financial corporate loan that had a value of 50 with a residual maturity of 10 months, 25 of which were encumbered for a remaining period of two months, and 25 of which were encumbered for a remaining period of for seven months, it would complete the template as follows:

	Amount		
	< 6 months	≥ 6 months to < 1 year	≥1 year
Loans to non-financial corporate clients with residual maturities less than one year			
Unencumbered			
Encumbered, of which:			
Remaining period of encumbrance < 6 months		25	
Remaining period of encumbrance ≥ 6 months to < 1 year		25	
Remaining period of encumbrance ≥ 1 year			

Assets encumbered for exceptional central bank liquidity operations ⁵³ where national supervisors and central banks have agreed to a reduced RSF factor (not lower than the RSF factor applied to the equivalent asset that is unencumbered) should report such values separately in panel E of the NSFR template. Values reported in panel E should not be included in Required Stable Funding section in panel B to avoid double counting.

Row	Heading	Description	Basel III NSFR
			standards
			reference

B) Required stable funding

The required amount of stable funding is calculated by first assigning the **carrying value** of an institution's assets to the categories below, which are also listed in Table 2, page 11 of the Basel III NSFR standards. The amount assigned to each category is to be multiplied by an RSF factor and the total RSF is the sum of the weighted amounts.

Of note, definitions in the NSFR mirror those in the LCR, unless otherwise specified. In addition, for purposes of calculating the NSFR, HQLA is defined as all HQLA (defined in LCR paragraphs 24 to 68) without regard to LCR operational requirements (defined in LCR paragraphs 28 to 43) and LCR caps on Level 2 and Level 2B assets that may limit the ability of some HQLA to be included as eligible HQLA in the calculation of the LCR.

Assets that are deducted from capital should be reported in the relevant asset categories below.

Treatment of maturity

- Institutions should allocate all assets to the appropriate columns based on their residual maturity or liquidity value.
- When determining the maturity of an instrument, investors are assumed to exercise any option to extend maturity.
- For assets with options exercisable at the bank's discretion, supervisors should take into account reputational factors that may limit a bank's ability not to exercise the option. ⁵⁴ In particular, where the market expects certain assets to be extended in their maturity, banks and supervisors should assume such behaviour for the purpose of the NSFR and include these assets in the corresponding RSF category.
- For amortising loans, the portion that comes due within the one-year horizon can be treated in the less than one year residual maturity categories.

B1) On-balance sheet items

In general, exceptional central bank liquidity operations are considered to be non-standard, temporary operations conducted by the central bank in order to achieve its mandate in a period of market-wide financial stress and/or exceptional macroeconomic challenges.

This could reflect a case where a bank may imply that it would be subject to funding risk if it did not exercise an option to extend the maturity of its own assets.

Row	Heading	Description	Basel III NSFR standards reference
84	Coins and banknotes	Coins and banknotes currently held and immediately available to meet obligations. Banks should not report loans to counterparties in this row.	36(a)
85	Total central bank reserves	Total amount held in central bank reserves (including required and excess reserves) including banks' overnight deposits with the central bank, and term deposits with the central bank.	36(b)
86	Of which are required central bank reserves	Total amount held in central bank reserves related to minimum deposit requirements. Supervisors may agree with the relevant central bank on the RSF factor to be assigned to required reserves, based in particular on consideration of whether or not the reserve requirement must be satisfied at all times and thus the extent to which reserve requirements in that jurisdiction exist on a longer-term horizon and therefore require associated stable funding. Please refer to the instructions from your supervisor for the specification of this item.	FN17
88	Securities held where the institution has an offsetting reverse repurchase transaction when the security on each transaction has the same unique identifier (eg ISIN number or CUSIP) and such securities are reported on the balance sheet of the reporting institution	This category is only applicable for jurisdictions whereby accounting standards would require both the reverse repo transaction and the collateral to be reported on-balance sheet. Where this is the case, banks should report in this category, any securities reported on their balance sheet that are borrowed in reverse repurchase transactions. Reverse repo transactions that appear on their balance sheets as secured cash loans and deposits placed should not be reported in this category, rather should be reported with loans to financial institutions in rows 102 to 120. Securities in default should not be reported in this category, rather these should be reported in line 211.	
89	Unencumbered	Banks should report in this row all such unencumbered securities in the appropriate column according to their residual maturity.	
90	Encumbered, of which:	Banks should report in these rows all such encumbered securities, regardless of counterparty, in the appropriate column according to their residual maturity.	
91	Remaining period of encumbrance < 6 months	For each cell containing securities that have been encumbered, banks should in addition allocate them to a	
92	Remaining period of encumbrance ≥ 6 months to < 1 year	cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
93	Remaining period of encumbrance ≥ 1 year		

Row	Heading	Description	Basel III NSFR standards reference
94	Deposits held at other banks which are members of the same cooperative network of banks and which are subject to national discretion according to FN 10	In accordance with footnote 10 of the Basel III NSFR standards, this section should only be used to report deposits that exist between banks within the same cooperative network, provided they are received in the context of common task sharing and legal, statutory or contractual arrangements, and so long as the bank that has received the monies and the bank that has deposited participate in the same institutional network's mutual protection scheme against illiquidity and insolvency of its members. Such deposits can be assigned an ASF up to the RSF factor assigned by regulation for the same deposits to the depositing bank, not to exceed 85%. Deposits reported in this category should not be reported in any other RSF category. This category does not apply to banks in jurisdictions where deposits are required by law to be placed at the central organisation and are legally constrained within the cooperative bank network as minimum deposit requirements. Accordingly, banks that complete Panel D below, as specified in the instructions for row 39 above, should not report balances here.	FN 10, 43(c)
96	Other deposits at other banks which are members of the same cooperative network of banks; of which:	In accordance with footnote 10 of the Basel III NSFR standards, this section should only be used to report other deposits that exist between banks within the same cooperative network, provided they are received in the context of common task sharing and legal, statutory or contractual arrangements, which do not satisfy the conditions set forth in line 95 above. Deposits reported in this category should not be reported in any other RSF category. This category does not apply to banks in jurisdictions where deposits are required by law to be placed at the central organisation and are legally constrained within the cooperative bank network as minimum deposit requirements. Banks that complete Panel D below, as specified in the instructions for row 39 above, should not report balances here.	
97	Unencumbered	Banks should report in this row all such unencumbered deposits in the appropriate column according to their residual maturity.	
98	Encumbered, of which:	Banks should report in these rows all such encumbered deposits, regardless of counterparty, in the appropriate column according to their residual maturity.	
99	Remaining period of encumbrance < 6 months	For each cell containing deposits that have been encumbered, banks should in addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
100	Remaining period of encumbrance ≥ 6 months to < 1 year		
101	Remaining period of encumbrance ≥ 1 year		

Row	Heading	Description	Basel III NSFR standards reference
102	Loans to financial institutions, of which:	Loans to all financial institutions. Non-performing loans should not be included in this category, rather these should be reported in line 211. Deposits held at financial institutions for operational purposes should not be reported here and should instead be reported in line 139.	31, 38, 39(b), 40(c), 43(a), 43(c)
103	Loans to financial institutions secured by Level 1 collateral and where the bank has the ability to freely rehypothecate the received collateral for the life of the loan, of which:	All loans to financial institutions where the loan is secured against Level 1 assets, as defined in LCR paragraph 50, and where the bank has the ability to freely rehypothecate the received collateral for the life of the loan. Report loans to financial institutions secured by Level 1 assets where the bank does not have the ability to freely rehypothecate the received collateral for the life of the loan in line 109 below.	31, 38, 40(c), 43(a), 43(c)
104	Unencumbered	Banks should report in this row all such unencumbered loans in the appropriate column according to their residual maturity.	
105	Encumbered, of which:	Banks should report in these rows all such encumbered securities, regardless of counterparty, in the appropriate column according to their residual maturity.	
106	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
107	Remaining period of encumbrance ≥ 6 months to < 1 year		
108	Remaining period of encumbrance ≥ 1 year		
109	All other secured loans to financial institutions, of which:	All other secured loans to financial institutions, including both loans secured against collateral other than Level 1 assets and loans secured by Level 1 assets where the bank does not have the ability to freely rehypothecate the received collateral for the life of the loan.	31, 39(b), 40(c), 43(a), 43(c)
110	Unencumbered	Banks should report in this row all such unencumbered loans in the appropriate column according to their residual maturity. This includes both unencumbered loans secured against collateral other than Level 1 assets and unencumbered loans secured by Level 1 assets where the bank does not have the ability to freely rehypothecate the received collateral for the life of the loan.	
111	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity. This includes both encumbered loans secured against collateral other than Level 1 assets and encumbered loans secured by Level 1 assets where the bank does not have the ability to freely rehypothecate the received collateral for the life of the loan.	

Row	Heading	Description	Basel III NSFR standards reference
112	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
113	Remaining period of encumbrance ≥ 6 months to < 1 year		
114	Remaining period of encumbrance ≥ 1 year		
115	Unsecured loans to financial institutions, of which:	All loans to financial institutions that are unsecured.	31, 39(b), 40(c) 43(a), 43(c)
116	Unencumbered	Banks should report in these rows all such unencumbered loans in the appropriate column according to their residual maturity.	
117	Encumbered, of which:	Banks should report in all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	
118	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
119	Remaining period of encumbrance ≥ 6 months to < 1 year		
120	Remaining period of encumbrance ≥ 1 year		
121	Securities eligible as Level 1 HQLA for the LCR, of which:	Securities that, if unencumbered, would qualify as Level 1 liquid assets according to paragraph 50 of the Basel III LCR standards. Securities that would otherwise qualify according to that	31, 37, 40(b), 43(a)
		paragraph, but are excluded for operational or other reasons, are reported in this category. Coins and banknotes, and central bank reserves should be reported in lines 84, 85 and 86 respectively and not in this category. Securities in default should not be included in this category;	
122	Unencumbered	rather these should be reported in line 211. Banks should report in this row all such unencumbered securities in the appropriate column according to their residual maturity.	
123	Encumbered, of which:	Banks should report in these rows all such encumbered securities, regardless of counterparty, in the appropriate column according to their residual maturity.	
124	Remaining period of encumbrance < 6 months	For each cell containing securities that have been encumbered, banks should in addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
125	Remaining period of encumbrance ≥ 6 months to < 1 year		
126	Remaining period of encumbrance ≥ 1 year		

Row	Heading	Description	Basel III NSFR standards reference
127	Securities eligible for Level 2A HQLA for the LCR, of which:	Securities that, if unencumbered, would qualify as Level 2A liquid assets, according to paragraph 52 of the Basel III LCR standards. Securities that would otherwise qualify according to that paragraph, but are excluded for exceeding the 40% cap, or for operational or other reasons, are reported in this category. Securities in default should not be included in this category; rather these should be reported in line 211.	31, 39(a), 40(b), 43(a)
128	Unencumbered	Banks should report in this row all such unencumbered securities in the appropriate column according to their residual maturity.	
129	Encumbered, of which:	Banks should report in these rows all such encumbered securities, regardless of counterparty, in the appropriate column according to their residual maturity.	
130	Remaining period of encumbrance < 6 months	For each cell containing securities that have been encumbered, banks should in addition allocate them to a	
131	Remaining period of encumbrance ≥ 6 months to < 1 year	cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
132	Remaining period of encumbrance ≥ 1 year	section.	
133	Securities eligible for Level 2B HQLA for the LCR, of which:	Securities that, if unencumbered, would qualify as Level 2B liquid assets, according to paragraph 54 of the Basel III LCR standards. Securities that would otherwise qualify according to that paragraph, but are excluded for exceeding the 15% or 40% caps, or for operational or other reasons, are reported in this category. Securities in default should not be included in this category; rather these should be reported in line 211.	31, 40(a), 40(b) 43(a)
134	Unencumbered	Banks should report in this row all such unencumbered securities in the appropriate column according to their residual maturity.	
135	Encumbered, of which:	Banks should report in these rows all such encumbered securities, regardless of counterparty, in the appropriate column according to their residual maturity.	
136	Remaining period of encumbrance < 6 months	For each cell containing securities that have been encumbered, banks should in addition allocate them to a	
137	Remaining period of encumbrance ≥ 6 months to < 1 year	cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
138	Remaining period of encumbrance ≥ 1 year	section.	
139	Deposits held at financial institutions for operational purposes, of which:	Deposits held at financial institutions, including banks subject to prudential supervision, for operational purposes, as defined in LCR paragraphs 93 to 104.	31, 40(d), 43(a)
140	Unencumbered	Banks should report in this row all such unencumbered deposits in the appropriate column according to their residual maturity.	

Row	Heading	Description	Basel III NSFR standards reference
141	Encumbered, of which:	Banks should report these rows all such encumbered deposits, regardless of counterparty, in the appropriate column according to their residual maturity.	
142	Remaining period of encumbrance < 6 months	For each cell containing deposits that have been encumbered, banks should in addition allocate them to a	
143	Remaining period of encumbrance ≥ 6 months to < 1 year	cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
144	Remaining period of encumbrance ≥ 1 year	section.	
145	Loans to non-financial corporate clients with a residual maturity of less than one year; of which:	Loans to non-financial corporate clients having a residual maturity of less than one year. Non-performing loans should not be included in this category, rather these should be reported in line 211. Performing loans to non-financial corporate clients with a residual maturity of less than one year and with a greater than 35% risk weight under the Basel II standardised approach for credit risk should be reported in this category and not in line 181. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due. Banks should report in this row all such unencumbered	31, 40(e), 43(a)
146	Unencumbered	loans in the appropriate column according to their residual maturity.	
147	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	
148	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the	
149	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance. Attention is drawn to the worked example at the start of this	
150	Remaining period of encumbrance ≥ 1 year	section.	

Row	Heading	Description	Basel III NSFR standards reference
151	Loans to central banks with a residual maturity of less than one year; of which:	Loans to central banks having a residual maturity of less than one year that do not qualify to meet local reserve requirements. Balances (including term placements) that qualify toward reserve requirements should be considered as "total central bank reserves" and reported in row 85, even if these balances are in excess of the required level of reserves. Non-performing loans should not be included in this category, rather these should be reported in line 211. Performing loans to central banks with a residual maturity of less than one year and a greater than 35% risk weight under the Basel II standardised approach for credit risk should be reported in this category and not in line 181. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due.	31, 36(c), 40(c), 43(a)
152	Unencumbered	Banks should report in this row all such unencumbered loans in the appropriate column according to their residual maturity.	
153	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	
154	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the	
155	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance. Attention is drawn to the worked example at the start of this	
156	Remaining period of encumbrance ≥ 1 year	section.	
157	Loans to sovereigns, PSEs, MDBs and NDBs with a residual maturity of less than one year; of which:	Loans to sovereigns, PSEs, MDBs and NDBs having a residual maturity of less than one year. Loans to the Bank for International Settlements, the International Monetary Fund and the European Commission should also be reported in this category. Non-performing loans should not be included in this category; rather these should be reported in line 211.	31, 40(e), 41, 43(a)
		Performing loans to sovereigns, PSEs, MDBs and NDBs with a residual maturity of less than one year and a greater than 35% risk weight under the Basel II standardised approach for credit risk should be reported in this category and not in line 181. Performing loans are considered to be those that are not	
		Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due.	
158	Unencumbered	Banks should report in this row all such unencumbered loans in the appropriate column according to their residual maturity.	
159	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	

Row	Heading	Description	Basel III NSFR standards reference
160	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
161	Remaining period of encumbrance ≥ 6 months to < 1 year		
162	Remaining period of encumbrance ≥ 1 year		
163	Residential mortgages of any maturity that would qualify for the 35% or lower risk weight under the Basel II standardised approach for credit risk, of which:	Residential mortgages of any maturity that would qualify for the 35% or lower risk weight under the Basel II standardised approach for credit risk. Non-performing residential mortgages should not be reported in this category; rather these should be reported in line 211. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due.	31, 40(e), 41(a) 43(a)
164	Unencumbered	Banks should report in this row all such unencumbered mortgages in the appropriate column according to their residual maturity.	
165	Encumbered, of which:	Banks should report in these rows all such encumbered mortgages, regardless of counterparty, in the appropriate column according to their residual maturity.	
166	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the	
167	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
168	Remaining period of encumbrance ≥ 1 year	section.	
169	Other loans, excluding loans to financial institutions, with a residual maturity of one year or greater, that would qualify for the 35% or lower risk weight under the Basel II standardised approach for credit risk, of which:	Include balances of all other loans, excluding loans to financial institutions, with a residual maturity of one year or more, that would qualify for the 35% or lower risk weight under the Basel II standardised approach for credit risk. Non-performing loans should not be reported in this category; rather these should be reported in line 211. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days	31, 41(b), 43(a)
170	Unencumbered	past due. Banks should report in t all such unencumbered loans in the appropriate column according to their residual maturity.	
171	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	

Row	Heading	Description	Basel III NSFR standards reference
172	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the	
173	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance. Attention is drawn to the worked example at the start of this	
174	Remaining period of encumbrance ≥ 1 year	section.	
175	Loans to retail and small business customers (excluding residential mortgages reported above) with a residual maturity of less than one year; of which:	Loans to retail (eg natural persons) and small business customers (as defined in the LCR) having a residual maturity of less than one year. Non-performing loans should not be reported in this category, rather these should be reported in line 211. Performing loans to retail and small business customers with a residual maturity of less than one year with a greater than 35% risk weight under the Basel II standardised approach for credit risk should also be reported in this category and not in line 181. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due.	31, 40(e), 43(a)
176	Unencumbered	Banks should report in this row all such unencumbered loans in the appropriate column according to their residual maturity.	
177	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	
178	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the	
179	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
180	Remaining period of encumbrance ≥ 1 year	section.	
181	Performing loans (except loans to financial institutions and loans reported in above categories) with risk weights greater than 35% under the Basel II standardised approach for credit risk; of which:	Performing loans, not captured by one of the above categories, with a greater than 35% risk weight under the Basel II standardised approach for credit risk, excluding loans to financial institutions. Non-performing loans should not be reported in this category, rather these should be reported in line 211. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due.	31, 40(e), 42(b) 43(a), FN19
182	Unencumbered	Banks should report in this row all such unencumbered loans in the appropriate column according to their residual maturity.	
183	Encumbered, of which:	Banks should report in these rows all such encumbered loans, regardless of counterparty, in the appropriate column according to their residual maturity.	

Row	Heading	Description	Basel III NSFR standards reference
184	Remaining period of encumbrance < 6 months	For each cell containing loans that have been encumbered, banks should in addition allocate them to a cell in one of the	
185	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
186	Remaining period of encumbrance ≥ 1 year	section.	
187	Non-HQLA exchange traded equities, of which:	Exchange traded equities that do not qualify as Level 2B assets. This includes exchange traded FI equities as well as exchange traded non-FI equities that do not meet all of the requirements outlined in paragraph 54(c) of the Basel III LCR standards.	31, 42(c), 43(a)
		Amounts related to non-HQLA exchange traded equities that are deducted from capital should not be reported here, rather these should be reported in the ≥ 1 year column in row 247.	
188	Unencumbered	Banks should report in this row all such unencumbered equities in the appropriate column according to their residual maturity.	
189	Encumbered, of which:	Banks should report in these rows all such encumbered equities, regardless of counterparty, in the appropriate column according to their residual maturity.	
190	Remaining period of encumbrance < 6 months	For each cell containing equities that have been encumbered, banks should in addition allocate them to a cell in one of the	
191	Remaining period of encumbrance ≥ 6 months to < 1 year	three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
192	Remaining period of encumbrance ≥ 1 year	section.	
193	Non-HQLA securities not in default, of which:	Securities that are not eligible for HQLA treatment as defined by Basel III LCR standards, other than non-HQLA exchange traded equities, which should be reported in line 187, and which are not in default.	31, 40(e), 42(c) 43(a)
		Securities in default should not be reported in this category; rather these should be reported in line 211.	
194	Unencumbered	Banks should report in this row all such unencumbered securities in the appropriate column according to their residual maturity.	
195	Encumbered, of which:	Banks should report in these rows all such encumbered securities, regardless of counterparty, in the appropriate column according to their residual maturity.	
196	Remaining period of encumbrance < 6 months	For each cell containing securities that have been encumbered, banks should in addition allocate them to a	
197	Remaining period of encumbrance ≥ 6 months to < 1 year	cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this	
198	Remaining period of encumbrance ≥ 1 year	section.	
199	Physical traded commodities including gold, of which:	Total balance of physical traded commodities including gold should be reported in the ≥ 1 year maturity column.	31, 42(d), 43(a)

Row	Heading	Description	Basel III NSFR standards reference
200	Unencumbered	Banks should report in this row all such unencumbered physical traded commodities including gold.	
201	Encumbered, of which:	Banks should report in these rows all such encumbered physical traded commodities including gold, regardless of counterparty.	
202	Remaining period of encumbrance < 6 months	For each cell containing physical traded commodities including gold that have been encumbered, banks should in	
203	Remaining period of encumbrance ≥ 6 months to < 1 year	addition allocate them to a cell in one of the three rows directly below according to the remaining period of encumbrance.	
204	Remaining period of encumbrance ≥ 1 year	Attention is drawn to the worked example at the start of this section.	
205	Other short-term unsecured instruments and transactions with a residual maturity of less than one year, of which are:	Banks should report the balances of other short-term unsecured instruments with outstanding maturities of less than one year. Such instruments include but are not limited to: short-term government and corporate bills, notes, and obligations; commercial paper; negotiable CDs; bankers' acceptances; money market mutual funds. Banks should not report in this row any central bank reserves, Level 1, Level 2A and Level 2B assets, unsecured interbank and other money market placements (eg	31, 40(e), 43(a)
		federal funds or euro currencies sold) or instruments in default. These are reported elsewhere on the template.	
206	Unencumbered	Banks should report in this row all such unencumbered instruments and transactions in the appropriate column according to their residual maturity.	
207	Encumbered, of which:	Banks should report in these rows all such encumbered instruments and transactions, regardless of counterparty, in the appropriate column according to their residual maturity.	
208	Remaining period of encumbrance < 6 months	For each cell containing instruments that have been encumbered, banks should in addition allocate them to a	
209	Remaining period of encumbrance ≥ 6 months to < 1 year	cell in one of the three rows directly below according to the remaining period of encumbrance . Attention is drawn to the worked example at the start of this section.	
210	Remaining period of encumbrance ≥ 1 year		
211	Defaulted securities and non-performing loans	All defaulted securities and non-performing loans should be reported in this line and not in one of the above categories. Performing loans are considered to be those that are not past due for more than 90 days in accordance with paragraph 75 of the Basel II framework. Conversely, non-performing loans are considered to be loans that are more than 90 days past due.	43(c), FN19

Row	Heading	Description	Basel III NSFR standards reference
213	Derivative assets, gross of variation margin received	Report derivative assets based on the replacement cost for derivative contracts (obtained by marking to market) where the contract has a positive value. When an eligible bilateral netting contract is in place that meets the conditions as specified in paragraphs 8 and 9 of the annex of the Basel III Leverage ratio framework, the replacement cost for the set of derivative exposures covered by the contract will be the net replacement cost. The value reported here should be gross of variation margin received. That is, it should represent derivative assets prior to the deduction of variation margin received.	
214	Of which are derivative assets where the counterparty is exempt from BCBS-IOSCO margin requirements; of which:	Derivative assets (as described above), gross of variation margin received, where the counterparty is exempt from BCBS-IOSCO margin requirements, as laid out in paragraph 2(c) of the BCBS-IOSCO Margin requirements for non-centrally cleared derivatives ⁵⁵ .	
215	Non-financial entities that are not systemically important	Derivative assets with non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	
216	Sovereigns/Central Banks/MDBs/BIS	Derivative assets with sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	
218	Variation margin received, of which:	Collateral received in the form of variation margin in connection with derivatives contracts.	
219	Cash variation margin received, meeting conditions as specified in paragraph 25 of the Basel III Leverage Ratio Framework and Disclosure Requirements	Collateral received in the form of cash variation margin in connection with derivatives contracts meeting the conditions as specified in paragraph 25 of the Basel III Leverage Ratio Framework and Disclosure Requirements or further specified in any related FAQ.	
221	Of which is received from counterparties exempted from BCBS-IOSCO margining requirements; of which:	Cash variation margin meeting the conditions as specified in paragraph 25 of the <i>Basel III Leverage Ratio Framework and Disclosure Requirements</i> or further specified in any related FAQ that is received from counterparties exempt from BCBS-IOSCO margin requirements, as laid out in paragraph 2(c) of the BCBS-IOSCO margining requirements for non-centrally cleared derivatives.	
222	Non-financial entities that are not systemically important	Cash variation margin received from non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	
223	Sovereigns/Central Banks/MDBs/BIS	Cash variation margin received from sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	
225	Other variation margin received	All other collateral received in the form of variation margin in connection with derivatives contracts that is not reported in line 219 above.	

Basel Committee on Banking Supervision and Board of the International Organization of Securities Commissions, *Margin requirements for non-centrally cleared derivatives*, September 2013, www.bis.org/publ/bcbs261.htm.

Row	Heading	Description	Basel III NSFR standards reference
226	Of which is received from counterparties exempted from BCBS-IOSCO margining requirements	Other collateral not included in line 219 above that is received as variation margin from counterparties exempt from BCBS-IOSCO margining rules, as laid out in paragraph 2(c) of the BCBS-IOSCO margining requirements for noncentrally cleared derivatives.	
227	Non-financial entities that are not systemically important	Other variation margin received from non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	
228	Sovereigns/Central Banks/MDBs/BIS	Other variation margin received from sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	
230	NSFR derivative assets (derivative assets less cash collateral received as variation margin on derivative assets)	Non-entry field. In calculating NSFR derivatives assets, collateral received in connection with derivatives contracts may not offset the positive replacement cost amount, regardless of whether or not netting is permitted under the bank's operative accounting or risk-based framework, unless it is received in the form of cash variation margin and meets the conditions as specified in paragraph 25 of the Basel III Leverage Ratio Framework and Disclosure Requirements or further specified in any related FAQ. 56	35, FN 16
231	Required stable funding associated with derivative liabilities	Non-entry field. In accordance with paragraph 43(d), the value here equals 20% of derivative liabilities (ie negative replacement cost amounts or negative net replacement cost where applicable) before deducting variation margin posted.	43(d)
232	Total initial margin posted	All cash, securities or other assets posted as initial margin for derivative contracts (eg, including any independent amount received in relation to OTC contracts).	
233	Of which, is initial margin posted on bank's own behalf, of which:	All cash, securities or other assets posted as initial margin for derivative contracts taken on the bank's own behalf. This would not include initial margin posted on behalf of a customer, which should be reported in line 237 below. Where securities or other assets posted as initial margin for derivative contracts would otherwise be included in a category receiving a higher RSF factor, they should be reported within that category and not here. Do not include here cash or other assets provided to contribute to the default fund of a CCP, which should be	42(a)
234	Initial margin posted in the form of cash	reported in line 245 below. Cash posted as initial margin for derivative contracts taken on the bank's own behalf. This would not include initial margin posted on derivative contracts taken on behalf of a customer, which should be reported in line 237 below.	
235	Initial margin posted in the form of Level 1 securities	Initial margin posted in the form of Level 1 securities for derivative contracts taken on the bank's own behalf. This would not include initial margin posted on derivative contracts taken on behalf of a customer, which should be reported in line 237 below.	

NSFR derivative assets = (derivative assets) – (cash collateral received as variation margin on derivative assets)

Row	Heading	Description	Basel III NSFR standards reference
236	Initial margin posted in the form of all other collateral	Initial margin posted in the form of collateral other than cash or Level 1 securities taken on the bank's own behalf and included in above in rows 234 to 235. This would not include initial margin posted on derivative contracts taken on behalf of a customer, which should be reported in line 237 below.	
237	Of which, is initial margin posted on behalf of a customer	Cash, securities or other assets posted as initial margin posted on behalf of a customer, where the bank does not guarantee performance of the third party. Balances reported here should not be included in lines 234 to 236 above.	FN 18
239	Initial margin posted on bank's own behalf, in the form of any collateral type, according to residual maturity of associated derivative contract(s)	All cash, securities or other assets posted as initial margin for derivative contracts taken on the bank's own behalf. Report initial margin balances in this category according to the residual maturity of the derivative contract(s) directly associated with the initial margin. In the case of pooled collateral, report the maturity of initial margin balances according to the maturity of the derivative contract with the longest term in the applicable netting set that contributes to an initial margin requirement. Contracts that are fully offsetting (ie long and short positions in identical contracts) and do not contribute to an initial margin requirement may be excluded from the determination of maturity. This category would not include initial margin posted on derivative contracts taken on behalf of a customer. The sum of this category should be equal to the sum of rows 234 to 236 above.	
241	Initial margin posted on bank's own behalf, in the form of any collateral type, to counterparties exempt from BCBS-IOSCO margin requirements; of which:	Cash, securities or other assets posted as initial margin for derivative contracts taken on bank's own behalf to counterparties exempt from BCBS-IOSCO margin requirements for non-centrally cleared derivatives. This category would not include initial margin posted on derivative contracts taken on behalf of a customer.	
242	Non-financial entities that are not systemically important	Cash, securities or other assets posted as initial margin for derivative contracts to non-financial entities that are not systemically important and are exempt from BCBS-IOSCO margin requirements.	
243	Sovereigns/Central Banks/MDBs/BIS	Cash, securities or other assets posted as initial margin for derivative contracts to sovereign/Central Bank/MDB/BIS counterparties exempt from BCBS-IOSCO margin requirements.	
245	Cash or other assets provided to contribute to the default fund of a CCP	Cash or other assets provided to contribute to the default fund of a CCP. Do not include here cash, securities or other assets posted as initial margin for derivative contracts, which should be included in categories above.	42(a)
246	Required stable funding associated with initial margin posted and cash or other assets provided to contribute to the default fund of a CCP	Non-entry field. In accordance with paragraph 42(a), required stable funding associated with initial margin posted and cash or other assets provided to contribute to the default fund of a CCP.	42(a)
247	Items deducted from regulatory capital	Includes all items deducted from Basel III regulatory capital.	43(c)

Row	Heading	Description	Basel III NSFR standards reference
248	Trade date receivables	The amount of receivables arising from sales of financial instruments, foreign currencies and commodities that (i) are expected to settle within the standard settlement cycle or period that is customary for the relevant exchange or type of transaction, or (ii) have failed to, but are still expected to, settle.	36(d)
249	Interdependent assets	National supervisors have discretion in limited circumstances to determine interdependent assets and liabilities in accordance with paragraph 45 of Basel III NSFR standards. Report here asset items which, on the basis of contractual arrangements, are interdependent on corresponding liabilities report above in line 75 such that: the liability cannot fall due while the asset remains on the balance sheet, the principal payment flows from the asset cannot be used for something other than repaying the liability, and the liability cannot be used to fund other assets. For interdependent items, supervisors may adjust RSF and ASF factors so that they are both 0%, subject to the following criteria: The individual interdependent asset and liability items must be clearly identifiable. The maturity and principal amount of both the liability and its interdependent asset should be the same. The bank is acting solely as a pass-through unit to channel the funding received (the interdependent liability) into the corresponding interdependent asset. The counterparties for each pair of interdependent liabilities and assets should not be the same.	45
251	All other assets not included in above categories that qualify for 100% treatment	Include the carrying value of all other assets not included in the above categories.	43(c)
B2)	Off-balance sheet items		
255	Irrevocable and conditionally revocable liquidity facilities	Balances of undrawn committed liquidity facilities extended by the bank that are either irrevocable or conditionally revocable.	47
256	Irrevocable and conditionally revocable credit facilities	Balances of undrawn committed credit facilities extended by the bank that are either irrevocable or conditionally revocable.	47
257	Unconditionally revocable liquidity facilities	Balances of undrawn liquidity facilities where the bank has the right to unconditionally revoke the undrawn portion of these facilities.	47
258	Unconditionally revocable credit facilities	Balances of undrawn credit facilities where the bank has the right to unconditionally revoke the undrawn portion of these facilities.	47
259	Trade finance-related obligations (including guarantees and letters of credit)	Balances of trade finance-related obligations (including guarantees and letters of credit)	47
260	Guarantees and letters of credit unrelated to trade finance obligations	Balances of guarantees and letters of credit unrelated to trade finance obligations.	47
261	Non-contractual obligations, such as:		

Row	Heading	Description	Basel III NSFR standards reference
262	Debt-buy back requests (incl related conduits)	Potential requests for debt repurchases of the bank's own debt or that of related conduits, securities investment vehicles and other such financing facilities.	47
263	Structured products	Structured products where customers anticipate ready marketability, such as adjustable rate notes and variable rate demand notes (VRDNs).	47
264	Managed funds	Managed funds that are marketed with the objective of maintaining a stable value such as money market mutual funds or other types of stable value collective investment fund, etc.	47
265	Other non-contractual obligations	Other non-contractual obligations not entered above.	47
266	All other off balance-sheet obligations not included in the above categories	All other off balance-sheet obligations not reported in lines 255 to 265 above. Please refer to the instructions from your supervisor for the specification of this item.	47

6.2.3 For completion only by banks in jurisdictions where deposits between banks within the same cooperative network are required by law to be placed at the central organisation and are legally constrained within the cooperative bank network as minimum deposit requirements (panel D)

Panel D is only intended for banks in jurisdictions where deposits between banks within the same cooperative network are required by law to be placed at the central organisation and are legally constrained within the cooperative bank network as minimum deposit requirements, and where such banks are applying the treatment in footnote 10 of the Basel III NSFR standards.

- The reporting institution must be the centralised institution of a cooperative network which has supervisory approval to use this treatment. **All other banks should leave this section blank.**
- This section should only be used to report deposits that exist between banks within the same cooperative network, provided they are required by law to be placed at the central organisation and are legally constrained within the cooperative bank network as minimum deposit requirements.
- The total amount of funding reported in this section should be equal to that reported in line 39 above.

This section **should not** be used by banks within the same cooperative network where deposits are placed in the context of common task sharing and legal, statutory or contractural arrangements.

Row	Heading	Description	Basel III NSFR standards reference
277– 294	Categories are identical to those reported in Panel A	Category definitions are identical to rows 6 to 76 with the exception of rows 42 to 47 where all secured borrowings and liabilities may be reported and, unlike the first panel, there is no qualification on the type of assets used as collateral.	FN 10

6.2.4 For completion only by banks with assets encumbered for exceptional central bank liquidity operations, where the supervisor and central bank have agreed to a reduced RSF factor (Panel E)

Panel E collects data on assets that are encumbered for exceptional central bank liquidity operations. In accordance with paragraph 31 and FN 15 of the Basel III NSFR standards, this section should only include those balances where the supervisor and central bank have agreed to a reduced RSF factor. All other banks should leave this section blank. Values reported in this section should not be reported in Panel B(1) above to avoid double counting.

Row	Heading	Description	Basel III NSFR standards reference
302– 378	Categories are identical to those reported in Panel B	Category definitions are identical to rows 88 to 205.	31, FN 15

7. Trading book

(removed)

8. Standardised approach to credit risk

The templates covering the standardised approach to credit risk are designed to collect data for two purposes: first, to identify the impact of the changes to the approach as set out in the consultative document on Revisions to the Standardised Approach for credit risk (in this section, "the consultative document"), ⁵⁷ and second, to assist the Committee in determining the appropriate calibration of the revised approach, including specific design elements. The data requested in these templates will also be used to calibrate the floor to capital requirements under the Internal Ratings-Based Approach. ⁵⁸ As such, it is crucial that all banks (large and small) fill in as many items as possible in these templates.

All banks are expected to provide data in the summary section of the "BB SA general" worksheet in order to provide the Committee with information on the entirety of their credit risk portfolio. It is expected that large banks, particularly including those regularly involved in the QIS process, will provide a full – or close to full – breakdown of exposures in the remainder of the "BB SA general" worksheet, including the relevant IRB data for banks using such approaches. For smaller banks, the breakdown will be expected for all material asset classes. Breakdowns should be provided on a best efforts basis according to the instructions below, including suitable estimates of necessary risk drivers if needed.

For the "BB SA additional" worksheet, all banks are expected to fill in panels A and B. IRB banks will also provide information in panels C and D.

See <u>www.bis.org/bcbs/publ/d307.pdf</u>.

See the consultative document *Capital floors: the design of a framework based on standardised approaches*; available at www.bis.org/bcbs/publ/d306.htm.

8.1 Paragraph references

	Basel II	Consultative document (see Annex 1)
Exposures to sovereigns	53-56	4-7
Exposures to PSEs	57-58	8-9
Exposures to MDBs	59	10-11
Exposures to banks	60-64	12-18
Exposures to securities firms	65	19
Exposures to corporates	66-68	20-32
Exposures to retail	69-71	33-35
Exposures secured on real estate	72-74	36-48
Past due loans	75–78	
Other assets	81	61
Off-balance sheet exposures	82-89	49-59
CRM	109-210	61-138

8.2 The "BB SA general" worksheet

The "BB SA general" worksheet consists of a summary panel, and sections for each asset class. In each asset class section, the rows refer to different buckets within that asset class partitioned by both the existing and proposed exposure buckets. The columns refer to different data items needed for the exposures collected in each row.

The summary table at the top of the worksheet (rows 6 to 21) collects aggregate data by asset class as well as additional information on partial use, expected loss (EL) and provisions, as necessary to assess the impact of the proposed capital floor. Rows 8 to 19 and row 21 should include only performing (non-defaulted) exposures. All past due or defaulted exposures – according to the IRB definition in Basel II, paragraph 452 – should be included in row 20.

Following the summary panel, there is a cross-check panel (rows 22 to 51) to summarise what is included in the asset class sections. The Committee acknowledges that requiring a breakdown for small sets of niche exposures may be an undue burden on firms. Such niche exposures should be included in the summary table but they may be excluded from the breakdown and cross-check panels. However, it is expected that the vast majority of exposures will be included in the breakdown panels (certainly for large banks typically participating in QIS exercises).

Counterparty credit risk exposures such as non-cleared derivatives and securities financing transactions should be included in these panels. For columns I to K, banks can use their current approach to counterparty credit risk; however, for columns L to N banks will have to use supervisory haircuts for SFTs; and SA-CCR for OTC derivatives. In the IRB-only columns, banks can use their current approach to counterparty credit risk (including IMM and repo-VaR).

Exposures treated under other specific frameworks (eg securitisation, equity investments in funds and exposures to central counterparties) should **not** be included in the BB SA worksheets.

8.2.1 Options

The current standardised approach contains a number of options for the treatment of certain asset classes. These are reflected in column C. Banks should only report data for the relevant option which has been implemented in their jurisdictions.

In the proposed revisions, the Committee is considering several alternatives for particular asset classes as outlined in the consultative document and these instructions. These options are all included in the template in different panels (eg see panel F. Retail exposures) Banks should complete data requested for all options under consideration. The total exposure for each option should be the same.

8.2.2 Background on column definitions

- Data should be provided in the reporting unit and currency specified on the "General Info" worksheet.
- On-balance sheet exposures (column G) should be net of specific provisions.
- Non-performing loans (NPL) (column O) should represent the losses for each portfolio experienced during the last reporting year.
- Exposures treated under the Simple Approach for recognising collateral or for which guarantees
 allow risk weight substitution should be included in the rows relevant for the **direct** obligor for
 on/off balance sheet exposure and pre-CRM columns (ie columns G to I and L). For all other
 columns, these exposures should be included in the rows relevant for the **guarantor** or
 collateral originator.
- IRB banks should submit data in columns P to W. Data should be provided aggregating all exposures that meet the criteria set out in columns C to F. Banks without approval for the IRB approach to credit risk should leave columns P to W empty. (Note: As for SA columns, rows 8 to 19 and row 21 should include all performing exposures. Defaulted exposures will be reported in row 20 only.)
- For the cross-check and breakdown panels (row 22 onwards) IRB parameters, average risk weight, and EL amounts provided in columns P to S, should not include defaulted exposures. Columns T to W are not required for the cross-check or breakdown panels.
- Columns Q to T should be left blank for rows in which the bank's IRB exposure as reported in column P is zero. Similarly, column V should be left blank in rows with zero exposure subject to partial use as reported in column U.
- Provisions in column W should be the amount of total eligible provisions according to paragraph 380 of the Basel II framework.
- Average RW is defined as total RWA / total EAD. For the calculation of the average RW, PD and LGD should include the applicable floors.
- Average PD and LGD should be calculated on an exposure-weighted basis. For this purpose, PD and LGD should not be floored.

8.2.3 Background on row definitions

- The set of exposures to be aggregated in each row is determined in columns C to F. These categories refer to the options and definitions given in the Basel II standardised approach, the consultative document, and these instructions.
- All exposures should be categorised based on their treatments under the existing and revised standardised approaches. Current IRB treatments will therefore be reflected in the IRB RWA column only.

8.3 Asset class specific information in "BB SA general"

8.3.1 Sovereigns, exposures to PSEs and MDBs

- Although sovereign exposures are out of the scope of the consultation, they are included in the
 QIS for completeness in order to assess the impact of revisions with respect to total current
 capital requirements under the standardised approach.
- Sovereign claims on official entities as given in Basel II paragraph 4 and paragraph 7 of Annex A of the consultative document should be included in row 54.
- The current treatment for exposures to MDBs is given in Basel II paragraphs 5 and 6. In some cases MDBs may currently be treated as banks.
- Exposures to MDBs should be included in panel 4 (rows 93 to 100) according to the treatment proposed in given in paragraphs 10 and 11 of the Annex A of the consultative document. MDB exposures that are currently treated as banks but under the proposed revisions should be treated as corporates are to be reported in row 100 (just for columns G and H). Note however that these data will be used as memorandum item only to assess how many MDBs will be treated as corporates. These cells will not add-up to the total exposures. Instead, these MDB exposures that will be treated as corporates under the proposed revisions should also be reported in the corporate panel.

8.3.2. Exposures to banks

- Banks should report data in panel 1 wherever possible, ie where the obligor's disclosure is sufficient. Within a jurisdiction, banks will only fill in rows labelled option 1 or option 2 as appropriate, referring to the method currently in use in their jurisdiction under Basel II. Under option 1, it is the rating of the obligor's home sovereign that is used. Panel 2 should be used where data for assigning the proposed new buckets are unavailable.
- To calculate the CET1 ratio, banks must use their obligor banks' most recent Pillar 3 reports. To
 calculate the net NPA ratio, banks must use their obligor banks' most recent financial year data
 available. In all circumstances, the CET1 ratio and the net NPA ratio should be calculated for the
 same reporting period.
- The net NPA ratio risk driver must be calculated according to paragraph 15 of Annex 1 of the consultative document, and **not** following the Basel II framework definition of default (paragraph 452). (Note that, as mentioned above, for column O, NPL follows the definition of the default described in the Basel II framework, paragraph 452.)
- The CET1 ratio and the net NPA ratio should be calculated for the obligor bank and not its upstream consolidated group, that is at the consolidated or sub-consolidated level as specified in Basel Committee guidance. If the information is not available for the obligor entity, estimates of these ratios may be used for the purposes of the QIS. These estimatesmay be based either on the relevant ratios at the consolidated group or on other suitable regulatory or accounting data. The CET1 ratio and the net NPA ratio should be calculated for the same entity.
- Banks must apply a 300% risk weight to exposures:
 - to banks for which the relevant information is not published within the frequency required by the Pillar 3 disclosure requirements; and
 - in cases where they are aware that one of their obligor banks has breached any binding minimum prudential standard to which they are subjected by their national supervisor.
- For short term claims, banks may apply a 20 percentage point reduction to the risk weight assigned by the table in paragraph 13 of Annex 1 of the consultative document, provided that

such risk weight is no greater than 100%. The risk weight applied to short-term claims will in no case be lower than 30%. Short-term claims are defined as interbank claims having an original maturity of three months or less that are not expected to be rolled over. (See paragraph 18 of Annex 1 of the consultative document.)

8.3.3 Corporate

- Note that for the purposes of the QIS, revenue, leverage and profitability data need not be from published accounts if the bank has reliable evidence that these data are accurate.
- If data are insufficient to categorise senior corporate exposures in panel 1 (rows 565 to 734 and 735 to 856), they should be categorised as such in panel 2 (rows 858 to 918), either as "New company < 1 year old" (as covered in paragraph 26) or "No leverage/revenue data".
- Remaining corporate exposures (as covered in paragraphs 28 to 32) should also be categorised in panel 2.

Option A

- For Corporate option A, the template is collecting data on revenue crossed with leverage.
- The obligor's revenue and leverage used for assigining risk weights under paragraph 22 of Annex A of the consulative document should be those of the legal entity against which the exposure is held. If the information is not available on an obligor entity basis, ratios for the consolidated group of the obligor entity could be used.
- To calculate leverage and revenue, banks must use the obligor's year-end accounts for the most recent financial year available. Revenue includes all income the company received from business activities as determined by the accounting standards of the relevant jurisdiction (ie it is a sales measure not a profit measure). Leverage means total assets/total equity, where both total assets and total equity are determined by the accounting standards of the relevant jurisdiction.

Option B

- For Corporate option B, the template is collecting data on profitability crossed with leverage.
- Profitability is defined as (Earnings Before Interest Taxes, Depreciation and Amortisation -EBITDA) / (Total Assets - TA) for audited companies only. Non-audited firms should be treated as a separate category. The buckets for profitability are:
 - EBITDA/TA ≤ 0%
 - 0% < EBITDA/TA ≤ 5%
 - 5% < EBITDA/TA ≤ 10%
 - 10% < EBITDA/TA
 - Non-audited firms
- RWA data is not collected for corporates panel 1B, as no risk weights have been proposed for this methodology.

General remarks for Options A and B

• Exposures that are currently treated under the supervisory slotting criteria approach (Basel II paragprah 249 ff) should be included in the summary information panel (Rows 6-21). Where material, these exposures should also be included in the breakdown panels as appropriate, but need not be reflected in columns P to W for IRB exposures.

Paragraphs 28 and 29 of the consultative document specify that risk weights should be the
higher of counterparty risk weight and the applicable specialised lending risk weight. However,
for the purposes of the QIS, exposures may be assigned the corresponding specialised lending
risk weight (120% or 150%) to calculate risk-weighted assets, without comparing to the
counterparty risk weight.

8.3.4 Exposures secured by residential real estate (RRE)

Option A

Option A collects exposures according to loan-to-value (LTV) and debt service coverage (DSC) ratios, as defined in paragraphs 40 and 41, respectively, of Annex 1 in the consultative document.

Option B

- Option B collects exposures according to loan-to-value (LTV) and debt-to-income (DTI) ratios.
- DTI should be calculated as follows:
 - "Debt" is the total amount of the loan at origination as defined in the first bullet point of paragraph 40 of Annex 1 in the consultative document, plus any other known debt.
 - "Income" is the obligor's total income, as described in the second bullet point of paragraph 41 of Annex 1 in the consultative document, calculated on an annual basis.

General remark for both options

• "Currency of main income" cells should include the total of all exposures for the applicable asset type where the loan and the primary income of the borrower is in the same currency.

8.3.5 Exposures secured by commercial real estate (CRE)

Option A

- Option A in paragraph 44 of Annex 1 of the consultative documents risk weights exposures secured by commercial real estate according to the risk weight applicable to an unsecured exposure to the counterparty. This is grouped in buckets in column D.
- Column E collects data to assess the impact of the national discretion allowed under conditions laid out in footnote 59 of the consultative document.

Option B

- Option B in paragraph 45 and 46 of Annex 1 of the consultative documents risk weight exposures secured by commercial real estate based on LTV ratio.
- Exposures reported in this panel must meet the requirements in paragraph 45 of Annex 1 of the consultative document. In this regards, see paragraph 47.
- According to Column F, exposures should be distinguished between:
 - Exposures secured by mortgages on office and/or multipurpose commercial premises and/or multi-tenanted commercial premises, and
 - Other exposures.

8.3.6 Retail

- "Currency of main income" cells should include the total of all exposures for the applicable asset type where the loan and the primary income of the borrower is in the same currenty.
- Cells should be filled in according to new standardised approach definitions, eg with the 0.2% granularity limit. However, where applying the new retail definition criteria would be an excessive burden, the retail definitions currently in use may be applied.
- "Length of established relationship" is defined in the following way: the length of time for which a customer has had at least one product from the bank either a transactional account or credit product.
- DSC is defined as per the definition for residential real estate esposures (see paragraph 41 of Annex 1 of the consultative document).

8.4 Background on the "BB SA additional" worksheet

The "BB SA additional" worksheet collects further information on exposures to banks, exposures secured by residential real estate collateral, exposures with a currency mismatch (ie where the currency of the loan is different from that of the borrower's main source of income); as well as for estimated credit conversion factors (CCFs) under the Advanced IRB approach.

8.4.1 Panel A – Exposures to banks

- Exposure for top counterparties should be measured by post-CCF, pre-CRM exposure under the proposed standardised aproach.
- Consistent with the proposal, the top counterparty determination should be made at the legal entity level, not the consolidated group level. As a reminder, panel A is identifying the top 20 counterparties that qualify to be risk weighted as bank under the proposal.

8.4.2 Panel B – Residential real estate exposures

- Panel B is intended to identify differences in debt service coverage (DSC) ratios net of taxes and gross of taxes.
- For each obligor in the bank's residential real estate portfolio, the DSC should be computed according to paragraph 41 of Annex A of the consultative document. This is termed the "net basis DSC". The "gross basis DSC" should be computed for each obligor applying the definition of paragraph 41 but without deducting taxes.
- Exposures in the RRE portfolio should be grouped into deciles according to the net basis DSC (lowest 10%, next 10% and so on to highest 10%). For each decile group, column D should be populated with the mean of the net basis DSC for those exposures, and column E should be populated with the mean of the gross basis DSC for those exposures.

8.4.3 Panel C – Exposures with a currency mismatch

Stressed PD and LGD may be calculated as follows.

- $LGD_{Stress} = (LGD+r) / (1+r)$, where r is respectively, 40% and 77% depending on the assumed FX rate stress.
- One of the possible solutions to calculate stressed PD is to use a distance-to-default approach (used eg similarly to the KMV model). One approach assumes that the obligor defaults when the level of the monthly instalments (capital plus interest) exceeds a specific portion of an

- obligor's monthly income (this value should be lower than 100% since the obligor has additional expenses apart from repaying his debt).
- Looking at an individual obligor, a stressed debt service coverage ratio can be based on his current debt service coverage: $DSC_{Stress} = DSC_{Current} * (1+r)$. The portion of the portfolio for which DSC_{Stress} exceeds the given threshold, is assumed to be in default. The "newly defaulted" portion of portfolio is a proxy for the necessary uplift in PD to give the stressed PD.
- If the stressed PD or LGD is calculated in another way please provide an explanation via your national supervisor.

8.4.4 Panel D – CCFs applied by AIRB banks

- This panel collects data on credit conversion factors (CCFs) and should be filled in by banks using advanced IRB. For each of the categories specified in rows 54 to 62, banks should provide the total amount of the facility limits (including both drawn and undrawn exposures) for such exposures, the total amount of undrawn exposures, and the average credit conversion factor as a percentage.
- Paragprahs 49 to 59 of Annex A of the consultative document provide more details on the category definitions for panel D.

9. TLAC

9.1 Scope of application

The TLAC proposals apply at various different levels within a G-SIB. These are set out in the following table. Please refer to the TLAC Term Sheet⁵⁹ carefully when filling out TLAC related worksheets.

TLAC proposals

Element of TLAC proposals	Application	Term sheet reference	QIS worksheets
External TLAC requirement – SPE	G-SIB consolidated level	Section 9	G-SIB TLAC External + G-SIB TLAC External (location)
External TLAC requirement – MPE	G-SIB consolidated level 60 + sub consolidated level for each resolution group	Section 9	General info for MPE+ G-SIB TLAC External + G-SIB TLAC External (location: resolution entity level only)
Internal TLAC	For each G-SIB material subsidiary	Section 21	G-SIB TLAC Internal
Treatment of holdings of TLAC	All non-G-SIB internationally active banks (consolidated basis)	Section 18	TLAC holdings (non-GSIB)

⁵⁹ See www.financialstabilityboard.org/wp-content/uploads/TLAC-Condoc-6-Nov-2014-FINAL.pdf.

TLAC minimum requirements will be applied to each resolution entity within a G-SIB. For MPE, this may not necessarily include meeting TLAC at the G-SIB consolidated group level if this is not itself a resolution entity. However, for the purpose of the QIS and in order to understand any differences between the application of the requirements between SPE and MPE G-SIBs, MPE G-SIBs should also complete the QIS at the consolidated level.

G-SIBs under a Single-Point-of-Entry (SPE) strategy will fill out the "G-SIB TLAC External" and "G-SIB TLAC location" worksheets on a **consolidated** basis. G-SIBs under a Multiple-Point-of-Entry (MPE) strategy will fill out the "G-SIB TLAC External", "G-SIB TLAC location" and "General info for MPE firms" worksheets for each resolution group level, ie data should be **sub-consolidated** at the level of a resolution group. Each resolution group should fill out a separate "TLAC MPE reporting template". In addition, MPE G-SIBs will also fill out the "G-SIB TLAC External" worksheet on a **consolidated** level⁶⁰ by filling out the Basel III monitoring reporting template. The purpose of this is to analyse any differences between SPE and MPE strategies.

All G-SIBs, both SPE firms and each resolution entity of MPE firms, should fill out the "Internal TLAC" worksheet to calculate and sum up internal TLAC requirements of material subsidiaries. If the material subsidiary is located in a jurisdiction that has not yet implemented Basel III, then data should be provided on the basis that Basel III is implemented and fully phased-in (on a best efforts basis).

Each G-SIB needs to work closely with its home supervisor (who will in turn coordinate with the relevant Crisis Management Group (CMG)) to confirm the approach according to its resolution strategy.

Non-G-SIBs will complete the "TLAC holdings" worksheet. All data in this worksheet should be filled out on a banking group consolidated basis.

All banks should enter the TLAC related amounts (eg when calculating regulatory capital as well as TLAC liabilities that are not included in regulatory capital) according to the national implementation of the Basel III standards fully phased in as in 2022.

9.2 General info for MPE

G-SIBs under an SPE strategy **do not need to fill out this section** as they will already complete the "General info" worksheet at the consolidated level as part of the Basel III monitoring exercise.

For G-SIBs under an MPE strategy, each resolution group will fill out the "General Info for MPE" worksheet **on a sub-consolidated basis**. The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description		
A1)	A1) Reporting data				
3	С	Country code	Leave blank		
4	С	Region code	Leave blank		
5	С	Resolution Group number	Leave blank		
6	С	Bank number	Leave blank		
7	С	Domestic surcharges, CET1 capital	Leave blank		
8	С	Domestic surcharges, Tier 1 capital	Leave blank		
9	С	Domestic surcharges, total capital	Leave blank		
10	С	Conversion rate (in euros/reporting currency)	Leave blank		
11	С	Submission date (yyyy-mm-dd)	Leave blank		
12	С	Reporting date (yyyy-mm-dd)	Date as of which all data are reported in worksheets.		
13	С	Reporting currency (ISO code)	Three-character ISO code of the currency in which all data are reported (eg USD, EUR).		

Row	Column	Heading	Description
14	С	Unit (1, 1000, 1000000)	Units (single currency units, thousands, millions) in which results are reported.
15	С	Accounting standard	Indicate the accounting standard used.
16	С	Accounting total assets	Total assets following the relevant accounting balance sheet (considering the regulatory consolidation).
20	С	Total capital	Automatic calculation
21	С	Total Common Equity Tier 1 capital (after regulatory adjustments)	Enter the amount of Total Common Equity Tier 1 capital after regulatory adjustments on a sub-consolidated (ie resolution group) basis. This includes CET1 issued from the resolution entity and CET1 issued from subsidiaries within the same resolution
			group (subject to the limits specified in paragraphs 62 to 64 of the Basel III framework). Please refer to DefCap D64 for the calculation method.
22	С	Additional Tier 1 capital (after regulatory adjustments)	Enter the amount of Additional Tier 1 capital after regulatory adjustments on a sub-consolidated basis.
			This includes AT1 instruments issued from the resolution entity and AT1 instruments issued from its subsidiaries withi the same resolution group (subject to the limits specified in paragraphs 62 to 64 of the Basel III framework. Please refer to DefCap D82 for the calculation method.
23	С	Tier 1 capital (after regulatory adjustments)	Automatic calculation
24	С	Tier 2 capital (after regulatory adjustments)	Enter the amount of Tier 2 capital after regulatory adjustments on a sub-consolidated basis
27	С	Total RWA at the reporting date before application of the transitional floors	Enter the amount of risk-weighted assets of the resolution group on a sub-consolidated basis as at the reporting date before any changes for RWA effects from Basel III definition of capital and other national phase-in arrangements.
28	С	RWA impact of applying future definition of capital rules	Enter the amount of the RWA impact of applying fully the phased-in national implementation of the Basel III definition of capital. If items which will be deducted in the fully phased in treatment are currently risk weighted, this amount should be reported as a negative number.
29	С	RWA impact of national phase- in arrangements for CVA if any	Incremental impact of full implementation of the national CVA capital requirements. If the CVA capital requirements have already been fully phased-in, banks should report 0.
30	С	RWA impact of any other national phase-in arrangements	Incremental impact of full implementation of the national implementation of Basel III capital requirements. If the capit requirements have already been fully phased-in or no phase in agreements exist, banks should report 0.
31	С	Total RWA	Automatic calculation
34	С	Leverage ratio exposure measure	Enter the exposure amount of the Basel III leverage ratio (published in January 2014) for the resolution group on a sub-consolidated basis. Please refer to J103 in the "Leverage Ratio" worksheet in the Basel III monitoring reporting template.

Row	Column	Heading	Description
40, 43, 46	C, D	Deduction for investments in the capital of banking, financial and insurance entities that are outside the scope of regulatory consolidation and where the bank does not own more than 10% of the issued common share capital (excluding amounts held for underwriting purposes only if held for 5 working days or less)	Please refer to rows 53, 76 and 96 in the DefCap worksheet in the Basel III monitoring reporting template.
41, 44, 47	C, D	Deduction for significant investments in the capital of banking, financial and insurance entities that are outside the scope of regulatory consolidation (ie where the bank owns more than 10% of the issued common share capital or where the entity is an affiliate), excluding amounts held for underwriting purposes only if held for 5 working days or less	Please refer to rows 55, 77 and 97 in the DefCap worksheet in the Basel III monitoring reporting template.
52	C, D	Holdings of Common Equity Tier 1 net of short positions subject to risk weighting treatment	Please refer to row 106 in the DefCap worksheet in the Basel III monitoring reporting template.
53	C, D	Holdings of Additional Tier 1 capital net of short positions subject to risk weighting treatment	Please refer to row 107 in the DefCap worksheet in the Basel III monitoring reporting template.
54	C, D	Holdings of Tier 2 capital net of short positions subject to risk weighting treatment	Please refer to row 108 in the DefCap worksheet in the Basel III monitoring reporting template.
56	C, D	Holdings of Common Equity Tier 1 net of short positions subject to risk weighting treatment	Please refer to row 110 in the DefCap worksheet in the Basel III monitoring reporting template.
57	C, D	Holdings of Additional Tier 1 capital net of short positions subject to risk weighting treatment	Please refer to row 111 in the DefCap worksheet in the Basel III monitoring reporting template.
58	C, D	Holdings of Tier 2 capital net of short positions subject to risk weighting treatment	Please refer to row 112 in the DefCap worksheet in the Basel III monitoring reporting template.

9.3 External TLAC and deduction treatment for G-SIBs according to the national implementation of Basel III

The "G-SIB TLAC External" worksheet collects **consolidated data for each resolution group** of a G-SIB, as defined by the relevant CMG. For **SPE G-SIBs**, there is only one resolution group which is equal to the group consolidated level.

For **MPE G-SIBs**, there will be more than one resolution group and therefore each resolution group needs to prepare a "G-SIB TLAC External" worksheet at the sub-consolidated level (by using separate files, ie "TLAC MPE reporting template 144-"). In addition, MPE G-SIBs should also prepare this worksheet at a group consolidated level (like SPE firms). When doing so, they should assume that the top parent company is the relevant resolution entity and that the resolution group is equal to the group consolidated level.

Panel A of this worksheet identifies the regulatory capital instruments issued by each resolution group which is partially based on information that G-SIBs already provide as part of the "DefCap" worksheet. **Panel B** identifies liabilities that do not qualify as regulatory capital instruments but that might either qualify as TLAC or that could potentially be replaced by TLAC-eligible debts once liabilities mature. The worksheet is designed to test four different "cases" using a menu approach to test the impact of specific features of TLAC instruments as follows:

- **Case 1** includes all non-regulatory capital instruments that meet all of the criteria on the proposed TLAC term sheet.
- Case 2 includes all non-regulatory capital instruments that meet all of the criteria on the TLAC
 term sheet plus additional criteria that are currently applied to Tier 2 instruments under
 Basel III. The purpose of gathering this data is to analyse the impact of requiring these
 additional features.
- Case 3 analyses the subordination requirement. Case 3 includes all instruments that are captured in case 1 plus instruments that meet the TLAC term sheet criteria except for subordination. The purpose of case 3 is to analyse the impact of requiring TLAC instruments to be subordinated and how this impacts on the shortfalls observed in G-SIBs (by comparison with case 1).
- Case 4 is the widest case and includes all unsecured liabilities except liabilities arising from derivatives, liabilities not arising through contract, or deposits. Cases 4i and 4ii differ only by the subordination requirement. The purpose of case 4 is to gain insights into liabilities that could potentially, once matured, be replaced by liabilities that do qualify for TLAC. Please read the general instructions on panel B below carefully before filling out this part of the worksheet.

Panel C of this worksheet collects information on holdings of TLAC instruments following the same four cases set out above.

Panel D should be completed only by EU G-SIBs. This information is being used by the European Banking Authority to conduct analysis relating to the implementation of minimum requirement for own funds and eligible liabilities (MREL) under the EU Banking Recovery and Resolution Directive (BRRD).

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description	
A)	Regulatory	Regulatory capital		
4	С	Is the resolution entity a holding company or intermediate holding company?	Please select "yes" if the resolution entity is a non-operating holding company or intermediate holding company, or "no" if the resolution entity is the bank operating company.	
8	С	Total Common Equity Tier 1 capital (after regulatory adjustments)	Automatic calculation.	
9	С	Total Additional Tier 1 instruments (after regulatory adjustments); of which:	Automatic calculation.	

Row	Column	Heading	Description
10	С	Additional Tier 1 instruments issued under third country law without any contractual bail-in clause and no equivalent binding statutory provision for cross-border recognition of resolution actions (before regulatory adjustments)	This only includes those Additional Tier 1 instruments included in row 9 that are not issued under the law of the resolution entity and where (i) the instrument does not contain contractual terms that allow the home authority to expose the instruments to loss at the point of non-viability or in resolution and (ii) the law under which the instruments are issued does not recognise bail-in by the home jurisdiction of the resolution entity on a statutory basis. Please enter the amount before regulatory adjustments (eg deductions) when completing this cell. This means that the number in cell C10 could be larger than the number in C9.
11	С	Total Tier 2 instruments issued by the resolution entity (prior to amortisation, ie nominal amount) with a residual maturity ≥ 1 year (after other	This includes all Tier 2 capital instruments issued by the resolution entity that have a residual contractual maturity of more than one year. Numbers should be given before amortisation, ie before applying any prudential amortisation schedules (but after
		regulatory adjustments, eg corresponding deduction); of which	other regulatory adjustments, eg corresponding deduction under the Basel III framework if relevant).
12	С	Tier 2 instruments issued under third country law without any contractual bail-in clause and no equivalent binding statutory provision for cross-border recognition of resolution actions (prior to regulatory adjustments)	This only includes those Tier 2 instruments included in row 11 that are not issued under the law of the resolution entity and where (i) the instrument does not contain contractual terms that allow the home authority to expose the instruments to loss at the point of non-viability or in resolution and (ii) the law under which the instruments are issued does not recognise bail-in by the home jurisdiction of the resolution entity on a statutory basis. Please enter the nominal amount before regulatory
			adjustments (eg deductions) when completing this cell. This means that the number in cell C12 could be larger than the number in C11.
13	С	Total Tier 2 instruments issued by subsidiaries (after amortisation) eligible to count towards the consolidated Tier 2 capital of the resolution group (with a residual maturity ≥ 1	This includes all Tier 2 capital instruments issued by subsidiaries of the resolution entity and eligible to count towards the resolution group's Tier 2 position (as determined by the limits specified in paragraphs 62 to 64 of the Basel III framework) that have a residual contractual maturity of more than one year.
		year)	Numbers should be given on an amortised basis , ie after applying any prudential amortisation schedules.
14	С	Total Tier 2 instruments (after amortisation but before other regulatory adjustments eg corresponding deduction) with a residual maturity < 1 year	This includes all Tier 2 capital instruments issued by the resolution entity or its subsidiaries within the same resolution group (subject to the limits specified in paragraphs 62 to 64 of the Basel III framework) that have a residual contractual maturity of less than one year to the extent that these are allowed to qualify as regulatory capital requirements under the domestic implementation of Basel III.
			Numbers should be given on an amortised basis , ie after applying any prudential amortisation schedules. Please enter the amount (after amortisation but before regulatory adjustments eg corresponding deductions) when completing this cell.

Row	Column	Heading	Description
11–14	D	Average residual maturity (Years)	Please enter the weighted average residual (contractual) maturity of the instruments entered in cells C11 to C14.
			For perpetual instruments, you should assume a residual maturity of 10 years.
			For instruments in cells C11 to C14 the average maturity should be calculated before applying amortisation schedules, ie all instruments should be weighted on a nominal basis.
			The weighted average residual maturity should be calculated by the following formula.
			$\sum rac{ extit{Residual maturity} \cdot extit{Amount with the residual maturity}}{ extit{Total amount}}$
8–10, 14	E	Of which issued externally out of non-resolution entities	Enter the amount of the capital instruments entered in cells C8 to C10 and C14 that have been issued by subsidiaries within the same resolution group. Please enter the amount before regulatory adjustments eg
			corresponding deductions. This means that the numbers in cells E8 to E14 could be larger than the numbers in cells C8 to C14.
9	F	Of which issued in the form of debt (liabilities in balance sheet)	Enter the amount of Additional Tier 1 capital instruments issued in the form of debt, ie those that are classified as liabilities for accounting purposes.
9	G	Of which issued in the form of debt (not shares)	Enter the amount of Additional Tier 1 capital instruments that are not shares, eg common shares or preferred shares.
8–14	Н	(only for MPE groups) Of which held by entities in other resolution groups	Enter the amount of the capital instruments entered in cells C8 to C14 that have been issued to entities that are part of the same consolidated group but belonging to another resolution group. Please enter the amount before regulatory adjustments eg corresponding deduction when completing this cell.
			Leave these fields blank unless the template is being filled out for an individual resolution group of an MPE bank.

B) Unsecured liabilities

Please use nominal outstanding amounts (rather than accounting values) for rows 19 to 46. Liabilities include bonds and other forms of wholesale funding as well as deposits that meet the criteria outlined below. Throughout rows 19 to 46, please do not include any regulatory capital instruments or the amortised portion of Tier 2 instruments that no longer qualifies as a regulatory capital as these are already captured in cell C11.

'Subordinated liabilities' are liabilities that rank junior to all 'excluded liabilities' (specified in the TLAC term sheet) in the creditor hierarchy of the resolution entity. Excluded liabilities are (a) insured deposits; (b) any liability that is callable on demand without supervisory approval; (c) liabilities that are funded directly by the issuer or a related party of the issuer; (d) liabilities arising from derivatives or debt instruments with derivative-linked features; (e) liabilities arising other than through a contract, such as tax liabilities, (f) liabilities which are preferred to normal senior unsecured creditors under the relevant insolvency law; and (g) any other liabilities that, under the laws governing the issuing entity, cannot be effectively written down or converted into equity by the relevant resolution authority. Please note that in order for liabilities to meet criterion b (not callable on demand without supervisory approval), this could be implemented through supervisory regulations. Banks can therefore assume that all instruments will meet this criterion. Likewise, for criterion c (not funded directly by the issuer or a related party of the issuer), this is subject to the agreement of the CMG. Banks can therefore assume, for the purpose of the QIS, that the CMG agrees that liabilities issued to a parent of a resolution entity can count towards TLAC.

Subordination can be achieved by three methods:

- (a) contractually subordinated to excluded liabilities on the balance sheet of the resolution entity,
- (b) junior in the statutory creditor hierarchy to all excluded liabilities
- (c) issued by a resolution entity that does not have excluded liabilities on its balance sheet (for example a holding company).

Please note that the term sheet allows for G-SIBs from jurisdictions with certain resolution regimes to permit liabilities that rank alongside excluded liabilities to contribute towards TLAC. This is explained in the final two

ROW Column Heading Description	Row	Column	Heading	Description	
--------------------------------------	-----	--------	---------	-------------	--

paragraphs of Section 13 of the term sheet. For the purposes of the TLAC QIS, G-SIBs should not take these into consideration when populating the template. This will be dealt with in the analysis.

The TLAC term sheet specifies that in order to be eligible as TLAC, liabilities must

- be issued and maintained by resolution entities
- be unsecured
- not include insured deposits
- not be callable on demand without supervisory approval
- not arise from derivatives or having derivative-linked features (eg structured notes)
- not arise other than through contract (eg tax liabilities)
- not be preferred to normal senior unsecured creditors
- be able to be effectively written down or converted into equity under the laws governing the issuing entity⁶¹
- not be subject to set off/netting
- not be funded directly by the issuer or a related party of the issuer
- be subordinated in the sense discussed above

Where the worksheet asks for liabilities that satisfy the criteria in the TLAC term sheet you should only include liabilities that satisfy the requirements above. The exception to this approach is the requirement that liabilities should have a residual maturity of ≥ 1 year. **You should include all maturities unless explicitly specified otherwise**.

19	D–G	Case 1: Total subordinated liabilities excluding regulatory capital, issued by the resolution entity, meeting the Term Sheet criteria not taking into consideration instruments qualifying by way of the last two paragraphs of section 13 of the term sheet, of which:	Please enter liabilities that satisfy all of the criteria in the TLAC term sheet (except the final two paragraphs of section 13), as defined in the grey box above. Please split the total amount by residual contractual maturity and enter the corresponding amounts in columns D to G. The sum of columns E to G will give the overall amount of liabilities that satisfy the conditions described in the term sheet (excluding regulatory capital instruments).
20	С	not paid-in	This only covers liabilities that are not paid-in but that have been included in row 19. The number entered in cell C20 only includes liabilities > 1 year residual maturity and cannot be larger than the
21	C	with credit sensitive feature	corresponding number in cell C19. This only covers liabilities that contain 'credit-sensitive-features' (eg where interest payments are tied to the financial condition of the issuer) and that have been included in row 19. The number entered in cell C21 cannot be larger than the corresponding number in cell C19.
22	С	with a right of acceleration of the principal and interest outstanding in respect of the instrument outside liquidation	This only covers liabilities that contain clauses which permit a creditor to accelerate payment of the principal and interest outstanding even if the firm does not enter liquidation, and that have been included in row 19. The number entered in cell C22 cannot be larger than the corresponding number in cell C19.

Instructions for Basel III monitoring

129

Liabilities can be exposed to loss by the relevant authorities if they are within the scope of statutory resolution tools or contain contractual terms that allow them to be written down and/or converted into equity by the resolution authorities. If liabilities are issued under foreign law they have to (a) contain contractual bail-in clauses, (b) contain contractual terms acknowledging that they are subject to resolution actions by the home authorities, or (c) the law that they are governed by must acknowledge resolution actions by the home authorities on a statutory basis.

Row	Column	Heading	Description
23	D-G	including an incentive to redeem	This only covers liabilities that contain explicit incentives to redeem the liability ahead of its contractual maturity date and that have been included in row 19.
			For example, explicit incentives to redeem a liability early may consist of step-up clauses where interest payments increase after a number of years.
			Please split the total amount by residual contractual maturit and enter the corresponding amounts in columns D to G. The number calculated in cell C23 cannot be larger than the corresponding number in cell C19.
24	D-G	including an incentive to redeem (where the effective maturity date is the date of the incentive to redeem)	This covers the same liabilities as in row 23. However, please split the total amount by the effective residual maturity (which is defined as the time remaining until the first step-uclause becomes effective) and enter the corresponding amounts in columns D to G. The sum of columns E to G will give only liabilities for which the effective residual maturity is at least one year.
25	С	subordinated uninsured deposits	This covers any uninsured term deposits that have been included in row 19. The number entered in cell C25 cannot be larger than the
			corresponding number in cell C19.
26	D-G	Case 2: Total subordinated unsecured liabilities excluding regulatory capital, issued by the resolution entity, meeting the Term Sheet criteria, excluding liabilities of row 20,	Please enter only liabilities that satisfy all of the criteria in the TLAC term sheet as well as (a) being paid-in, (b) having no credit sensitive feature, (c) having no right of acceleration of the principal and interest outstanding in respect of the instrument outside liquidation. Please split the liabilities according to their effective residuation.
		21 and 22. Please use effective residual maturity as described in row 24.	maturity, which is defined as the time remaining until the fir incentive to redeem becomes effective (see row 24).
27	D-G	Case 3: Total subordinated liabilities excluding regulatory capital, issued by the	Please enter liabilities that satisfy all of the criteria in the TLAC term sheet plus those satisfying all of the criteria except for the subordination requirement.
		resolution entity, meeting the Term Sheet criteria (same as case 1) plus senior liabilities meeting the term sheet except the subordination criterion (Section 13a, 13b and 13c); of	Please split the total amount by residual contractual maturity and enter the corresponding amounts in columns D to G. The sum of columns E to G will give the overall amount of liabilities that satisfy these conditions. Note: this row includes liabilities that have been included in row 19.
28	С	which:	This only covers liabilities that are not paid-in but that have
		•	been included in row 27. The number entered in cell C28 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C27.
29	С	with credit sensitive feature	This only covers liabilities that contain 'credit-sensitive-features' (eg where interest payments are tied to the financi condition of the issuer) and that have been included in row 27.
			The number entered in cell C29 only includes liabilities >1 year residual maturity and cannot be larger than the corresponding number in cell C27.

Row	Column	Heading	Description
30	С	with a right of acceleration of the principal and interest outstanding in respect of the instrument outside liquidation	This only covers liabilities that contain clauses which permit creditor to accelerate payment of the principal and interest outstanding even if the firm does not enter liquidation, and that have been included in row 27.
			The number entered in cell C30 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C27.
31	D–G	including an incentive to redeem	This only covers liabilities that contain explicit incentives to redeem the liability ahead of its contractual maturity date and that have been included in row 27.
			For example, explicit incentives to redeem a liability early may consist of step-up clauses where interest payments increase after a number of years.
			Please split the total amount by residual contractual maturity
			and enter the corresponding amounts in columns D to G. The number entered in cell C31 cannot be larger than the
			corresponding number in cell C27.
32	D-G	including an incentive to	In principle, this covers the same liabilities as in row 31.
		redeem (where the effective maturity date is the date of the	However, please split the total amount by the effective residual maturity (which is defined as the time remaining
		incentive to redeem)	until the first step-up clause becomes effective) and enter the
			corresponding amounts in columns D to G. The sum of columns E to G will give only liabilities for which the
			effective residual maturity is at least one year.
33	С	uninsured deposits	This covers any uninsured term deposits that have been included in row 27.
			The number entered in cell C33 cannot be larger than the
			corresponding number in cell C27.
34	D-G	unsecured liabilities excluding (i), regulatory capital, (ii) liabilities arising from derivatives, (iii) liabilities arising	This row captures the amount of all subordinated unsecured liabilities excluding (i), regulatory capital, (ii) liabilities arising from derivatives, (iii) liabilities arising not through contract (Section 12e) and (iv) deposits.
			Liabilities in these cells don't need to meet the TLAC term
		not through contract (Section 12e) and (iv) deposits; of which:	sheet criteria. Please split the total amount by residual contractual maturity and enter the corresponding amounts columns D to G.
			This cell should be reported on a consolidated basis (intra-
			resolution group liabilities should be netted out), including issuance by non-resolution entities.
35	С	C issued by the resolution entity	This only covers liabilities that have been issued by the resolution entity and that have been included in cell C34.
			The number entered in cell C35 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C34.
36	С	currently issued by an	This only covers liabilities that have been issued by a direct
		operating bank immediately below a resolution entity that is a holding company	subsidiary of the resolution entity if the resolution entity is a holding company (or intermediate holding company) and that have been included in row C34.
			The number entered in cell C36 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C34.

Row	Column	Heading	Description
38	C	containing a contractual trigger or being subject to a statutory mechanism which permits the relevant resolution authority to expose TLAC to loss or convert to equity in resolution (Section 17)	This only covers liabilities that can be bailed in (or otherwise exposed to loss) by the relevant resolution authorities of the resolution entity (ie meeting the TLAC term sheet Section 17) and that have been included in row 34. Liabilities can be exposed to loss by the relevant authorities it they are within the scope of statutory resolution tools or contain contractual terms that allow them to be written down and/or converted into equity by the resolution authorities. If liabilities are issued under foreign law they have to (a) contain contractual bail-in clauses, (b) contain contractual terms acknowledging that they are subject to resolution actions by the home authorities, or (c) the law that they are governed by must acknowledge resolution actions by the home authorities on a statutory basis. The number entered in cell C38 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C34.
39	D-G	Case 4 (ii): Total senior unsecured liabilities excluding (i) liabilities arising from derivatives, (ii) liabilities arising not through contract (Section 12e) and (iii) deposits; of which:	This row captures the amount of all senior unsecured liabilities excluding (i) liabilities arising from derivatives, (ii) liabilities arising not through contract (Section 12e) and (iii) deposits. Please split the total amount by residual contractual maturity and enter the corresponding amounts in columns D to G. This cell should be reported on a consolidated basis (intraresolution group liabilities should be netted out), including issuance by non-resolution entities. Do not include any liabilities that have been included in row 34. The only difference between rows 34 and 39 is subordination.
40	С	issued by the resolution entity	This only covers liabilities that have been issued by the resolution entity and that have been included in cell C39. The number entered in cell C40 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C39.
41	C	currently issued by an operating bank immediately below a resolution entity that is a holding company	This only covers liabilities that have been issued by a direct subsidiary of the resolution entity if the resolution entity is a holding company (or intermediate holding company) and that have been included in cell C39. This column intends to capture the impact of senior unsecured liabilities currently not meeting the Term Sheet criteria particularly with regard to subordination but where there is reasonable expectation that the subordination criterion of section 13c would be met, once the liabilities issued by an operating bank immediately below a resolution entity are transferred to the resolution entity. The number entered in cell C41 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C39.

Row	Column	Heading	Description
43	С	Structured notes	Please enter the amount of structured notes. Structured notes should be defined as debt obligations that contain an embedded derivative component, with returns linked to an underlying security or index (public or bespoke, such as equities or bonds, fixed income rates or credit, FX, commodities etc).
			Structured notes do not include debt instruments that include call or put options only, ie the value of the instrument does not depend on any embedded derivative component.
44	С	containing a contractual trigger or being subject to a statutory mechanism which permits the relevant resolution	This only covers liabilities that can be bailed in (or otherwise exposed to loss) by the relevant resolution authorities of the resolution entity (ie meeting the TLAC term sheet Section 17) and that have been included in row 39.
		authority to expose TLAC to loss or convert to equity in resolution (Section 17)	Liabilities can be exposed to loss by the relevant authorities if they are within the scope of statutory resolution tools or contain contractual terms that allow them to be written down and/or converted into equity by the resolution authorities. If liabilities are issued under foreign law they have to:
			(a) contain contractual bail-in clauses;
			(b) contain contractual terms acknowledging that they are subject to resolution actions by the home authorities; o
			(c) the law that they are governed by must acknowledge resolution actions by the home authorities on a statutory basis.
			The number entered in cell C44 only includes liabilities > 1 year residual maturity and cannot be larger than the corresponding number in cell C39.
45	С	(only for MPE groups) Total subordinated liabilities excluding regulatory capital, issued by the resolution entity, meeting the Term Sheet criteria not taking into	For MPE banks at a 'resolution group' level: Total amount of subordinated liabilities (excluding regulatory capital) that meet the Term Sheet criteria except for the criterion of not being funded directly by a related party (Section 12c). These are only liabilities that are held by related entities belonging to other resolution groups.
		consideration the criterion of not being funded directly by a related party (Section 12c), ie such liabilities held by entities in other resolution groups	Leave this row blank unless the template is being filled out for an individual (ie sub-consolidated) resolution group of an MPE bank.
46	С	(only for MPE groups) Total senior liabilities excluding regulatory capital, issued by the resolution entity, meeting the Term Sheet criteria not	For MPE banks at a 'resolution group' level: Total amount of senior liabilities (excluding regulatory capital) that meet the Term Sheet criteria except for the subordination criterion (Section 13) and the criterion of not being funded directly by a related party (Section 12c).
		taking into consideration the subordination criterion (Section 13) and the criterion	These are only liabilities that are held by related entities belonging to other resolution groups. Do not include any liabilities that have been included in row
		of not being funded directly by a related party (Section 12c), ie	44. Leave this row blank unless the template is being filled out
		such liabilities held by entities in other resolution groups	for an individual (ie sub-consolidated) resolution group of an MPE bank.

Row	Column	Heading	Description
19–39	Н	Average residual maturity (years)	Please enter the weighted average residual maturity of all instruments in a given row.
			You should use the average contractual residual maturity in relevant rows except for rows 24, 26 and 32. Please use the average effective residual maturity In rows 24, 26 and 32, as defined above.
			The weighted average residual maturity should be calculated by the following formula.
			$\sum \frac{Residual\ maturity * Amount\ with the\ residual\ maturity}{Total\ amount}$
			Totalamount
34, 39	I	Of which not governed by home law and with no	Please enter all instruments that are issued under third country law and that do not
		equivalent binding statutory or	(a) contain contractual bail-in clauses,
		contractual provision for cross- border recognition of resolution actions and ≥ 1yr	(b) contain contractual terms acknowledging that they are subject to resolution actions by the home authorities, and
		(Section 16)	(c) where the law that they are governed by does not acknowledge resolution actions by the home authorities on a statutory basis.
			The number entered here is a sub-set of, and should not therefore be greater than, the number in 34C and 39C respectively.
19, 26, 34	J	Of which contractually subordinated and ≥ 1yr (Section 13 a)	Please enter all instruments that meets contractual subordination criterion (Section 13a) with a residual contractual maturity of more than or equal to one year.
			The number entered here is a sub-set of, the number in cells C19, C26 and C34 respectively. The sum of the numbers in cells J and K should be exactly equal to the number in cell C.
19, 26, 34	K	Of which structurally subordinated and ≥ 1yr (Section 13 c)	Please enter all instruments that meets structural subordination criterion (Section 13c) with a residual contractual maturity of more than or equal to one year.
			The cells in this column should be zero unless the resolution entity is a holding company which does not have excluded liabilities on its balance sheet. Instruments that meet the structural subordination requirements should only be included in this column and not in column J even if they are in the form of subordinated debt (issued by a holding company).
			The number entered here is a sub-set of, the number in cells C19, C26 and C34 respectively. The sum of the numbers in cells J and K should be exactly equal to the number in cell C.
C)	Deductions		
51	С	Holdings of Common Equity Tier 1 net of short positions	Enter the whole amount of holdings of Common Equity Tier 1 of banking, financial and insurance entities excluding entities that are both regulatory consolidated and included in own resolution group where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
			This number should be net of short positions, ie deduct permitted offsetting short positions.

Row	Column	Heading	Description
52	С	Holdings of Additional Tier 1 capital net of short positions	Enter the whole amount of holdings of additional Tier 1 capital of banking, financial and insurance entities excluding entities that are both regulatory consolidated and included ir own resolution group and where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
			This number should be net of short positions, ie deduct permitted offsetting short positions.
53	С	Holdings of Tier 2 capital net of short positions	Enter the whole amount of holdings of Tier 2 capital of banking, financial and insurance entities excluding entities that are both regulatory consolidated and included in own resolution group and where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). This number should be net of short positions, ie deduct
			permitted offsetting short positions.
57	С	Gross holdings of unsecured liabilities in Case 1 issued by other G-SIBs (or subsidiaries of G-SIBs) not taking into consideration instruments counting towards TLAC by way of the last two paragraphs of section 13 of the term sheet (same as row 19)	Enter the whole amount of unsecured liabilities in case1 (see definition of row 19), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). There is no need to include instruments that might count towards TLAC by way of the last two paragraphs of section 13 in the term sheet.
57	D	Gross holdings of unsecured liabilities in Case 2 issued by other G-SIBs (or subsidiaries of G-SIBs) not taking into consideration instruments counting towards TLAC by way of the last two paragraphs of Section 13 of the term sheet (same as row 26)	Enter the whole amount of unsecured liabilities in case2 (see definition of row 26), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). There is no need to include instruments that might count towards TLAC by way of the last two paragraphs of section 13 in the term sheet.
57	E	Gross holdings of unsecured liabilities in Case 3 issued by other G-SIBs (or subsidiaries of G-SIBs) (same as rows 27)	Enter the whole amount of unsecured liabilities in case3 (see definition of row 27), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
57	F	Gross holdings of unsecured liabilities in Case 4 issued by other G-SIBs (or subsidiaries of G-SIBs) (same as row 34 and 39)	Enter the whole amount of unsecured liabilities in case4 (see definition of row 34 and 39), excluding regulatory capital, issued by other G-SIBs/other resolution entities, where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
58	C–F	Permitted offsetting short positions in relation to the specific gross holdings included above	Enter the amount of permitted offsetting short position of unsecured liabilities included in row 57 in each case. Please use the same criteria as for regulatory capital holdings.
60	C–F	Risk weighted assets related to above, included in total risk- weighted assets	Enter risk weighted assets that are related to row 59 and that are included in total risk-weighted assets.

Row	Column	Heading	Description
62	C-F	Common Equity Tier 1 capital after all regulatory adjustments that do not depend on a threshold	[For SPE banks and MPE banks at a group consolidated level The cell refers to DefCap worksheet cell D52. Please leave as it is. [For MPE banks at a 'resolution group' level] Enter the amount of Common Equity Tier 1 capital after all regulatory adjustments that do not depend on a threshold. Please refer to DefCap worksheet cell D52 for the calculation.
85	C-F	Holdings of Common Equity Tier 1 net of short positions (ie risk weighted assets of exposures in row 75)	Enter risk weighted assets that are related to row 75 (ie amounts not deducted).
86	C-F	Holdings of Additional Tier 1 capital net of short positions (ie risk weighted assets of exposures in row 76)	Enter risk weighted assets that are related to row 76 (ie amounts not deducted).
87	C-F	Holdings of Tier 2 capital net of short positions (ie risk weighted assets of exposures in row 77)	Enter risk weighted assets that are related to row 77 (ie amounts not deducted).
88	C-F	Holdings of unsecured liabilities net of short positions (ie risk weighted assets of exposures in row 78)	Enter risk weighted assets that are related to row 78 (ie amounts not deducted).
90	C-F	Holdings of Common Equity Tier 1 net of short positions (ie risk weighted assets of exposures in row 80)	Enter risk weighted assets that are related to row 80 (ie amounts not deducted).
91	C-F	Holdings of Additional Tier 1 capital net of short positions (ie risk weighted assets of exposures in row 81)	Enter risk weighted assets that are related to row 81 (ie amounts not deducted).
92	C–F	Holdings of Tier 2 capital and unsecured liabilities net of short positions (ie risk weighted assets of exposures in row 82)	Enter risk weighted assets that are related to row 82 (ie amounts not deducted).
96	С	Gross holdings of unsecured liabilities in Case 1 issued by other G-SIBs (or subsidiaries of G-SIBs) not taking into consideration instruments qualifying as TLAC by way of the last two paragraphs of section 13 of the term sheet	Enter the whole amount of unsecured liabilities in case1 (see definition of row 19), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank owns more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). There is no need to include instruments that might count towards TLAC by way of the last two paragraphs of section
96	D	Gross holdings of unsecured liabilities in Case 2 issued by other G-SIBs (or subsidiaries of G-SIBs) not taking into consideration instruments qualifying as TLAC by way of the last two paragraphs of section 13 of the term sheet	13 in the term sheet. Enter the whole amount of unsecured liabilities in case2 (see definition of row 26), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank owns more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). There is no need to include instruments that might count towards TLAC by way of the last two paragraphs of section 13 in the term sheet.

Row	Column	Heading	Description
96	E	Gross holdings of unsecured liabilities in Case 3	Enter the whole amount of unsecured liabilities in case3 (see definition of row 27), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank owns more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
96	F	Gross holdings of unsecured liabilities in Case 4	Enter the whole amount of unsecured liabilities in case4 (see definition of row 34 and 39), excluding regulatory capital, issued by other G-SIBs/other resolution entities where the bank owns more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
97	C–F	Permitted offsetting short positions in relation to the specific gross holdings included above	Enter the amount of permitted offsetting short position of TLAC eligible liabilities included in row 96 in each case. Please use the same criteria as for regulatory capital holdings.
99	C–F	Risk weighted assets related to above, included in total risk- weighted assets	Enter risk weighted assets that are related to row 98 and that are included in total risk-weighted assets.
D)	Deposit bre	akdown for EU banks (EU banks	only, see BRRD Article 108)
106	C-G	Senior deposits; of which:	Please include deposits from all classes of depositor. Demand deposits should be included under column D (<1 year residual maturity). The total should equal the sum of rows 107 to 110.
107	C-G	Covered by a deposit guarantee scheme under Directive 2014/49/EU	Please enter deposits covered by any EU or EEA deposit guarantee scheme officially recognised under either the 2014 or 1994 Deposit Guarantee Schemes Directive. Include only deposits below €100,000 (or local currency equivalent).
108	C-G	That part of deposits eligible for coverage but in excess of the coverage limit	Please enter the amount of deposits from eligible depositors (natural persons, or micro-, small-, and medium-sized enterprise) in excess of €100,000 (or local currency equivalent).
109	C-G	Would be eligible for coverage were they not made through branches outside the Union	Please enter the whole amount of deposits in branches outside of the EU or EEA from depositors who would be eligible if depositing in the EU/EEA (natural persons, or micro-, small-, and medium-sized enterprise). Include the entire amount of such deposits.
110	C–G	Not eligible for coverage	Please enter the full amount of any other deposits
106–110	Н	Average residual maturity (years)	Average maturity only of time deposits with residual maturity of over one year.
106–110	I	Non-maturity deposits	Non-maturity deposits (NMDs) are generally defined as liabilities of banks in which the claimant (ie depositor) is free to withdraw anytime since they have no contractually agreed maturity date

9.4 Internal TLAC

The "G-SIB TLAC Internal" worksheet collects and sums up data on internal TLAC of material subsidiaries for a G-SIB (both SPE and MPE). One column (columns D to AB) should be completed for each material subsidiary, ie an entity incorporated in a national jurisdiction other than that in which the resolution entity is incorporated and meeting at least one of the criteria in section 21 of the term sheet:

- (a) has more than 5% of the consolidated risk-weighted assets of the G-SIB group; or
- (b) generates more than 5% of the consolidated revenues of the G-SIB group; or
- (c) has a total leverage exposure measure larger than 5% of the G-SIB group's total leverage ratio measure; or
- (d) has been identified by the firm's CMG as material to the exercise of the firm's critical functions.

If the CMG has not yet determined the scope of the material subsidiaries, then for the purpose of the QIS, the material subsidiaries should be based on the quantitative thresholds listed in a-c above.

The worksheet is designed to test three different "cases" using a menu approach to test the impact of specific features of internal TLAC instruments as follows:

- **Case 1** includes all non-regulatory capital instruments that meet all of the criteria on the proposed TLAC term sheet.
- Case 2 analyses the subordination requirement. This case includes instruments that meet the
 TLAC term sheet criteria except for subordination. The purpose is to analyse the impact of
 requiring TLAC instruments to be subordinated and how this impacts on the shortfalls of
 internal TLAC observed in G-SIBs. Note this is the same case as case 3 in the G-SIB TLAC
 external worksheet.
- **Case 3** is the widest case and includes all unsecured liabilities excluding regulatory capital, liabilities arising through derivatives, liabilities not arising through contract and deposits. Cases 3i and 3ii differ only by the subordination requirement. The purpose of case 3 is to gain insights into liabilities that could potentially, once matured, be replaced by liabilities that do qualify for TLAC. Note this is the same case as case 4 in the G-SIB TLAC external worksheet.

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description	
A)	Regulatory capital			
5	D-AB	Resolution group number to which the material subsidiary belongs (only for MPE)	Please enter the resolution group number to which the material subsidiary belongs. This is only relevant for MPE. To find out the relevant number please refer to your national supervisor.	
8	D-AB	Total Common Equity Tier 1 capital of the subsidiary (after regulatory adjustments): of which	Enter the amount of net Common Equity Tier 1 capital issued by the material subsidiary according to national rules in 2022 (fully phased in). Please refer to Section 20 of the term sheet to fill out this cell, eg for CET1 instruments issued externally by the subsidiary, please only count towards TLAC to the extent that they count towards the resolution group's consolidated CET 1 position.	
9	D-AB	owned directly by the resolution entity (before regulatory adjustments) ie no intermediate entities	Enter the nominal amount of Total Common Equity Tier 1 capital that is owned directly by the resolution entity and that is included in row 8. Please note that the value in row 9 could be higher than the amount in row 8 because row 8 is after regulatory adjustments.	
10	D-AB	Issued indirectly to the resolution entity via intermediate entities (before regulatory adjustments). Note that if the intermediate entity is itself another material subsidiary then it should not populate the amount in rows 9 and 10 to avoid double counting of internal TLAC	Enter the nominal amount of Total Common Equity Tier 1 capital that is owned indirectly by the resolution entity (ie via intermediate entities) and that is included in row 8. Please note that the value in row 10 could be higher than the amount in row 8 because row 8 is after regulatory adjustments.	

Row	Column	Heading	Description
11	D-AB	issued to external investors but counting towards the resolution group's consolidated capital requirements (as per Section 20)	Enter the nominal amount of Total Common Equity Tier 1 capital that is issued to external investors and eligible for inclusion in the resolution group's regulatory capital according to paragraphs 62 to 64 of Basel III (ie meeting TLAC term sheet Section 20) and that is included in row 8.
12	D-AB	issued, via subsidiary of the material subsidiary, to external investors but counting towards the resolution group's consolidated capital requirements (as per Section 20)	Enter the nominal amount of Total Common Equity Tier 1 capital that is issued, via subsidiary of the material subsidiary, to external investors and eligible for inclusion in the resolution group's regulatory capital according to paragraphs 62 to 64 of Basel III (ie meeting TLAC term sheet Section 20) and that is included in row 8.
13	D-AB	Total Additional Tier 1 of the subsidiary (after regulatory adjustments); of which	Enter the amount of net Additional Tier 1 instruments issued by the material subsidiary according to national rules in 2022 (fully phased in). Please refer to Section 20 of the term sheet to fill out this cell, eg for Additional Tier 1 instruments issued externally by the subsidiary, please only count towards TLAC to the extent that they count towards the resolution group's consolidated Tier 1 position.
14	D-AB	owned directly by resolution entity (before regulatory adjustments)	Enter the nominal amount of Additional Tier 1 instruments that is owned directly by the resolution entity and that is included in row 13. Please note that the value in row 14 could be higher than the amount in row 13 because row 13 is after regulatory adjustments.
15	D-AB	Issued indirectly to the resolution entity via intermediate entities (before regulatory adjustments). Note that if the intermediate entity is itself another material subsidiary then it should not populate the amount in rows 14 and 15 to avoid double counting of internal TLAC	Enter the nominal amount of Additional Tier 1 capital that is owned indirectly by the resolution entity (ie via intermediate entities) and that is included in row 13. Please note that the value in row 15 could be higher than the amount in row 13 because row 13 is after regulatory adjustments.
16	D-AB	issued to external investors but counting towards the resolution group's consolidated capital requirements (as per Section 20)	Enter the nominal amount of Additional Tier 1 instruments that is issued to external investors and eligible for inclusion in the resolution group's regulatory capital according to para 62 to 64 of Basel III (ie meeting TLAC term sheet Section 20) and that is included in row 13.
17	D-AB	issued, via subsidiary of the material subsidiary, to external investors but counting towards the resolution group's consolidated capital requirements (as per Section 20)	Enter the nominal amount of Additional Tier 1 instruments that is issued, via subsidiary of the material subsidiary, to external investors and eligible for inclusion in the resolution group's regulatory capital according to para. 62 to 64 of Basel III (ie meeting TLAC term sheet Section 20) and that is included in row 13.
18	D-AB	issued under third country law without any contractual bail-in clauses and no equivalent binding statutory provision for cross-border recognition of resolution actions (before regulatory adjustments)	Enter the nominal amount of Additional Tier 1 instruments that is issued under the third country law without any contractual bail-in clauses and no equivalent binding statutory provision for cross-border recognition of resolution actions and that is included in row 13.

Row	Column	Heading	Description
19	D-AB	Total Tier 2 of the subsidiary (after regulatory adjustments); of which	Enter the amount of net Tier 2 instruments issued by the material subsidiary recognised in its resolution entity's consolidated regulatory capital according to national rules in 2022 (fully phased in). Please refer to Section 20 of the term sheet to fill out this cell. For more detail, (a) For Tier 2 instruments issued externally by the subsidiary, please only count towards TLAC to the extent that they count towards the resolution group's consolidated Tier 2 position. (b) For Tier 2 issued to the resolution entity, please include the full nominal amount with a residual maturity of > 1 yr. To the extent that part of the instruments no longer qualifies as Tier 2 due to amortisations it still qualifies a "eligible liabilities". (c) Please include the non-amortised fraction of Tier 2 instruments with a residual maturity of less than one year if any.
20	D-AB	owned directly by resolution entity (before regulatory adjustments and prior to amortisation)	Enter the nominal amount of Tier 2 instruments that is owned directly by the resolution entity and that is included in row 19. Please note that the value in row 20 could be higher than the amount in row 19 because row 19 is after regulatory adjustments.
21	D-AB	issued indirectly to the resolution entity via intermediate entities (before regulatory adjustments). Note that if the intermediate entity is itself another material subsidiary then it should not populate the amount in rows 20 and 21 to avoid double counting of internal TLAC	Enter the nominal amount of Tier 2 capital that is owned indirectly by the resolution entity (ie via intermediate entities) and that is included in row 19. Please note that the value in row 21 could be higher than the amount in row 19 because row 19 is after regulatory adjustments.
22	D-AB	issued to external investors but counting towards the resolution group's consolidated capital requirements (as per Section 20) (before regulatory adjustments but after amortisation)	Enter the amount of Tier 2 instruments (before regulatory adjustments but after amortisation) that is issued to external investors and eligible for inclusion in the resolution group's regulatory capital according to para. 62-64 of Basel III (ie meeting TLAC term sheet Section 20) and that is included in row 19.
23	D-AB	issued, via subsidiary of the material subsidiary, to external investors but counting towards the resolution group's consolidated capital requirements (as per Section 20)	Enter the amount of Tier 2 instruments (before regulatory adjustments but after amortisation) that is issued, via subsidiary of the material subsidiary, to external investors and eligible for inclusion in the resolution group's regulator capital according to para. 62-64 of Basel III (ie meeting TLAC term sheet Section 20) and that is included in row 19.
24	D-AB	issued under third country law without any contractual bail-in clause and no equivalent binding statutory provision for cross-border recognition of resolution actions (before regulatory adjustments but after amortisation)	Enter the amount of Tier 2 instruments (before regulatory deductions but after amortisation) that is issued under the third country law without any contractual bail-in clauses and no equivalent binding statutory provision for cross-border recognition of resolution actions and that is included in row 19.

Row	Column	Heading	Description
B)	Unsecured	liabilities excluding regulatory ca	pital
28	D-AB	Case 1: Total liabilities excluding regulatory capital, issued directly or indirectly to the resolution entity, meeting the Term Sheet criteria	Enter the total amount of liabilities excluding regulatory capital, issued by the material subsidiary directly or indirectly to the resolution entity, meeting the Term Sheet criteria. Include only instruments that meet all of the requirements in the term sheet. However, please do not take into consideration instruments that only qualify by means of the last two paragraphs of section 13.
30	D-AB	Case 2: Total liabilities excluding regulatory capital issued directly or indirectly to the resolution entity, meeting the Term Sheet criteria not taking into consideration the subordination criterion (Section 13a, b, c)	Enter the total amount of liabilities excluding regulatory capital issued directly or indirectly to the resolution entity, meeting the Term Sheet criteria except for the subordination criterion (Section 13a, b, c). This case also includes instruments meeting case 1 and should, in the majority of cases, be a larger number than the number in row 28.
32	D-AB	Case 3(i): Total subordinated unsecured liabilities excluding regulatory capital, liabilities arising through derivatives, liabilities not arising through contact and deposits	Enter the total amount of subordinated unsecured liabilities issued by the material subsidiary, not eligible to regulatory capital and deposits
33	D-AB	of which held by the resolution entity	Enter the amount of liabilities that is held by the resolution entity and that is included in row 32.
35	D-AB	Case 3(ii): Total senior unsecured liabilities excluding regulatory capital, liabilities arising through derivatives, liabilities not arising through contract and deposits	Enter the total amount of senior unsecured liabilities issued by the material subsidiary, not eligible to regulatory capital. Do not include any liabilities that have been included in row 32.
36	D-AB	of which held by the resolution entity	Enter the amount of liabilities that is held by the resolution entity and that is included in row 35.
38	D-AB	Deposits	Enter the amount of deposits of the material subsidiary.
C)	Collateralis	ed guarantees	
41	D-AB	Collateralised guarantees provided by the resolution entity to the material subsidiary (meeting conditions in the term sheet Section 23)	Enter the total amount of guarantees that are provided to the material subsidiary by the resolution entity that are collateralised and in addition meet the conditions in section 23 of the TLAC term sheet.
D)	Denominat	or	
44	D-AB	Total risk-weighted assets of the material subsidiary (sub- consolidated basis)	Enter the total amount of risk-weighted assets of the reporting subsidiary on a sub-consolidated basis.
45	D-AB	Basel leverage exposure measure of the material subsidiary (sub-consolidated basis)	Enter the exposure measure (the denominator) according to the Basel III leverage ratio framework (published in January 2014) for the reporting subsidiary on a sub-consolidated basis.
46	D-AB	Total risk-weighted assets of the material subsidiary (standalone)	Enter the total amount of risk-weighted assets of the reporting subsidiary on a standalone basis.
47	D-AB	Basel leverage exposure measure of the material subsidiary (stand-alone)	Enter the exposure measure (the denominator) according to the Basel III leverage ratio framework (published in January 2014) for the reporting subsidiary on a standalone basis.

9.5 Holdings of TLAC (non-G-SIBs)

The "TLAC holdings" worksheet collects data on non-G-SIB's holdings of TLAC **on a consolidated basis**. Data on regulatory capital should be filled out according to national rules fully phased in as in 2022. The worksheet is designed to test three different "cases" using a menu approach to test the impact of specific features of TLAC instruments as follows:

- **Case 1** includes all non-regulatory capital instruments that meet all of the criteria on the proposed TLAC term sheet.
- **Case 2** analyses the subordination requirement. This case includes instruments that meet all the TLAC term sheet criteria (excluding regulatory capital) **plus** instruments that meet all the TLAC criteria except for subordination. The purpose is to analyse the impact of requiring TLAC instruments to be subordinated (by comparing to case 1).
- **Case 3** is the widest case and includes all unsecured G-SIB liabilities (excluding regulatory capital), not arising from derivatives or through contract or deposits. The purpose of case 3 is to gain insights into liabilities that could potentially, once matured, be replaced by liabilities that do qualify for TLAC.

The following table provides a description of the data to be entered in each row.

Row	Column	Heading	Description
A)	Regulatory	capital	
5	С	Total Common Equity Tier 1 capital after all regulatory adjustments	Automatic calculation (refer to DefCap D64).
6	С	Total Additional Tier 1 instruments after all regulatory adjustments	Automatic calculation (refer to DefCap D82).
7	С	Total Tier 2 instruments after all regulatory adjustments	Automatic calculation (refer to DefCap D101).
B)	Holding of	TLAC instruments	
12	С	Holdings of Common Equity Tier 1 net of short positions	Enter the whole amount of holdings of Common Equity Tier 1 of banking, financial and insurance entities excluding entities that are regulatory consolidated where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). This number should be net of short positions, ie deduct permitted offsetting short positions.
13	С	Holdings of Additional Tier 1 capital net of short positions	Enter the whole amount of holdings of Additional Tier 1 capital of banking, financial and insurance entities excluding entities that are regulatory consolidated and where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). This number should be net of short positions, ie deduct permitted offsetting short positions.
14	С	Holdings of Tier 2 capital net of short positions	Enter the whole amount of holdings of Tier 2 capital of banking, financial and insurance entities excluding entities that are regulatory consolidated and where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). This number should be net of short positions, ie deduct permitted offsetting short positions.

Row	Column	Heading	Description
18	С	Gross holdings of unsecured liabilities excluding regulatory capital, issued by G-SIBs in Case 1.	Enter the whole amount of unsecured liabilities issued by G-SIBs meeting the TLAC term sheet criteria, excluding regulatory capital, issued by G-SIBs/resolution entities where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). Please do not include any senior instruments that might qualify for TLAC by the issuing bank by way of the last two
			paragraphs of section 13 of the TLAC term sheet.
18	D	Gross holdings of unsecured liabilities excluding regulatory capital, issued by G-SIBs in Case 2.	Enter the whole amount of unsecured liabilities meeting the TLAC term sheet criteria except for Section 13 (subordination), excluding regulatory capital, issued by G-SIBs/resolution entities where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). Please note that case 2 also includes case 1 and hence the number entered in D18 will therefore, in the majority of cases, be larger than the number in C18.
18	E	Gross holdings of unsecured liabilities excluding regulatory capital, issued by G-SIBs in Case 3.	Enter the whole amount of unsecured liabilities, excluding regulatory capital, those arising from derivatives and those arising not through contract (section 12e), where the bank does not own more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). Please note that case 3 also includes case 1 and hence the number entered in E18 will therefore, in the majority of case be larger than the number in C18.
19	C–E	Permitted offsetting short positions in relation to the specific gross holdings	Enter the amount of permitted offsetting short position of TLAC eligible liabilities included in row 18 in each case. Please use the same criteria as for regulatory capital
		included above	holdings.
32	C–E	Holdings of Common Equity Tier 1 net of short positions (ie risk weighted assets of exposures in row 28)	Enter risk weighted assets that are related to row 28 (ie amounts not deducted).
33	C–E	Holdings of Additional Tier 1 capital net of short positions (ie risk weighted assets of exposures in row 29)	Enter risk weighted assets that are related to row 29 (ie amounts not deducted).
34	C–E	Holdings of Tier 2 capital and TLAC eligible liabilities net of short positions (ie risk weighted assets of exposures in row 30)	Enter risk weighted assets that are related to row 30 (ie amounts not deducted).
38	С	Gross holdings of unsecured liabilities excluding regulatory capital, issued by G-SIBs in Case 1	Enter the whole amount of unsecured liabilities meeting the TLAC term sheet criteria, excluding regulatory capital, issued by G-SIBs/resolution entities where the bank owns more tha 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less). Please do not include any senior instruments that might qualify for TLAC by the issuing bank by way of the last two paragraphs of section 13 of the TLAC term sheet.

Row	Column	Heading	Description
38	D	Gross holdings of unsecured liabilities excluding regulatory capital, issued by G-SIBs in Case 2	Enter the whole amount of unsecured liabilities issued by G-SIBs meeting the TLAC term sheet criteria except for Section 13 (subordination), excluding regulatory capital, issued by G-SIBs/resolution entities where the bank owns more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
38	E	Gross holdings of unsecured liabilities excluding regulatory capital, issued by G-SIBs Case 3	Enter the whole amount of unsecured liabilities, excluding regulatory capital, those arising from derivatives and those arising not through contract (section 12e), where the bank owns more than 10% of the issued common share capital (excluding amount held for underwriting purposes only if held for five working days or less).
39	C–E	Permitted offsetting short positions in relation to the specific gross holdings included above	Enter the amount of permitted offsetting short position of TLAC eligible liabilities included in row 38 in each case. Please use the same criteria as for regulatory capital holdings.
45	C–E	Risk weighted assets related to the unsecured liabilities in each case, excluding regulatory capital issued by G-SIBs, included in row 44	Enter risk weighted assets in each case that are related to rows 20 and 40 and that are included in row 44.
50	C–E	The number of G-SIB groups represented in the exposure above	Enter the number of G-SIB groups that are represented in the sum of the TLAC exposures entered in row 49 in each case.

9.6 External TLAC (location)

The "External TLAC (location)" worksheet collects data on external TLAC of G-SIB's resolution entities and its location of issuance. The worksheet is designed to analyse amounts and pricing of unsecured liabilities issued in different markets. Please read the general instructions in panel B paragraph 9.3 carefully before filling out this part of the worksheet.

Each resolution group will fill out this worksheet on a consolidated (or sub-consolidated) basis unless specified otherwise. Please note that the term sheet allows for G-SIBs from jurisdictions with certain resolution regimes to permit liabilities that rank alongside excluded liabilities to contribute to TLAC. This is explained in the final two paragraphs of Section 13 of the term sheet. For the purposes of the TLAC QIS, G-SIBs should *not* take these into consideration when populating the template.

For each item it is required to report with reference as of end-2014;

- (a) Total nominal amount outstanding;
- (b) Weighted average residual yield to maturity (as a percentage, using the nominal amount outstanding as weights). 62 The yield should be calculated to the contractual maturity date. If possible, the calculations should include the impact of hedges;
- (c) Only for the instruments that meet all of the criteria on the proposed TLAC term sheet (see case 1 below), the maximum residual yield to maturity. The yield should be calculated to the contractual maturity date. If possible, the calculations should include the impact of hedges;

144

The residual yield to maturity is the return anticipated on a bond if held from the reference date (ie end-2014) until the end of its lifetime.

(d) Weighted average z-spread in basis points, using the nominal amount outstanding as weights. ⁶³ Please provide best estimates of the market level for instruments that do not have an observable market quote available. The relevant swap rates should be used as a benchmark curve for the calculation of the spreads. If possible, please provide the spreads in local terms, ie relative to each local swaps curve depending on the currency of denomination.

The following table provides a description of the data to be entered in each row. The references to Term Sheet criteria include the national specificities, when applicable.

The worksheet is designed to test similar "cases" to those described in Section 9.3 above. Please note that not all cases are identical due to the different nature of this part of the analysis to understand the cost impact of the TLAC proposals.

- **Case 1** includes all non-regulatory capital instruments that meet all of the criteria on the proposed TLAC term sheet. The definition is consistent with Case 1 in paragraph 9.3.
- **Case 2(i)** includes all non-regulatory capital subordinated instruments that meet all of the criteria on the proposed TLAC term sheet but are issued by subsidiaries different from the resolution entity. The purpose of gathering this data is to analyse the cost of migrating the issuance of debt securities from the subsidiaries to the resolution entity.
- **Case 2(ii)** includes all non-regulatory capital senior instruments that meet all of the criteria on the proposed TLAC term sheet but are issued by subsidiaries different from the resolution entity. The purpose of gathering this data is to analyse the cost of migrating the issuance of debt securities from the subsidiaries to the resolution entity.
- Case 3 analyses the subordination requirement. It includes all senior unsecured liabilities that
 meet all the proposed TLAC term sheet criteria except for the subordination requirement. The
 purpose of Case 3 is to analyse the impact of requiring TLAC instruments to be subordinated
 (same as Case 3 in paragraph 9.3 minus subordinated liabilities).
- Case 4(i) includes all subordinated unsecured liabilities issued by the resolution entity not considered under Case 1, excluding instruments arising from derivatives (eg structured notes as in Case 5), not through contracts and deposits. The purpose of Case 4(i) is to calculate the total amount of subordinated liabilities and possible differences in pricing compared with Case 1 above. The definition is consistent with Case 4(i) minus Case 1 in paragraph 9.3, excluding issuance by entities different from the resolution entity.
- Case 4(ii) includes all non-subordinated senior unsecured liabilities issued by the resolution entity not considered under Case 3, excluding instruments arising from derivatives (eg structured notes as in Case 5), not through contract and deposits. Adding Cases 2 to 4(ii) would provide the total amount of subordinated non TLAC-eligible and senior unsecured liabilities with the respective pricing to calculate differences with TLAC-eligible instruments in Case 1 above, gaining insights into liabilities that could potentially be replaced by liabilities that do qualify for TLAC. The definition is consistent with Case 4(ii) minus structured notes minus Case 3 in paragraph 9.3, excluding issuance by entities different from the resolution entity.
- **Case 5** includes structured notes issued by the resolution entity, with the purpose of assessing the outstanding amount and pricing of those instruments. The definition is consistent with row 43 in Section 9.3 excluding issuance by entities different from the resolution entity.

Instructions for Basel III monitoring

The z-spread is the basis point spread that would need to be added to the implied spot yield curve such that the discounted cash flows of a bond are equal to its present value (its current market price).

Case	Column	Heading	Description
1	G–R	Total subordinated liabilities excluding regulatory capital, issued by the resolution entity, meeting the Term Sheet criteria	Subordinated unsecured liabilities which would not be eligible to regulatory capital and would meet all of the criteria in the term sheet (except instruments qualifying by way of the final two paragraphs of section 13).
2(i)	G–R	Total subordinated liabilities excluding regulatory capital, issued by subsidiaries different from the resolution entity, meeting the Term Sheet criteria not taking into consideration the issuer criteria (Section 9)	Subordinated unsecured liabilities different from regulatory capital which would meet all the Term Sheet criteria excluding the issuance by resolution entity.
2(ii)	G–R	Total senior liabilities excluding regulatory capital, issued by subsidiaries different from the resolution entity, meeting the Term Sheet criteria not taking into consideration the issuer criteria (Section 9)	Senior unsecured liabilities (not subordinated) different from regulatory capital which would meet all the Term Sheet criteria excluding the issuance by resolution entity.
3	G-R	Total senior liabilities excluding regulatory capital, issued by the resolution entity, meeting the Term Sheet criteria not taking intoconsideration the subordination criteria (Section 13a, 13b and 13c)	Senior unsecured liabilities (not subordinated) excluding (i) regulatory capital (ii) liabilities arising from derivatives (iii) liabilities arising not through contract (Section 12e) and (iv) deposits.
4(i)	G-R	Other subordinated liabilities issued by the resolution entity, excluding (i) regulatory capital, (ii) liabilities arising not through contract (Section 12e), (iii) deposits and (iv) all liabilities reported in Case 1 and arising from derivatives	Subordinated unsecured liabilities excluding (i), regulatory capital, (ii) liabilities arising not through contract (Section 12e), (iii) deposits and (iv) TLAC-eligible liabilities (Case 1) or liabilities arising from derivatives (eg structured notes in Cas 5). Liabilities in these cells do not meet the TLAC term sheet criteria, except that they have to be issued by the resolution entity (Section 9) and with a remaining maturity of at least one year (Section 11).
4(ii)	G-R	Other senior liabilities issued by the resolution entity, excluding (i) regulatory capital, (ii) liabilities arising not through contract (Section 12e), (iii) deposits and (iv) all liabilities reported in Case 3 and arising from derivatives	Senior unsecured liabilities excluding (i), regulatory capital, (ii) liabilities arising not through contract (Section 12e), (iii) deposits and (iv) liabilities included in Case 1 or liabilities arising from derivatives (eg structured notes in Case 5). Liabilities in these cells do not meet the TLAC term sheet criteria, except that they have to be issued by the resolution entity (Section 9) and with a remaining maturity of at least one year (Section 11).
5	G-R	Structured notes issued by the resolution entity	Structured notes should be defined as debt obligations that contain an embedded derivative component, with returns linked to an underlying security or index (public or bespoke, such as equities or bonds, fixed income rates or credit, FX, commodities etc). Structured notes does not include debt instruments that include call or put options only, ie the value of the instrument does not depend on any embedded derivative

Case	Column	Heading	Description
1b	G–R	Unsecured liabilities with remaining maturity less than 1 year	Unsecured liabilities meeting the criteria as in Case 1 except for the minimum remaining maturity.
2(i)b			Unsecured liabilities meeting the criteria as in Case 2(i) except for the minimum remaining maturity.
2(ii)b			Unsecured liabilities meeting the criteria as in Case 2(ii) except for the minimum remaining maturity.
3b			Unsecured liabilities meeting the criteria as in Case 3 except for the minimum remaining maturity.
4(i)b			Unsecured liabilities meeting the criteria as in Case 4(i) except for the minimum remaining maturity.
4(ii)b			Unsecured liabilities meeting the criteria as in Case 4(ii) except for the minimum remaining maturity.
5b			Unsecured liabilities meeting the criteria as in Case 5 except for the minimum remaining maturity.
1–5, 1b–5b	G–N	Location of issuance	Jurisdiction where the unsecured liabilities were issued.
1–5, 1b–5b	I	of which EU home country	To be compiled only by G-SIBs whose home country is in a EU jurisdiction different from the UK; eg for a German G-SIB, the amount of liabilities issued in the EU countries (including Germany and excluding UK) should be reported in column H and the amount of liabilities issued in Germany in column I.
1–5, 1b–5b	М	Location of resolution entity, if different from US, EU, JP, or HK (please specify)	To be completed only by G-SIBs with resolution entities in a jurisdiction different from the US, EU (excluding UK), UK, JP and HK. Please specify the relevant jurisdiction.
1–5, 1b–5b	0	Average original maturity	Average based on the contractual maturity date. In case of step-up clause or explicit financial incentive to redeem, the date when the first step-up can be exercised should be considered. The average should be weighted by the outstanding amount of the liabilities as calculated for columns G to N. Perpetual maturities should be treated as 10 years.
2–5, 2b–5b	Р	Of which not governed by home country law and with no equivalent binding statutory or contractual provision for crossborder recognition of resolution actions and ≥ 1yr (section 16)	All instruments issued under third country law and that (i) do not contain contractual bail-in clauses (ii) do not contain contractual terms acknowledging that they are subject to resolution actions by the home authorities, and (iii) are not governed by a law that acknowledge resolution actions by the home authorities on a statutory basis.
1, 2(i), 4(i) and (ii), 1b, 2(i)b, 4b(i) and (ii)	Q	Of which contractually subordinated and ≥ 1yr (Section 13a)	All instruments that meet the contractual subordination criterion (Section 13a) with a residual contractual maturity of more than or equal to one year.
1, 4(i) and (ii), 1b, 4b(i) and (ii)	R	Of which structurally subordinated and ≥ 1yr (Section 13c)	All instruments that meet the structural subordination criterion (Section 13c) with a residual contractual maturity of more than or equal to one year.

Annex 1: Changes compared to versions 2.8.x of the reporting template

Compared to the versions 2.8.x of the reporting template which were used for reporting of data as of 30 June 2014, the following main changes have been implemented:

- The "General Info", "Requirements" and "DefCap" worksheets have been revised substantially to reflect the fact that banks now report many of these items in regulatory reporting.
- The worksheets for the QIS on the fundamental review of the trading book have been revised completely.
- The "OpRisk" worksheet has been removed.
- The "NSFR" and "Leverage" worksheets have been amended.
- The worksheets on floors, the review of the standardised approach to credit risk and TLAC have been added.

Annex 2: Tentative schedule for upcoming Basel III monitoring exercises

Basel III monitoring as of end-December 2014 ¹		
End-March 2015	Deadline for data submission to national supervisors.	
September 2015	Publication of results	
1		

¹ Or equivalent in countries with financial years which differ from the calendar year.

Basel III monitoring as of end-June 2015¹

early August 2015 Circulation of Basel III monitoring reporting templates to banks.

September 2015 Deadline for data submission to national supervisors.

March 2015 Publication of results

 $^{^{\}rm 1}\,$ Or equivalent in countries with financial years which differ from the calendar year.